Form 604

Corporations Law Section 671B

Notice of change of interests of substantial holder

To: Company Name/Scheme	West Australian Newspapers I	Holdings Limited	
ACN/ARSN	053 480 845		
1. Details of substantial hold	er		
Name	UBS AG and its related bodie	s corporate	
ACN/ARSN (if applicable):			
There was a change in the inter	ests of the substantial holder on	12 April 2011	
The previous notice was given t	o the company on	24 February 2011	
Provious and procent veting	a nower		

Previous and present voting power

The total number of votes attached to all the voting shares in the company or voting interests in the scheme that the substantial holder or an associate (2) had a relevant interest (3) in when last required, and when now required, to give a substantial holding notice to the company or scheme, are as follows:

Class of securities (4)	Previo	us Notice	Pre	sent Notice
	Person's Votes	Voting Power (5)	Person's Votes	Voting Power (5)
Ordinary	28,547,618	13.00%	33,477,444	7.28%

3. Changes in relevant interests

Particulars of each change in, or change in the nature of, a relevant interest of the substantial holder or an associate in voting securities of the company or scheme, since the substantial holder was last required to give a substantial holding notice to the company or scheme are as follows:

Date of change	Person whose relevant interest changed	Nature of Change (6)	Consideration given in relation to change (7)	Class and Number of securities affected	Person's votes affected
Please see	Annexure A.				

4. Present relevant interests

Particulars of each relevant interest of the substantial holder in voting securities after the change are as follows:

Holder of relevant interest	Registered holder of securities	Person entitled to be registered holder (8)	Nature of relevant interest (6)	Class and number of securities	Person's votes
UBS AG, Australia Branch	Seven (WAN) Pty Ltd	N/A	Relevant interest in shares held by Seven (WAN) Pty Ltd, a subsidiary of SGH, to the extent arising under the terms of the Block Trade Agreement pursuant to section 608(8) of the Act. See UBS AG Change in Substantial Holding notice dated 23 February 2011 for a copy of the agreement.	26,712,134 Ordinary	26,712,134
UBS AG, Australia Branch	UBS Nominees Pty Ltd	UBS AG, Australia Branch	Prime Broker with power to control the exercise of the power to dispose of shares pursuant to a Prime Broking Agreement (see attached)	101,480 Ordinary	101,480
UBS AG	Various custodians	UBS AG	Fund Manager with power to exercise control over voting shares	6,497 Ordinary	6,497
UBS Global Asset Management Life Limited	Various custodians	UBS Global Asset Management Life Limited	Fund Manager with power to exercise control over voting shares	7,251 Ordinary	7,251
UBS AG, London Branch	Various custodians	UBS AG, London Branch	Power to control disposal over shares pursuant to stock borrowing and lending activities (see attached)	217,024 Ordinary	217,024
UBS AG, London Branch	Various custodians	UBS AG, London Branch	Beneficial owner	60,542 Ordinary	60,542
UBS Securities Australia Ltd	Warbont Nominees Pty Ltd	UBS Securities Australia Ltd	Power to control disposal over shares pursuant to stock borrowing and lending activities (see attached)	1,700,000 Ordinary	1,700,000
UBS Securities Australia Ltd	Brispot Nominees Pty Ltd	UBS Securities Australia Ltd	Beneficial owner	4,404,626 Ordinary	4,404,626
UBS Securities LLC	Various custodians	UBS Securities LLC	Power to control disposal over shares pursuant to stock borrowing and lending activities (see attached)	267,890 Ordinary	267,890

5. Changes in association

The persons who have become associates (2) of, ceased to be associates of, or have changed the nature of their association (9) with, the substantial holder in relation to voting interests in the company or scheme are as follows:

Name and ACN/ ARSN (if applicable)	Nature of association
N/A	N/A

6. Addresses

The addresses of persons named in this form are as follows:

Name	Address
Details of all UBS offices can be found through the following link: http://apps.ubs.com/locationfinder	

SIGNATURE

Print Name: Boris Lo Capacity: Authorised signatory

Sign Here: Date: 13 April 2011

Print Name: So Young Kim Capacity: Authorised signatory

Sign Here: Date: 13 April 2011

Contact details for this notice:

Tiffany Leung Legal & Compliance (T) +852 2971 8042

(F) +852 2971 7895

25-Feb-11	25-Feb-11	25-Feb-11	20-Fe0-11	25.Fab. 11	25-Feb-11	25-Feb-11	JE 505 11	25 Eph 11	- 094-CZ	25-Feb-11	25-FeD-	27.150-11	22.760.11	25.50	7E Ech 11	25 50	25.Feb. 11	75-Feb. 11	25-Feb-11	75-Feb-11	25-Feb-11	25-Feb-11	75-Feb-11	25	25 Eab 11	25-Fah-11	75 Feb 11	25 Feb 11	25-Feb-11	25-FeD-1	25-FeD-11	25-Feb-11	70-10-1	25-reD-11	75 Esh 11	25-Feb-11	25 Feb 11	25-Feb-11	25-Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24 Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24-Feb-11	24 Feb 11	24-Feb-11	24-Feb-11	24 Feb 11	24-Feb-11	24-Feb-11	24-Feb-11	24-1-60-11	24-rep-1	24 Feb	24-FeD-11	24-F80-11	24-Feb-11	24 Fab 11	74-Feh-11	chance	2000										
UBS Securities Australia Ltd	UBS Securities Australia Ltd	I BS Securities Australia Ltd	OBS Securities Australia Ltd	LIBS Securities Australia Ltd	LIBS Securities Australia I to	I IBS Securities Australia Ital	I IBC Cookition Allertalia 144	I IBC Cook they Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Obs Securities Australia Ltd	Logo Securities Australia Lto	I DC Contribes Australia I to	LIDE Contribution Australia 14d	I IBS Sociation Australia (4)	I BC Securities Australia I M	IBS Securities Australia Ltd	UBS Securities Australia I to	LIBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia I to	LIBS Securities Australia Hd	LBS Securities Australia Ltd	Link Cook of the Australia 11d	IRS Securities Australia I to	IDS Contribute Australia 144	Obo Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	USS Securities Australia Ltd	Oco peculities Australia ciu	OBS Securities Australia Lig	OBS Securities Australia Ltd	UBS Securities Australia Ltd	LBS Securities Australia Ind	Obs securities Australia Ltd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS Securities Australia Ltd		UBS Securities LLC	UBS Sécurities (LC	UBS Securities LLC	UBS Securities LLC	Securities LL	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Ind	LBS Securities Australia Ind	LIRS Securities Australia Ind	I RS Sequities Australia I to	LIDS Sociation Australia Ltd	UBS Securities Australia Ltd	UBS AG, LONGON BLANCH	UBS AC Landon Brook	Lips Countries Australia Ltd	LIBS Securities Australia Ltd		SBU	rester of one temperature on charges																		
		7,100																																																Stock returned	Stock returned	Stock returned	Stock retu	Stock retu	Stock retu	Stock returned	Stock retu	Stock retu	Stock retu																			st clienijeti kathe si change	and the same of th
Buy 6		107	342 342	28	Buy I	PCE.		BUY 8/0		3uy 15,854	Buy 6,592			BUY /,310			5Uy 623		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Din.			Box 5000			200	6UY /24		Buy 34	572	Buy 11	505		Buy 505		Buy 185	Buy 11	Buy 90		BUV 240			Seill 201		Sell 202	Sell 202	Sell 202	Buy 30,404	32,			ned N/A			ned N/A	ned WA	ned N/A		ned N/A	96	Sell 50	Sell 11	-			5 361				Sell I	Sell 1,65	Sell 99,002	Buy 5,172	Buy! 19,318	14,042	14.04	6 N 150 150	Sons de l'adout grande	
10	,	19	61	5	2	10	101	155	36	2,826	1,175	13	823	1,303	15/	1	5 =	4	2 2	1,551	1 551	570	1 073	006	04	2	67		6	102	2	90	5	90		EE	2	16	36	43							(36)	5,		_		(50,414)		}		(6,000	(14.932		-	(17.977	io i	73		1500		(973			1,000	2	(300		942		2,542	1	8	organisms.	8
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	ordinary	urdinary	Ordinary		Ordinal A	Ordinary	Ordinary	Cruiraly	Ordinary	T	Ordinary	Ordinary	Ordinary	Т	Ordinary	Ordinary	Ordinary	Ordinary	T	l	Ordinary	П	П	Ordinary	T	7	Ť	7	Ť	Ť	T	Ordinary	T	П	Ť			П) Ordinary	Т	7	Ī	T	Ordinary	Ordinary	Ordinary	Ť	T	7	1	Ť	Ť	T	Ť	Ť	Т	Ordinary	T	T	Ordinary	5	Lister.	

01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01- Vidi-11	01 140	01 May 11	01 149-11	01 May 11	Of Mar 11	OI-Mar-	01-Mar-11	01.149	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11					- 1	- 1	- 1	01-Mar-11	7	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	OI-Mar-11	O1 Mar 11	UI-Mar-II	OI-Mar-II	01-Mar-11	01-Mar-11	O1-Mar-11	Ol-Mar-11	01-Mar-11	O1-Mar-1	01-Mar-11	01-Mar-11	28 Cab 11	28-reb-11	28-Feb-11	2/-heb-11	2/-Heb-11	27-Feb-11	2/-reb-11	2/-reb-11	27-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11	25-Feb-11						
UBS Securities Australia Ltd	OBS Securities Australia Ltd	Und Committees Australia Liu	Ope Securities Australia Ltd	I IDC Cocurios Australia Livi	LIBS Con Little Australia Eto	LIBS COCILIES AUSUALIA LIQ	LOBS Securities Australia Ltd	Libe counties Australia Ltd	DBC Committee Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Sequities Australia I fo	I IBS Securities Australia 11	Obs Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Itd	LIBS Securities Australia 114	UBS Securities Australia I to	IRS Securities Australia III	IRS Securities Australia Ltd	UBS Securities Australia 14	UBS ACTIFICACIONES AUSTRAIRA LEG	CDS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities LLC	UBS Securities LLC	UBS Securities LLC	UBS Securities Australia Ltd																																															
80	Buy	Bu	B.	Bu	BU	Cha		, BL	2 80	BU	BUy	200	D- 00	B	Bu	Suv Buv	В	Bu	yu8	-8	BL	BL	Buy	Bu	Buy	В	Виу	BL	B	2	22/5	B	S	S	Se	2	E C			Виу		Виу	S	Ві	18	Buy	D .	B. D.	0.00	Sel Sel	2	200	5	200		7		B	9	, v.	Ñ	S	S	Bi	S.	Stock returne	Stock returned	Stock returne	BI BI	В	Bi	Ē	9	B	Виу	B	E	8		В
7/09 3/11/20	1,118		ıy 1,304						3,855				770	1 127			2,070			1,392	17	Jy 51	1				1,835		1.739		1 337	W 17	77		731	4.2	73	23	23	73			ell 73										12.816		J 37,803													ed N/A		uy 2,261		uy 847	uy 774		uy 937		uyl 6	uy 39	Buy 6	uv] 34
199	199	232	232	561	561	242	189	686	686	115		48	210	340	202	80	367	197	197	247	3	9	308	159	213	50	325	7	OCC.	7	227		(12)	(A)	(12)	(758)	/20	750	. 4	13	(4)	148	(13)	13	500	163	977		, / ₈	750	(4,9/5)	4,9/5	2,2/8	(942)	5,653	(26)	(2)	4,506			(645)		(471)		(1)	(24,000)	(111,200)	(42,068)	628	403	291	151	8£1	1	167	6	1	7	1	6
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cidillala	Ordinary	Ordinany	Ordinany	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary			П		Т	Ordinary																																			

03-Mar-11 03-Mar-11	П	03-Mar-11						٦		-	_	_	03-Mar-11	03-Mar-11	US-IVIAI-I	US-Mar-II	U3-Mar-I	OS-MI-I	OD-IVIdI-1	O3-Mar-	00 140 - 11	03-Mar-11	OS-IVIdI- I	OS-Mar-11	03-Mar-11	U3-Mar-11	03-Mar-11	03-Mar-11	U3-Mar-I1	OS-Mar-II	O IVIdi - I	00 140	03-Mar-11	03-Mar-11	OR-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	- PIAL-CO	OS-IVIAIT-11	03-Mar-11	OR-Mar-11	O3-Mar-11	03-Mar-11	02-Mar-11	02-Mar-11	07-Mar-11	02-Mar-11	02-Mar-11	01-Mar-11	OI NAT 11	OI Mar-II	C1-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11	OI-Mar-II	OI-Mar-II	OI-IVIGE-1	01 New 11	Ol TIVIDI - I	OI-Ividi-	O1 N-W31-11	OI-MAI-II	OI-MAI-II	01-14-11	01-Mar-11	01-Mar-11	01 Mar 11	01-Mar-11	01-Mar-11	01-Mar-11	01-Mar-11							
UBS Securities Australia Ltd	Cos securities Australia Ltd	UBS Securities Australia Ltd	UBS Con they Australia Ltd	Dec Cool littles Australia Ltd	Coo Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	Obs securities Australia Ltd	Opp Securities Australia Ctd	LIBS Securities Australia 14d	LIBS Securities Australia 11d	I BS Societies & Istralia Itd	LIRS Securities Australia Ite	I IBS Securities Australia Itd	I BS Sportition Augustalia I to	DE COST SECURIOS AUSU dila LUO	UBS Securities Australia Ltd	UBS Securities Australia I td	I IRS Securities Australia 11d	I BS Securities Australia I to	I BS AG LODGO Branch	URS Securities Australia I to	LBA Securities Australia Ltd	UBS Securities Australia Ltd	I BS Securities Australia I to	UBS AG London Branch	Opposecutives Australia Ita	OBS Securities Australia Lto	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	COO DECOLUTED MODIFICATION	OBS Securities Australia LLO	Coo perorities Australia (10	Op Securities Auswalld Ltd	OBS Securities Australia Ltd	OBS Securities Australia Ltd	OBS Securities Australia Ltd	TIBS Contition Australia III	TIDS Securities Australia 14d	IBS Securities Australia Ltd	I IDC Cocurties Australia Ltd	IBS Securities Australia Itd	II IBS Sprintine Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd																																	
BL BL	В	Buv	BL	Buy	Se	В	В	Buv	E I	P. S.	2	ЭВ	light.	Buy	Bu	Bu	B	8.	Bu	B	80	Виу	B1	B	Bu	Buy	Виу	Bu	В.	8	B		9				Buy		51	B	Bi	Виу	Bi	B	В	B	200	Buy	0.00	2		Sel	2	2		200			8	8	8	В	В	8		8			8				8		200					B	В
y 49 1,507	Jy 114	JV 114	27	y 949		189		W 27		2 091	588	313		Jy 2,355	JV 16		Jy 6,543			Jy 226	485	7y 593		3		900		Jy 22	425		Jy 205		/48	192	LLY HO	2,000	S S S S S S S S S S S S S S S S S S S	Uy O4	Ly 538	uy 705	uy 16	16)	uy 16	uy 178	393	uvi 206	70	75	uy /o	25						ell 124,/69			Buy 2,266		Buy 589	uy 757	5,	uy 1,823				Buy 264				3,816				754		_	Buy 2 522		2
9 279	21	21	5	176	(176)	35	87	лř	102	388	109	58	_	437	3	83	1,214	234	23	42	8	110	194	563	106	167	21	4	79	4	38	58	139	130	100	200	,		100	131	3	3	3	33	73	38	10 1	1/2	14	(351)	(55)	(36)	(35)	(36)	7367	(22,254)	154	256	404	714	105	135	973	325	17	114	114	4/	61	235	234	679	61	190	0	54	54	044	27.	77	483
Ordinary Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	T	T	Ordinary	Ordinary) Urdinary	Ordinary	Ť	Ť	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	t	H	Н	Г	Г	Н	┢	T	t	١	t	t	Н	t	t	t	+	t	+	Ordinary	Ordinary	Ordinary																											

<u> </u>	191	ଣ	ଘାର	315	្រ	Gi				3[6	318	319	315	218	313	äis	3 5	315	10	10	9	<u></u>	្លា	္သု	<u> </u>	রা	318	315	:[⊆]c	9	ļo.	0		9	্রা	ा	ol:	ा व	ा	गट	ा	io.	Ö	[OI	ol	9	<u>ol</u>	া	া	ग्र	ंडि	ाट	ा⊂	गठ	ile.	i c	lo.	Ιö	Tö	10	ा	ाठ	I O	ı	0	lo	ाट	i o	olo.	ol C	गट	जट	io.	ol C	77-
Mar-11	-Mar-11	-Mar-11	-Mar-11	-War-	-Mar-11	-Mar-11	-Mar-11	7	7	Mar	-	= =	-Wei-		Mar-11	1	1	-Nar-II	-Mar-11	3	-Mar-11		-1	3-Mar-11	-1	1	Nar-11	s-Mar-II	s-Mar-II	s-Mar-11	S-Mar-11	3-Mar-11	Mar	-Nar-1	Y-Widi-1	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	a-Mar-	R-Mar-11	S Not 1	3-Mar 11	S-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11						
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Austral	UBS Securities Austral	UBS Securities Austral	I BS Securities Austra	LIBS Sociation Austra	I IBS Securities Austra	Do Securities Austra	TOBS Securities Austra	Obs Securities Austra	PINCHA CAPITION OF COLOR	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	I IRS Securities Austra	LIBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Australia Ltd	LIBS Securities Austra	Decorres Austra	Libs Securities Australia Ltd	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	LIBS Securities Austra	I IBS Securities Austra	IBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	Securitie	UBS Securities Austra	UBS Securities Austr																												
lia Ltd	lia Ltd	PTP	la Liu	100	lia Ltd	lia Ltd	lia Ltd	lia Ltd	lia I to		ia cio		ווא רוע	a Lic		2 10	na Ltd	lia Lto	liaLtd	lia Ltd	lia Ltd	lia Ltd	lia ttd	lia Ltd	lia Ltd	is to	isa Ltd	lia Ltd	lia Ltd	illa Ltd	iia Ltd	ilia Ltd	ilia Ltd	ilia Ltd	ilia Ltd	lifa Ltd	lia Ltd	la Ltd	alia tea	sia Cio	alia Ltd	alia Ltd	lia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ito	alia Itd	Sia Ita	Sia Ltd	alla Ltd	aliaLtd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia £td	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	alia Ltd	s Australia Ltd	alia Ltd	Australia Ltd	alia Ltd
																																													İ																															-
Buy 93	Buy	Buy 27	22	Buy 1,888		Buy 200	Buy 1	Buv 23	1,40 1,00	9		812 670											_	8uv 70			-	Buy 2,041				Buy	Buy 1	Buy	Buy	Buy 2		Bir				- Luci			Виу 2							Buy				1,		Виу			8uy 5	Виу 1,5	Buy						_		Buy	Buy	Buy			1,
172	10	51	101			37	73 32	44	100	J		00.1						78 33			54 10		2	70 13			100	378		444 82	_	22 4	79 33	16 3	3	39		08 20							238 44			60 11			43/	27	86		81		79 33	38 7	400 74		63. 104	53 287	92 17	378 70		,231 (228	37 118		,739 322		26 116	27 . 5		676 125		307 242
Ordinary Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	П	T	7	T	Ordinary	1	Т	Т	П	П		П	1	1	+	+	Т	Т	1	Т	Т	т	Т	┰	Н	Н	Н	Ordinary	Ordinary	Ī	Н	┢	Н	Ι	Н	_		ı	+-	Ordinary	1	_	1

04-Mar-11	04-Mar-11	04-Mar-11	U4-Mar-11	04-Mar 11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	11-Mar-10	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	03-14-11	03-Mar-11	03.Mar-11	03 Mar-11	03-Mar-11	OR-Mar-11	OR Mar 11	03-Mar.11	03-Mar-11	03-Mar-11	03-Mar-11	OS NAC 11	OS-IVIAI- I	OS-Mar-II	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03 Mar 11	03-Mar-11	03-Mar-11	ON Mar 11	03-Mar-11	03-Mar-11	03 Mar 11	03-Mar-11	OB-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03 Mar 11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11	03-Mar-11									
UBS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	THE CONTRACT A STATE OF THE	I IBS Sequities Alletralia Ita	LIBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltc	UBS Securities Australia Ltc	UBS Securities Australia Ltc	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltc	UBS Securities Australia Ltd	UBS Securities Australia Lto	UBS Securities Australia I to	UBS Securities Australia I to	UBS Securities Australia I to	IBS AG London Branch	IBS Securities Australia Ltd	IBS Societies Australia to	IBS Securities Alletralia Its	I BS Securities Australia I to	LIRS Securities Australia In	LBS Securities Australia I to	I BS Securities Australia I to	LIBS Sen introduction for	IBS Securities Australia Ltd	I IBS Securities Australia I to	UBS Securities Australia (a	IBS Securities Australia I t	I DE Contilles Australia Lt	OBS Securities Australia Lit	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia L	UBS Securities Australia II	UBS Securities Australia Ltd	IBS Securities Australia I t	IBS Securities Australia I	185 Securities Australia II	IRS Securities Adeleral a 1td	IRS Securities Australia Ind	IRS Societion & letralia 1	LBS Sequifies Australia I to	IBS Securities Australia	IBS Securities Australia II	URS Securities Australia I	UBS Securities Australia II	BS Securities Australia I	UBS Securities Australia Ltd	UBS Securities Australia Li	UBS Securities Australia Ltd	UBS Securities Australia Li	UBS Securities Australia Li	UBS Securities Australia Li	UBS Securities Australia Ltd	IUBS Securities Australia L										
																				,	3												1			1	2 6								a	a	d	1	d	ď	d	id	d	d	d	d	ä	a	a k		ā	1	1 2		1 6	1 2		3 16	1 2		5 6	3. 8	1 8	EL.	ă	5 .	ä	6.	ã.	ã	ă
56	8	В	В	200	9	000	D. T.	0	Bus	B	Die.		Ring	Sell	8	5	B	В	В	8	B	В	8	Die Control	P	4.0			200									340										E																		,,,,,,		District Control													
2	1,394	Jy 219	Jy 41	Jy 24	1,54	195		1 065		77		1 214		ell 1.065			uv. 6	1,77	Buy 3	995 vu			Uv 53			222			-	V	auy .	1,/2	1 7	L L	The state of the s	Uy O	LVY.	2,10	Suy 316	E S						3uy 167		3uy 1,806									Buy 1,274													Bin Sin	3			Riv	BUY	Buy 2		BLW	BUV S	Buy 1,0	HIM
(2)	4 261		78	46	288				777	7		207	1122			(227)	4	333	7	_	66					12		(21,793)			. 0	025	300	360	-	20	120	400	300	99		312	140	9	37 222	37 31		335									74 236							67					111	577	436		36	22	22 4	11 39	61	27 5	93	sq 198	401 26
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ciulidiy	Ordinary	Ordinary	Cigillaly	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinaly	Ordinary	Ţ	Т	1	Ordinary	Ordinary	T	1	T	T	1	1	T	T	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	٦	T		Т	╛	T	╗	Ordinary	T	Т	T	Т	T	T	T	T	1	T	Ť	Ordinaly	Calary	Ordinary	1	t	1	1	Ordinary	1	T	1	7	

\mathbf{I}	08-Mar-11	П		ı	١	07-Mar-11	- 1	- 1	07-Mar-11	ļ	07-Mar-11	١	07-Mar-11	- 1	-		- 1	07-Mar-11	07-Mar-11	06-Mar-11	-	06-Mar-11	۱	04-Mar-11	-	-	04-War-11	ı		04-Ividi-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar 11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11	04-Mar-11		04-Mar-11	ŀ	ı	04-Mar-11		04-Mar-11																															
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS AG, London Branch	Securities	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia I to	UBS Securities Australia Ltd	LBC Securities Australia Ltd	I IBS Securities Australia III	On Decolutes Most and Ltd	CIDS Securities Australia 110	COV Decorners Australia Ltd	OBS Securities Australia Lto	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	LIBS Spourities Australia I td	LIBS Securities Australia III	UBS Securities Australia Ltd	JUBS Securities Australia Ltd	UBS Securities Australia Ltd	URS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd																																																
Buy Sell	Sell	Sell	Buy	Buy	Stock returned	Stock returned	BIN	B) av	Riw	D: 52	Pin	Buy	Buv	BUV	Ruy	Buv	Riv	Buy	ATIA TIME	Sell		Self.	Bin	Bis.	Buy	000	BUY	buy	, una	Buy	Buy	Buy	Buy	Виу	Виу	Buy	Buy	Buv	Bliv	Buy	Bin	Buy	Buy	Виу	Buy	Buy	Buy	Виу	Buy	Buy	Виу	Buy	Ang Ang	Bly	Buy	Виу	Buy	Buy	Виу	Sell	Buy	Buy	Sell	Buy	Buy	Виу	Buy	Buy	Виу	Buy	Виу	Sell						
101 42	64	673	3,927	Ī		N/A	7.438	4,307	4507	3 121	\$0	3 715	2.183	2044	V25 C	15.583	7.525	2.647	17.877	52	877	157	1841	1 52/	150	160	51.6	2/6	1,561	616	3,515	38	222	222	2,306	498	1,035	296	100	710	VEG	710	656	79	1,074	476	583	159	233	814	692	64	1 575	287	377	384	1,887	267	567	267	267	1,873	267	267	977	230	219	11	872	915	32	840	11	11	5	5	11	11
19 Ordinary (8) Ordinary	ll	(127) Ordinary	754 Ordinary	Т	(100,000) Ordinary	200 000 Ordinary	468 Ordinary	116 Ordinary	965 Ordinary	Cidilaly	17 Ordinary	Т	419 Ordinary	Т	Т	Т	Т	SOS Ordinary	Т	Т	1	1	+	+	32 Ordinary	t	۲	Ordinary	t	Т	Т	Г		_	П	1	┪	╅	┰	Т	Т	Т	Т	Г	Г		٦	30 Ordinary	7	153 Ordinary	+	1	T	T	119 Ordinary	T	Н	50 Ordinary		(50) Ordinary	Г	350 Ordinary	Ť	П	183 Ordinary	П	7	2 Ordinary	163 Ordinary	171 Ordinary	6 Ordinary	157 Ordinary	2 Ordinary	2 Ordinary	1 Ordinary	1 Ordinary	2 Ordinary	(2) Ordinary

BUSS SCOLITES Autrials LID Set 25.7 BUSS SCOLITES Autrials LID Set 25.7 BUSS SCOLITES Autrials LID Set 3.8 LUS SCOLITES Autrials LID Sup 3.8 LUS SCOLITES Autrials LID Sup 3.7 LUS SCOLITES Autrials LID <th< th=""><th></th><th></th><th></th><th></th></th<>				
Ang	٦	53	Buy	IUBS Securities Australia Ltd
AN B BAN B B	7	35	Buy	UBS Securities Australia Ltd
Ang	+	14	Ring	UBS Securities Australia Ltd
사용 사용 사용 사용 사용 사용 사용 사용	٣	1,25	D.W.	LIBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	۳	1,42	BUY	UBS Securities Australia Lio
Sell Sell Sell Sell Sell Sell Sell Sell	ľ	25	Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	۲	1,30	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell		49	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Buy	-24	9	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	러	14	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	Ÿ	1,19	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	7	3	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	Ÿ	2	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	٣	33	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	-	8	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	~	95	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	٦	56	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	Ť	53	Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Buy	۶	1	Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Buy Buy Buy Buy Buy Buy Buy B	٦		Áng	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Buy	7		Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	ω.		Sell	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Suy Buy Buy Buy Buy Buy Buy Buy Buy Buy B	w		Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	3		Buy	IUBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	_		duy	LOS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	T		buy	Coo decorres Australia Co
Sell Sell Sell Sell Sell Sell Sell Sell	Ť		Day.	IBC Specialities Australia 144
Sell Sell Sell Sell Sell Sell Sell Sell	Ť		Ding	TRY Securities Australia 111
Sell Sell Sell Sell Sell Sell Sell Sell	۳		Brw	URS Securities Australia (11d
Sell Sell Sell Sell Sell Sell Sell Sell	╣		Riv	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	Ť		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	51		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	8		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	73		Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	0		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	4		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	ĺ	5/	Ruy	Opo peculities Ausualia Liu
Sell Sell Sell Sell Sell Sell Sell Sell	Ť		The stay of the st	LIDS COUNTY AND A CONTROL LIST
Sell Sell Sell Sell Sell Sell Sell Sell			Day,	CDC COLUMN AND DOOR ON A LIVE
Sell Sell Sell Sell Sell Sell Sell Sell	1		duy	TIDO CONTRACTO A CATALON LAND
Sell Sell Sell Sell Sell Sell Sell Sell			- Puly	COO Decountries According Ltd.
Sell Sell Sell Sell Sell Sell Sell Sell	T		200	LIDO COCUMENTO ACCORDING TO
Sell Sell Sell Sell Sell Sell Sell Sell	- 0		Din	LIRK Securities Australia I to
Sell Sell Sell Sell Sell Sell Sell Sell	Ť		Bir	LIBS Securities Australia I to
Sell Sell Sell Sell Sell Sell Sell Sell	Ť	•	P. C.	UBS Securities Australia I tri
Sell Sell Sell Sell Sell Sell Sell Sell	7		Buv	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	7		Ang	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	٣		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	4		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	Ψ.		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	7		Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	7		Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	0		Buy	CBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell			, leading the second se	Opposed the Adodona Fig
Sell Sell Sell Sell Sell Sell Sell Sell	T		Ding	LIBS Sportifies Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	1		Bin.	BA Securities Australia I td
Sell Sell Sell Sell Sell Sell Sell Sell	7		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	Ĭ		Buv	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	2		Виу	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	7.		Buy	UBS Securities Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	C		Виу	Coo Securities Australia Eta
Self Self Self Self Self Self Self Self	1		500	I BE Consisting Australia Ltd
Sell Sell Sell Sell Sell Sell Sell Sell	#		Page	I BS Sequities Australia 11d
MAG MAG MAG MAG MAG MAG MAG MAG	J		Fi ny	UBS Securities Australia I to
1985 1985	7		Buy	UBS Securities Australia Ltd
Ang	5	,	Buy	UBS Securities Australia Ltd
192 192 193 193	0	,	Buy	UBS Securities Australia Ltd
Ang Ang Ang Ang Ang Ang Ang Ang Ang Ang	v	6	Buy	UBS Securities Australia Ltd
Ang	Ì		(na	ODD DECUIRES MUSI dila LUC
Ang Ang Ang Ang Ang Ang Ang Ang Ang Ang	+		Day.	Opp Securities Australia co
MAG MAG MAG MAG MAG MAG MAG MAG	1		Bir	IIBS Securities Australia Ita
Ang Ang Ang Ang Ang Ang Ang Ang Ang Ang	2		Buv	UBS Securities Australia Ltd
Ang Ang Ang Ang Ang Ang Ang Ang Ang Ang	12		Buy	UBS Securities Australia Ltd
Ang	7	100	Buy	UBS Securities Australia Ltd
Ang Ang Ang Ang Ang Ang Iles	٩		kns	UBS Securities Australia Ltd
Ang Mng Mng Mng Mng Mng Mng Mng Mng Mng M	2		BUY	UBS Securities Australia Ltd
Ang Ang Ang Ang Ang Ang Ang Ang Ang Ang	4		Buy	OBS Securities Australia Ltd
Sell	1		000	Opposed the September 11
And				IBS Societies Australia 1+d
	n c		Din.	II BS Spourities Australia I to
NAB	7	- The state of the	Res	UBS Securities Australia I to
	4		Buy	UBS Securities Australia Ltd
Sell Sell	ō		Sei	UBS Securities Australia Ltd
Sell	4		Se	UBS Securities Australia Ltd
	3		Sel	UBS Securities Australia Ltd
BUV	۲		(Na	OBS Securities Australia (VO
Re NY	十		Bro	It ind Can white A setrolin ted
	:) Sel	UBS Securities Australia Ltd

<u>ाठाठा</u>	516	ol c	ol c	जट	ol C	T	ि	<u></u>		0	0	0	10		<u> </u>		16	100	<u></u>				5			_		- T					<u> </u>	<u> </u>	·	<u> </u>		ol:	<u> </u>			~T-	~1	-1-	1	1 -	~T-		-T-		a I a		1.5	1.			-										_						_					_		_
	y-Mar-11	Ī	-	=	9-Mar-11	-	=)9-Mar-11)9-Mar-11			=	l)9-Mar-11	=)9-Mar-11)9-Mar-11)9-Mar-11	ŀ		1	Ш)9-Mar-11)9-Mar-11) 9- Mar-11)9-Mar-11	09-Mar-11	09-Mar-11)9-Mar-11	1	1	38-Mar-11	9-Mar-1	1	08.NASE-11		DR-Mar-11	DO-MINI-1	DS-Mar-11	os-Mar-II	De Mar 11	U8-Mar-11	08-Mar-11	ı	08-Mar-11		08-Mar-11	OG-IVIdi - I I	OG-IVIdI- I I	00 1437 11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11	08-Mar-11												
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Sequities Australia Hd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS AG, London Branch	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Itd	LBS Securities Australia I to	UBS Securities Australia Hd	I BS Securities Australia I o	I BS Securities Australia I d	FIRS Securities Australia Ital	I IBC Sportition Alletralia I to	UBS Securities Australia Ltd	Cos securites Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	Opp Securities Australia Ito	OBS Securities Australia Ltd	De constitue Australia Ital	I IRS Securities Australia I to	I IBS Securities Australia I to	LIRS Securities Australia Ltd	I IRS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd																																																	
Sell Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Śell	Sell	Seil	Sell	Sell	Buy	Buy	Sel	Sel	Sel	Sel	Sel	Set	Sel	Sel	Sel	Sel	Sel	Sel	Buy	Buv	Bun Bun	≠	Bin	Rin	Bin	Buy	Big	000	Ana	80	85	Bu	Bu	Bu	Se	Buy	Bu	Buy	Bu	NA SA	BU.			Big	Bu	P.	21		Bu	Bu	Ви	В	Buy	Bu	Buy	Bu														
190 11	189	189	190	190				169			189												89									190		16	79			1	1 200			10	10		1	-	7		177		125			y 1,578		y 42		977		y 68		y 848		V					130		10	л	5	5		1.009			y 130	1
(2)	(36)	(36)	(36)	(36)	(18)	(36)	(36)	(32)	(4)	(18)	(36)	(36)	(33)	(3)	(36)	(62)	(36)	(36)	(17)	(17)	(13)	(17)	17	17	(6)	(36)	(36)	(36)	(81)	(33)	(3)	(36)	(33)	(3)	(15)	(3)	226	239	342 347	(25 00)	270	J .	3	2	200	336	30 0	,34	· ·	2	24	9	(304)	304		8	125	188	2	13	21	163	u	33	27		17	200	×1	3 6	2 -	_	_		_	194	72		25	38
Ordinary Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	П	7	╗	7	╗	╗	╗	П			Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ciglialy	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	ordinary	Ordinary	t	+	+	+	┿	Ordinary	-	\vdash	Н						

10-Mar-11	10-Mar-11	10-Mar-11	10-Mar-11	O-Mar.11		10-Mar-11	1		1	10-Mar-11		ı	ı	ı	10-Mar-11	ı	ı		io-Mar-1	I U-Mar-11	IU-Mar-F1	10-Mar-11	I -Mar-II	10-Mar-11	10-Mar-11	10-Mar-11	10-Mar-11	10-Mar-11	Og-Mar-11	00 May 11	1-1PIN-60	OS Mar 11	09 Mar 11	09-Mar-11	09 Mar 11	09-Mar-11	09-Mar-11	09-Mar-11	09-Mar-11	Og-Mar. 11	OG Mar 11	00 142 11	00 Mar 11	00-149-11	09-Mar-11	09-Mar-11	09-Mar-11	09-Mar-11	09-Mar-11	09-Mar-11	00 Mar 11	00 Mar 11	09-Mar-11	00 May 11	09 Mar 11	09-Mar-11	09-Mar-11	09-Mar-11	OG-Mar-11	ng-Mar-11	On Mar 11	09 Mar 11																		
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	COO DECOLUTES MOSCIALITY	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS AG, London Branch	UBS Securities Australia Ltd	URS Securities Australia 11d	UBS AG Spran Branch	UBS Securities Australia Ltd	UBS Securities Australia 1td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I BS Securities Australia I to	I IBS Securities Australia I to	I IBC Socialities Australia 1td	I IDS Securities Australia Liu	PT SECURITY SOUTH SOUTH	The coordinate Australia 14	I BS Securities Australia I td	IBS Securities Australia 14	UBS Securities Australia Itd	LIBS Securities Australia I to	UBS Securities Australia Ltd	IRS Securities Australia Ind	TIRS Secriptions Australia 1td	LIBS Societies Australia (10	LIRS Securities Australia 14	ac all les	IRS Securities Australia Ltd	UBS Securities Alistralia to	UBS Securities Australia Ito	UBS Securities Australia 11d	I IBS Securities Australia Ltd	UBS Securities Australia Itd	I IRS Sprintips Australia I to	li IRS Spormities Australia I to																																
				į	1				-							i				-																							_																																	+			+	
Se	95	80	P 16	56) x	26	Se	Se	Se	56	Se	S	Se	5	Sel	Ş	Š	S.	Sel	Ş	Ş	Se	S	Ş	Buy	S	3	S	5	2	S	S	S	5	В	8		0	o la	,			B	B	8	В	В			8																													
	70			/3	2	94		73	21		88		36	92	119	187		188	1			188		7		,			ell 18	E III III	S III	ell 187	ell 18	ell 9	8	Buy 8.21		737 486	,	1,594						Buy 3,225	7		Buy 66		Buy 2.299			-		Busy 1,430				Buy 908		y y	109			37	Buy 1 21									Ruy A				,
(14)	(18)	4	(4)	(14)	(81)	(18)		(14)	(4)	(1)	(17)	(18)	(18)	(18)	4 (18)	7 (36)	5 (5)	36)	(20)	(16)	8 (36)	8 (36)	8 (36)	(3E) B	(3		(18)	(18)	8 (36)			7 (36)		9 (19)	9 (17)	2 1.571	7 797	1 41	נאנ	265				101			5 834		2 126		9 437			200	712	2/4	777			1/3			35				227	7 0	17	(1)		/ /		20	17			751 143		
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	t	†	1	t	t	Ť	Ť	t	Ordinary	Cidilaly	Croinary	Ordinary	1	۲	1	Ť	Ť	Ordinary	T	Т	T	٦

	Buy
7,714	Buy Buy
	Ang Ang
	Bu
	Bu
	<u> </u>
	Bu
	Bu
	Buy
	Br.
	a B
	Bu
	Buy
	B. B.
	Se
	Buy
	BL
	P. D.
	Buy
	Buy
	Se
	S. B.
	Şell
	812
	2 %
	56
	- Se
	Se
	25
	Buy
	25
	Sell
	Stock portowe
	Bt Stock bostonia
	BL
	52
	Sell
	0 4
	S
	22
	2 V
	20
	S
	25
	2
	2
	52
	Sell
	C 4
	100
	מ

14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11	14-Mar-11 14-Mar	14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11 14-Mar-11
ubs Securities Australia I.d.	UBS Securities Australia Ltd UBS Securities A	UBS Securities Australia Ltd
Aven Aven	Ang B	Buy Buy Buy Buy
444 151 94 57 57 47 47 8,429 1,529 1,529 6,464 827 546	188 188 188 505 505 167 1667 1667 159 159 159 159 159 159 159 159 159 159	3,553 5 3,782 1,579 1,224
29 Ordinary 119 Ordinary 119 Ordinary 11 Ordinary 11 Ordinary 11 Ordinary 153 Ordinary 1624 Ordinary 1,624 Ordinary 1,624 Ordinary 1,624 Ordinary 1,624 Ordinary 1,624 Ordinary 1,625 Ordinary 1,625 Ordinary 1,626 Ordinary 1,627 Ordinary 1,628 Ordinary 1,628 Ordinary 1,628 Ordinary		682 Ordinary 1 Ordinary 726 Ordinary 303 Ordinary 235 Ordinary

निर्	. 1-	-1	ı.	E	.1-	4	1-	ĻΤ	. -T	<u>,-1</u> .	<u>-1-</u>	ন-	-1=	T=	1-	1-	<u> </u>	<u>-1</u>	1-	-1-	·I	1-4	<u>=</u> 1	→ 1.	I-	-1-	·1	1⊐	<u> </u>	<u> </u>	<u> </u>	-: r-	-I -	·1	1=1		<u> </u>	-1-	1=	T=-1	ابد	_1			.1	125				-1					7			-11	_										_	
111	4-Mar-11	4-Mar-11	П	4-Mar-11	1	1	l	П	- 1		П	1	T	ı	1-	ŀ	П	Į	4-Mar-11	1	ı				4-Mar-11	4-Mar-1	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	4-Mar-11	A-Mar-11	4-Mar-11	4-Mar-11	14-Mar-11	14-IVIdI-I I	14-IVIdI-11	14-Mar-11														
UBS Securities Australia Ltd UBS Securities Australia Ltd	URS Securities Australia Ltd	CBO Decreiones Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBC Securities Australia Ltd	UBS Securities Australia Hd	UBS Securities Australia Ltd	UBS Securifies Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	ILIBS Securities Australia IIId	OBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBS Securities Australia I td	It IDS Securities Australia I fri	UBS Securities Australia Ltd																																													
kng Ang	Виу	Buy	Buy	Виу	Birry	Buy	Виу	Ang.	Bliv	Spill	Buy	Виу	Виу	Buy	Виу	Buy	Buy	Buy	Blu	Buy	Виу	Виу	Buy	Riv	Buy	Виу	Buy	Виу	Buy	Виу	Buy	Buy	Buy	Buy	Buy	Виу	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Виу	Buy	Buy	Buy	Bu)	Buy	Виу	Buy	ng.	But	Buy	Bu	813	BU	Bell	BI	B	Rep	l l								
4,590 744	5 210		36			634		213					1					-	4/6					1 594	31	2,251	1,912	99	26	406	552 552	2,0/2	1,610	214	151	114		16	_	26		2689.0			1,924		***************************************		468		1,180		213			810		192								4.196				192
881 143	1,000	7	7	12	286	122	9	41	41,	(17)	30	500	231	18	36	36	36	61	20 0	129	36	36	1	305		432	367	19	5	78	106	41	905	41	29	22	18		24	5	7	515	3/	7	370	164	35	23	909		227	39	41		113		489		58											37
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary		П	П			- 1	Τ	Т	Г		Т	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Н	7	+	+	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	+	۰	Ordinary			П	7	_	_	_	_	┰	_	Ordinary									

.

15-Mar-11 15-Mar-11	15-Mar-11	15-Mar-11	U-IVIdI-I	U-Mai-	D-Mar-11	D-Mar-1	D-Mai-	15-Mar- 11	11-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	4 Mar 1	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14 Mar 11	14 1497 11	14-Mar-11	14-Mar-11	14-Mar-11	14-Vidi - 1	14-War-11	14 Mar	14-War-11	14-Mar-11	14-War-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mai-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-10Id - 1	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11	14-Mar-11
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	OBS Securities Australia	UBS Securities Australia	OBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	UBS AG, London Branch	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS AG, London Branch	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia 11d	UBS Securities Australia	LIBS Securities Australia	ILRS Securities Australia I to	LIBS Societies Australia	IBS Securities Australia	LIBS Securities Australia	LIBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	URS Securities Australia	LIBS Securities Australia	UBS Securities Australia	IRS Securities Australia	DC Securities Australia	LIBS Securities Australia	Upo Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	OBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia Ltd	<u>@</u> :	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Austrana Ltd	Con Securities Australia	LIBS SOCIETY ACSTRAIN	LIRS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australi																									
[td	Ltd	8			100	[2	ta	6	Itd	Ltd	Ltd	Ltd	Ltd	Ltd	Ltd	Ltd	Ltd		Ltd	Ltd	LEG		Ltd.	Itd	The last	Tid.	1	1	1	5. 2	The last	1	Ltd	Ltd	Ltd	Lo.	Ltd.	16.		7 2	2 2	100	1 8	1.8	100	1 LIO	LTO	J. Tid	a Ltd	3 TIQ	la la	Ltd	aLtd	a Ltd	a Ltd	aLtd	a Ltd	a Ltd	a Ltd	a Ltd	aLtd	a Ltd	aLtd	a Ltd	attd	aLtd	aLtd	a Ltd	9 10	9 10	a Lio	1	a td	a id	aLtd	aLtd					
	100																	1												100																			·in.—																									1,100							
ell 103 ell 273	ell 82			186	186						Buy 93						sell 9.			20.243		8	1,14 6E	11/1	2,032	7.	1000		4,992			0000 0000 0000 0000 0000 0000 0000 0000 0000				Buy 109	7	Rily 1 497	25	SCIII 25		Neil Sell			BUY 26	1,893				3	Виу 187					Buy 1,40	Buy 17	Buy	Виу	Buy Yu8	Buy	Buy 4.70	6 Ang		Buy, 385					Buy 5		Buy 2,175			Buy				Buy		4
											3				(36)	_					9 227 800	120	3 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	710	066						1,730			77		2000	1		(00)	48		(48)						75 149			37 36			87 36	,	270	72 33	38	31	38 17	19	905	10 175					3	82 35	21 100		75 419				1/0			55 49	796 153	8
Ordinary Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cadinary	Ordinary	Ordinary	Ordinary	Ordinary	Ciclial	Ordinary	Ordinary	Cicion	Ordinary	1	Т	T	+	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	т	1	_	✝	_	1	Н	Н	\vdash		7	Ordinary	7	7	1	1	1	7	7	7	T	T	╛	Ordinary	T				Г	Ordinary	Г	Т	T	Ordinary	Ţ											

15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	-Nar-11	Mar-II	15-Mar-11	.15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15 Mar 11	D-Ividi-II	15-Mar-11	16 V-Midi-	15 May 11	TE MAY 1	D-Mdr-II	15-Mar-11	5-Nar-11	15-Mar-11	15-Mar-11	15-Mar-11	15-Mar-11	15 Mar-11	15. Mar. 11	15 Mer 11	- IPINI-C	- JPINI-C1	15-Mar-11	D WIN	15-Mar-11	15-Mar-11	15-Mar-11	- Nar-i	IS-Mar-II	15-Mar-11	5-Mar-	15-Mar-11	15-Mar-11	I > IVIAI - I	- IDIVI-C	15-Mar-11	15-Mar-11	15-Mar 11	15-Mar-11	15-Mar-11	D Md																														
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	U85 Securities Australia I td	URS Securities Australia I to	LIBS Securities Australia Ind	Coo Securites Australia Ltd	LIBS Securities Australia Ltd	Upo pecurites Australia Ltd	Opposed in the second s	LIBC Cocuraios Australia Lid	UBS Securities Australia Ltd	UBS Securities Australia I td	LIBS Securities Australia Ind	UBS Securities Australia I to	UBS Securities Australia I to	LBS Securities Australia Ltd	IRS Securities Australia Ltd	DIS Securities Australia Ltd	UBS Securities Australia Ltd	U85 Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	I IRS Securities Australia I M	LIBS Securities Alistralia trd	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd																																																		
B. B.	В	Buy	Buy	Bı	Bu	В	Buy	Bu	Виу	Bı	B	Buy	Bu	Buy	B	Bi		S	Ş	5	2	2	\$		\$	2					S	S		0.00		9							S	В	2	S	S	S		20											2			2		2													
лу 251 лу 1,288		4,550							uy 3,642		uy 10,799			uy 321		uy 275		Sell 272			sell 93	ell 457		457	185	185	93	93	02	93	D. C.		45.7	457	195	200	93	459	2/3	185	P	300	458	uy 459	459	vell 459	Sell 366	366	3	366	366	1	772	100	502	2/3	2/3	bell 2/3	2/3	Sell 2/3	Sell 273	185		sell 185					93			Sell 185		-					
49 Ordinary 251 Ordinary	Н	887 Ordinary	+	┪		186		Ĥ	710 0	┪	┪	7	7	62	┪	53	(36)	Ĭ	7	(89)	7	7	7	╗	(36)	+	┰	Т	(19)	╅	┰	Т	7	7	+	+	(18)		1	۲	T	┲	1	_	П	(89)	- 1	_	T	T	(71)	T	T	Т	(10) Ordinary	T	Ť	(53)	Ĭ	Γ	(53) Ordinary	Г		П			Г				Г	(36)	Г	Т	1	T	T		Г

			OBS Securities Australia Ltd
T	273	Sell	Coo Securities Australia Ltd
Ť	93	Sell	UBS Securities Australia Ltd
	185	Sell	UBS Securities Australia Ltd
Ť	272	Sell	UBS Securities Australia Ltd
Ť	272	Selli	UBS Securities Australia Ltd
Ï	273	Seli	UBS Securities Australia Ltd
٦	93	Sell	UBS Securities Australia Ltd
Ť	36	Sell	UBS Securities Australia Ltd
Ť	501	8	UBS Securities Australia Ltd
Ť	25.5	Sel	UBS Securities Australia I td
Ť	41	25	UBS Securities Australia Ltd
Ť	773	6	UBS Securities Australia Ltd
Ť	270	[42	UBS Securities Australia Ltd
1	273	l eS	UBS Securities Australia Ltd
	55	lleS	UBS Securities Australia Ltd
Ť	27:	Sell Sell	UBS Securities Australia Ltd
1	546	Spil	UBS Securities Australia Ltd
1	277	Sall	UBS Securities Australia Etd
Ť	77.	Sal	UBS Securities Australia Hd
Ť	л.	Call	URS Securities Australia Etd
Ť	10	Sal	UBS Securities Australia Ltd
Ť	n i	Sall	UBS Securities Australia Ltd
Ť	91	Sall	UBS Securities Australia Ltd
٦	57	% 	UBS Securities Australia Ltd
~	273	Sell	UBS Securities Australia Ltd
Ĩ	27:	Sell	UBS Securities Australia Ltd
	129	Sell	UBS Securities Australia Ltd
+3	144	Sell	UBS Securities Australia Ltd
٦	10	Sell	CDO DECOLRES AUSTRALIA CIG
T	10	guy.	I IBC Cocurrence According Land
Ť		Dell'	I IBC Contribut Australia 144
Ť	93	Call	UBS Securities Australia 11d
"	16	[95	UBS Securities Australia I to
1	23	45	UBS Securities Australia Ltd
Ĭ	27:	Sell	UBS Securities Australia Ltd
3	:72	Self	UBS Securities Australia Ltd
	1,5	Sell	UBS Securities Australia Ltd
5	186	Sell	UBS Securities Australia Ltd
Ť	188	¥.	CDS Securities Australia Ltd
Ĭ	181	¥	Coo becomes Australia Lto
Ĭ	9	YES	Cap pecalities Hustialia Liu
ľ	181	Sel	Coo peculicies Australia Ltu
	į	Jell Jell	Opp pecquires wood and Fig
T	2		LIBS Cocurities Augustiana Etg
Ť		-	I BC Securities Allestralia Ltd
Ť	101	Call	I IRS Securities Australia I to
,,,,	196	Sall	UBS Securities Australia I to
Ť	12	(A)	UBS Securities Australia I to
٦	7.760		UBS Securities Australia Ltd
J	7.760	Buy	UBS Securities Australia Ltd
7	334,03	Sell	UBS Securities Australia Ltd
7	334,03	Buy	UBS Securities Australia Ltd
٦	7,76	Buy	UBS Securities Australia Ltd
/	27,697	Buy	OBS Securities Australia Ltd
1	51.5	Buy	OBS Securities Australia Ltd
Ì	334,03	Buy	Cop Secouries Washing Fro
		, was	TIPE CONTRIBUTE ACTUAL TO
T		D	I BC Secription Ametralia 14d
1		Britis	URS Son without Australia Ltd
7		Buy	IUBS Securities Australia Ltd
4		Buy	UBS Securities Australia Ltd
4		Buy	UBS Securities Australia Ltd
F		Buy	UBS Securities Australia Ltd
4		Buy	OBS Securities Australia Ltd
f		виу	Con Securities Australia Lia
+		yua	Our permittee Adollate Fig.
ľ		64	LIDS Consisting Australia Lau
7		Din.	I IRS Securities Australia I td
7	5,180	Buy	UBS Securities Australia Ltd
2		Buy	UBS Securities Australia Ltd
4		ния	UBS Securities Australia Ltd
1		751	COO DO, LOS COST CIGNICA
7		Sal	UBS AG. London Branch
Ť		<u> </u>	ITIRS AG London Branch
4		<u>≨</u>	UBS Securities Australia Ltd
ω		Sell	UBS Securities Australia Ltd
2		Se	UBS Securities Australia Ltd
	Action (Name and Associated	Stock Policomed	יכים הבימו וופס אינטומומ רוני
Ť			I IRS Securities Atternalia I to
9		Sel	UBS Securities Australia Ltd
w	93	Sell	UBS Securities Australia Ltd
ņ		Sell	UBS Securities Australia Ltd
u.		Self	OBS Securities Australia Ltg
· C		SE	Opposecutives Australia Lig
10	10,770	Jet Jet	Con permittee wood and end
۲		(a)	I DO Cocurition Australia Ind
=			
		Sel	UBS Securities Australia Ltd

Ordinary	130	670	Виу	UBS Securities Australia Ltd	16-Mar-11
۳	14	72	Buy	UBS Securities Australia Ltd	16-Mar-11
7	32	1,648	Виу	UBS Securities Australia Ltd	16-Mar-11
~1		26	Buv	UBS Securities Australia Ltd	16-Mar-11
Τ,	14	282	Pic	UBS Securities Australia I tri	16-Mar-11
"[21	Bi w	LBS Securities Australia Ltd	16-Mar-11
4		2	Rim	UBS Securities Australia I to	16-Mar-11
'nΓ		287	Buy	UBS Securities Australia Ltd	16-Mar-11
7		10	Biv	UBS Securities Australia I td	16-Mar-11
~[67	Buy	UBS Securities Australia Ltd	16-Mar-11
<u>,</u>		83	BLIV	UBS Securities Australia Ltd	16-Mar-11
_[5	βυν	UBS Securities Australia Ltd	16-Mar-11
7		10	Виу	UBS Securities Australia Ltd	16-Mar-11
٧		41	Buy	UBS Securities Australia Ltd	
2	1	62	Buy	UBS Securities Australia Ltd	16-Mar-11
+-		21	Виу	UBS Securities Australia Ltd	l
ıΨ	S	15	Buy	UBS Securities Australia Ltd	
Ľ	8	423	Buy	UBS Securities Australia Ltd	
۳	5	284	Buy	U85 Securities Australia Ltd	I P-Mar-1
ľ	×	208	SUY	Coo Securities Australia Ltd	Į
14	1	40	Sug	Opp Securities Australia Ltd	1
1`	4	245	000	December Australia 14d	ı
ľ		335	kna	Opp parchives woshalla fro	ı
Ţ.	20	/20/	ouy.	LIBC Contribing Australia Ltd	16 147
1.	404	1 007	000	Do Cook and Doctoring 14d	ı
٠ľ٠	45	7 407	D. W.	1 IBS Securities Australia I to	16-Mar-11
٦,	- 12	200	D. T.	I BS Securities Ametralia I to	ı
5	1	CS.	Rivi	UBS Securities Australia I to	16-Mar-1
+		21	Виу	UBS Securities Australia Ltd	16-Mar-11
-1		5	Buy	UBS Securities Australia Ltd	16-Mar-11
-		5	Buy	UBS Securities Australia Ltd	16-Mar-1
-		5	Buy	IUBS Securities Australia Ltd	Γ
١~-		5	Buy	OBS Securities Australia Ltd	1-
1-		0	Puy	Opp Securities Australia Lto	O-IVIGIT-I
1-		1 u	Buy	Opp decolates Apparation to	
1		л.	Riv	I JBS Securities Australia I tri	16-Mar-11
÷►		21	Buv	UBS Securities Australia Ltd	16-Mar-11
4	4	227	Buy	UBS Securities Australia Ltd	16-Mar-11
w		15	Buy	UBS Securities Australia Ltd	16-Mar-11
ľ		10	YUB	OBS Securities Australia Ltd	O-MINI-1
14	1,23	Ogo,o	And And	Con Decorbes Magnetic City	70170
ار		6 600	Ding	I IDC Contribut Australia I to	is Mar 11
uli,		67	Biv	II IBS Securities Australia I to	G-Mar-11
v		65	RUB	JUBS Securities Australia Ltd	16-Mar-11
49	34	1,801	Виу	UBS Securities Australia Ltd	16-Mar-11
4		588	Виу	UBS Securities Australia Ltd	I-Mar-II
ł١		1,094	ouy.	Con accorde American Lin	- Glylar
واد		1004	D. I.	LIBS Securities Australia Ltd	16 Mar-11
mi.	,	8	(a)	UBS Securities Australia Ltd	16-Mar-11
O)	(91)	83	Sell	UBS Securities Australia Ltd	16-Mar-1
OV.	1	83	Виу	UBS Securities Australia Ltd	16-Mar-11
6		83	Buy	UBS Securities Australia Ltd	16-Mar-11
M C			7.03	IDC Constitute Australia 144	10 10 11
nI.		58	Riv	URS Securities Australia 1 td	6-Mar-11
_	9	315	Buy	UBS Securities Australia Ltd	16-Mar-11
5		206	Buy	UBS Securities Australia Ltd	I D-IVIAIT-I
14		1/0	yug	ODD DECUMES AUSUALIA LIO	C-MdI-1
1		170	0	I DC Counting Australia I td	16.11
		200	Auv	UBS Securities Australia Ltd	16-Mar-11
	124	640	Buy	UBS Securities Australia Ltd	16-Mar-11
			Sell	UBS Securities Australia Ltd	16-Mar-11
	1.		Buy	UBS Securities Australia Ltd	6-Mar-11
		777	kna	COD DECOMBES AUSTRALIA	ro-Ividi - i i
		200	D. C.	I IDC Cocyclinics Australia 14d	16 14 11
			Ring	TRY Sportition Australia I to	16-Mar-11
			Buv	JUBS Securities Australia Ltd	16-Mar-11
			Виу	UBS Securities Australia Ltd	16-Mar-11
		2,358	Buy	UBS Securities Australia Ltd	16-Mar-11
-			Buy	UBS Securities Australia Ltd	16-Mar-11
			Buy	UBS Securities Australia Ltd	b-Mar-11
			yus	CBO Securities Australia cita	- Violet
	Ī		100	I DC Con within Australia I to	16 14 2 1
.13			Pin	I IRC Contribles Australia Ltd	16.14.71
ŭ			Buy	UBS Securities Australia Ltd	16-Mar-11
Si.			Buy	UBS Securities Australia Ltd	16-Mar-11
Ó			Buy	UBS Securities Australia Ltd	16-Mar-11
ıκ	160	826	Buy	UBS Securities Australia Ltd	16-Mar-11
ıτ			yua	Obo Securities Australia Liu	O-Ividi*
ál٠			Diny.	COUNTRICATION ACCORDING THAT	120
ωĽ			H. W.	HIRC Securities Australia Ltd	16-Mar-11
ನ			Buy	UBS Securities Australia Ltd	16-Mar-11
×			NI BUY	UBS Securities Australia Ltd	16-Mar-11
ďΚ			And	UBS Securities Australia Liu	I DIVIDIT I
Jii)		1,101	Duy.	COD Secondes Australia Cia	C IVIdI - I
ńΙŹ				II DO Conseiting Australia 144	1000
4	144	743	Buy	UBS Securities Australia Ltd	16-Mar-11
ň		_	Buy	IUBS Securities Australia Ltd	16-Mar-11
12			виу	UBS Securities Australia Ltd	16-Mar-11
119			D. Way	Opposed rices Adobtes a con-	10 Ividia 1
A 1			Buv 8	UBS Securities Australia Ltd	16-Mar-11
X,			Виу	UBS Securities Australia Ltd	16-Mar-11
. **			Виу	UBS Securities Australia Ltd	16-Mar-11
ı×	(64	334	DE.	U65 Securities Australia Liu	l b-lvidi - 1 -
ĸ١۶	66	ZE.		THOC Committee Ametralia Htd	is Mar. 11
ایم	2	332	Buv	II IR< Securities Australia I td	16-Mar-11
5	2	332	Buy	II IRS Securities Australia I td	

17-Mar-11 17-Mar-11	-	17-Mar-11	- 1		- 1	-1	17-Mar-11	ı	1	16-Mar-11	IJ	16-Mar-11	-	16-Mar-11	1	16-Mar-11	16 Nor 11	o Mar II	16-Mar-13	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16 Mar 11	16-Mar-11	16 Mar 11	16 Mar 11	io-ividi- i	16 Mar-11	16 Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16 Mar 11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16 Mar 11	16 Mar 11	C VIGIT	16 Mar 11	I O-IVIGI - I I	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11	16-Mar-11
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia	LJBS Securities Australia	UBS Securities Australia	UBS AG, London Branc	UBS AG. London Branc	UBS Securities Australia	LIBS Securities Australia	UBS AG, London Branc	UBS Securities Australia	I BS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australi	UBS Securities Australia	URS Securities Australia	IBS Securities Australi	LIBS Socialities Australia	OBS Securities Australi	OBS Securities Australi	OBS Securities Australi	UBS Securities Austral	UBS Securities Austral	URS Securities Austral	LBS Securities Austral	UBS Securities Austral	LIBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Austral	UBS Securities Australia Ltd	UBS Securities Austral	LIBS Securities Austral	IBS Securities Austra	Discount call the Color	UBS Securities Austra	UBS Securities Australia Liu	UBS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	URS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra																
) Ltd		1	Ltd	7	5	ind.	TIG		eLtd	e Ltd	aLtd	e Ltd	aLtd	Ltd	al fa	2 0	100	artd	aLto	altid	a Ltd	altd	aLtd	altd	aLtd	altd	3 10 10	100	ario	a Lid	aLtd	e IId	aLtd	ia Ltd	ia Ltd	eltd	altd	ald	2 10 10	100	Par I CO	iaLtd	iaLtd	ia Ltd	ia Ctd	ia Ltd	10.00	a to	ia Ltd	III LTO	II LTC	la Ltd	ia Ltd	ia Ltd	lia Ltd	lia Ltd	lia Ltd	10 to	ia lad	10.00	la Ltd	ra Liu	lia Ltd	liaLtd	lia Ltd	lia Ltd	lia Ltd	lia Ltd	lia Ltd	lia Ltd	hi di	1 E	lia I td	lia I td	ial to
Sel Sel	95	202	S	BL	6	S in	Stock borrowed	Se	Bu	Bı	BL.	В	Bu	P	P.	2) B	Br.	B.	Bu	Se	В	В	2	BL	2 9	2	2	5	5	,	Bi	Bi	8	Bi	Ş	99	P	P 0	0.00	8	В	B	В	8	B 6	B	0 00	5			В	В	В	8			D 6	R				9			9	9	E							_
40,498 20,249				6.995 N	21 270			2		124			350		303		2,642		سا	Jy 366			JV 284				316			Jy 619		uyi 26					330	0		TY YU	Buy	uy 26	Buy 335	7000	57 uvi		7 U	Buy 27	eri	uy 14	uy.	J33	uy 6	Buy 8	uv e	3.5			142		7	1,200	Buy	juy 345		LLY 7	89	Buy 5	uy .	3UY 27	Buy	12	1,0	1.669	25
(7,788)					(800 (0)			(4,265)						Ī	Ī	Ī	512		7								08				(51)	51					648	-		15	2	3 52				2	7	53	(2)	28	2	5 65	7 13	2 16	19	0	48		3 44		1		3 18		-	7 15	133	2 10	5	53	17	24	2 2	324	36
Ordinary	Ordinary	Ordinary	Ordinary	Odinav	Ordinary	Crumary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ciciary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cromary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary

17-Mar-11 17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17.Mar. 11	17-Mar-11	17 Mar-11	17-Mar-11	1 /-IVIAI-1 I	17 Mar 11	1/-Mar-11	17-Mar-11	7-Mar-11	17-Mar-11	17.Mar-11	1 /-Mar-11	17-Mar-11	17-Mar-11	7-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17.Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17 Mar 11	17 Mar- 11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar-11	17-Mar 11	17-Mar-11	17 Mar. 11	17 NA 11														
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Itd	Obs securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia Ltd	IRS Securities Australia Ltd	I IBS Securities Australia Ltu	De contine Australia Ltd	UBS Securities Australia I to	Obsider Australia Ita	OBS Securities Australia Ltd	UBS Securities Australia Ltd	URS Securities Australia I to	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Secrities Vistralia III	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	UBS Securities Australia I td	URS Securities Australia Ital	UBS Securities Australia Lto	Opposed in the Second S	I inc Consistion Australia I to	OBS SECUTIVES AUSTRALIA																																						
B1	뫄	B	B	B	201	P 6	B	2 0		3				8	В	В	8	8	S	000		5		2 0		8	8	В	60	8	В	8	B		2 0		8	В	В	В	and on the contract of the con	900		В	В	В				B		В			В	8	2	8	9			2 0	Harris and the second s	8			D. C						
Buy 444	2		Buy 1.034		Buy 341		20	1,290	uy 4,582	903	Buy 206				Buy 2,988			ě		BIN 161	31	euy.		Buy 83		-	Buy 2,208				Buy 3.030		NI STAN	722	1bl	Suy	Buy 150	3uy 2,750	3uy 1,28	Juy	5 / y	1,523 3117 708	Xell 21	Buy 2	3uy 21:	Juy Zi		583 560	3,85	Buy 2,197		83.	Buy 2.497		Buy		Sell 83			BIN 457		4,572	Buy 4,57.	À		Buy 4.572		auy		30y /,090			
103	483	729	2007	847	66	107															(27)			16			427	7			586		10	45	3	1	29	532	249	w	104	157	(42)	4	42	5	7 6				101					275	(16)		886	288						886			354				
Ordinary	Ordinary	Ordinary	Ordinary	Ord Day	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	П	П	Ţ	Т	T	Т	T	Т	Г	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary		П		T	T	Ť	+	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	T	П	Ordinary	す	Ť	T	Ordinary	Ordinary	٦		7	T	Ţ	Т		٦	7	Ordinary	Ť	T	T	1	1	T	Ordinary

21-Mar-11 21-Mar-11	21-Mar-11	21-Mar-11	ZI-Mar-I	21-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar-11	20-Mar 11	20-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	18-Mar-11	10 Mg	7-Mar-11	17-Mar-11	/- V a -	I -Nar-I	1Mar-1 1	17-Mar-11	1/-Mar-11	7-Mar-11	17 Mar 11	17-Mar-11	17-IVIdI-11	17-Mar-11	17-Mar-11	17-Mar-11																														
UBS Securities Australia Ltd UBS Securities Australia Ltd	I BS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS AG, London Branch	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia I td	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia i td	UBS Securities Australia Ltd	LIBC COLLEGE ALIGHT LIG	UBS Securities Australia Ltd	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia I id	UBS AG. London Branch	UBS AG. Jandon Branch	UBS Securities Australia i to	LIBS Societies Australia 144	I IBS AG I andan Branch	I IBS Societies Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Obo Securities Australia Ltd	IBC Cocurities Australia Ind	LBS Securities Australia Ind	II IRS Securities Australia Ltd																																														
																				-00																						,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,																The second secon																			
Sell Sell	Sell	£.	Buy	Sell	Se Se	Se	Sell	Sell	Sell	(a)	Buv	Sell	Sel.	Sell	<u>6</u>	Sell	<u> </u>	¥	Buy	Sell	Sell	Buy	Buy	Buy	Buy	Buy.	Buy	Buy	D. C.	Buy	Buy	Виу	Виу	Buy	Виу	Buy	Buy	Bluv	Busy	Buy	KUY KUY	Виу	Buy	Stock borrowed	Stock borrowed	Sel	Buy	Sell	Buy	Ş.	Sell	Sell	R	Pi n	6 1	8 5	Ser	200	BUV	Buy	Buy	Buy	Buy	Buy	Buy	УПЯ	ВПА	Вцу	DUY	Bin	Rin						
8,505 17,850	5,938	14,081	9,918	2,551	4,229	14,260	305	505	500		500	1.182	2.967	1.799	6.091	777	50		2 108	36.	250,11	3,270	200		48,435	183,039	8,56,	8,56	419	3,21	13,100	4,62	4.679	2 77 7	27.74	4,02	1,49	4,679	4,62	2,77	823	4,68	15.770	7	1 61	211	2,82	262	4	N/A	₹.	26,36	41.	29	16	90	55	13	192	207,02	36.96	4 72	71,27	27.77	542	88	25	5,58	6,971	1,03	88	3,35	28	35	3c 67'	1 20	7.93
(1,620)	-		(1,885)			(2		(115)			(29)	(225)			(1.158)				(400)		(5)		(60)	(1)			(1,634)				2										157		3.000	5,1	3700		579	50			500,000			5 (57)		(18)			27		Ī	Ī			201			1	1,351				16				1 538
Ordinary Ordinary	Ordinary	Cromary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	- 1	1			1		Ordinary	Ordinary	Ordinary	Ordinary	Urdinary	Ordinary	Ordinary	Ordinaly	Ordinary	Ordinary																					

ſI	21-Mar-11	ιı							21-Mar-11	- 1	- 1	ì	21-Mar-11	21-Mar-11	21-Mar-11	21 Nor 11	21 - IVIdI - I	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-42-11	21-Mar-11	71-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Wall-11	21-Md1-11	21-Mar-11	71 Mar 11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11																
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ho	I IRS Securities Australia I to	LIBS COOKING AUSTRALIA LEO	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd		UBS Securities Australia Ltd	188 Sequifies Australia I to	UBC Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	USS Securities Australia Ltd	UBS Securities Australia Ltd	URS Securities Australia I to	UBS Securities Australia I td	I JBS Securities Australia I td	UBC Contribute Australia Ltd	I IBS Securities Australia I td	LIBS Securities Australia Ltd	Coo Securities Australia Lici	Obs Securities Australia Ltd	LIBS Consister Augustia Livi	Upo Securities Australia Ltd	LIBS Securities Australia Ltd	LIBS Securities Australia I td	UBS Securities Australia Ltd	LURS Securities Australia Ltd	luss Securities Australia Ltd																																																
Sel.	Sell	Self	Sell	Sell	\$.	Sel	Sel	Sell	Sell	Sel	Sell	Sel	Sell	Buy	Seli	145	6 8	5	Ser.	2.8	Sel	Sel	Sel	Sel	Sel	Set	Sel	Bu	Se	\$	200	Se Se	Se Se	Sel	Se) Sel	Sel	Se.	Se	\$ 1	Se	S &	Se Se	Se	Se	Se	Se	Se	45	Se	Şe	÷	Se	Se	8	Se	6	Selection	45	S	50	See See	6		5	6 14	50	68	Sell	Se	} Se							
14,102 18,916	27,489	410	21,515	4,736	2,063				1,087			10,763	3,255	1,733	5,250	5,938	3,255	24,932	14	819		1,470			2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2				11,320	2,168					331						18,115			17,635						1,601	-					609		24,707				9,497						15.	л.		2,100	3	12,2	150		5.560	-	1896'E
(2,686) (3,603)	(5,236)	(78)	(4,098)	(902)	(393)	(607)	(1,005)	(4,793)	(207)	(297)	(3,703)	(2.050)	(620)	(330)	(1,000)	(1,131)	(620)	(4,749)	(2.704)	(156)	156	(280)	(1000 1)	(156)	(200)	(5,139)	(169)	(657)	(2,152)	(413)	(4,587)	(2,495)	(4,345)	(655)	23	(63)	(50)	(1,000)	(1,261)	(295)	(3,444)	(345)	3	(3,359)	(1,000)	(2,130)	(201)	(4,80/)	(2,000)	(305)	(524)	(1,503)	(3,174)	(168)	(5,195)	(116)	(120)	(4,706)	(689)	(3)	(1.188)	(1.809)	(5,000)	(1,000)	(4 363)	(1,000)	(1 721)	(3)	(1)		(400)	(60)	(2,451)	(3/12)	127	(1.059)	(2)	(756)
Ordinary Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	T	T	1	T	T	Ordinary	Ordinary	Ordinary	Ordinary	Т	Т	Т	Т	Ŧ	1	1	Ordinary	П	П																						

21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	- IDIAI-12	2 I-Widi- I	2 -	- DIAI	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	- IDIAI- 17	21 1401-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	Z [-NIdI-]	21-Widt-11	21 Mar 11	21 1/2 11	21-Mdl-1	2 1-1DIVI-1	21-Mar-11	71-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	1 - IDIAI-17	1 - 101/1-17	21 Mar 11	21 May 11	21_Mar_11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	7 - MIN-1 - 1 - 1	21-Mar-11	21 May 11	31 Ad-
UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia Lt	ODO Secol ities Australia Li	Coo Securities Australia Li	Securities Australia Et	ODD DECUITIES AUSTIGIIA LI	I IBS Securities Australia II	UBS Securities Australia I t	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Ltd	JUBS Securities Australia Lt	ODO SECULINES AUSTRALIA	UBS Societies Australia L	1 III Securities Australia	I BS Securities Australia I	IBS Securities & Instrutia I	UBS Sequities Australia I	UBS Securities Australia Li	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia (UBS Securities Australia L	UBS Securities Australia L	OBS Securities Australia L	Coo Securities Australia	Obo Decourties Australia	Con recontres wordings	OBS Securities Australia Ltd	Obs Securities Australia	Upo Cod retor Australia I	LIBS Securities Australia Ltd	LIBS Securities Australia	IBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia I	UBS Securities Australia	UBS Securities Australia I	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia I	UBS Securities Australia l	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia (UBS Securities Australia I	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	OBS Securities Australia	De Contres Acondia	IDS Coordinates Australia III	I library of the Australia	I IBC Con letting A letteria	1	-	١	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	Cap Secontries Australia	IBS Securities Australia Ltd	5 6	D														
d	ď	d	G	io.	ā	d	16.	ĺ				Σ	5.16		E .	Ta.	ā			1	3 8	2 6	3 6	a.	ā	ã	ď.	to	td	îđ.	ī	td	itd	td	to	đ	. 6	100		10			16	ā		5 a	1 2	5 6	3.	đ	đ	rd	3	d	td	Ltd.	td	T	td	Ltd	ta	ta	id.	īđ	- E	Ltd	Ltd	LEG	Ltd.	r.Id	ē					1	E .	Etd	Ltd	Id	. 6	LEG.	Ltd		: [6		16.2
,	î						1	1																			200												•	30.1			1																	*											The same of the sa													1		-	
Self	Sell	Sell	Séll	Seli	Sell	Self	Sell	Sell	Sell	N.	Xell	Sell	5.00	€	Sell	Sell	Sell	YE!	Sell	Sel	2 2		c de	6 2	S 1	S	6	S :	8	Sp.	Sell	Se.	Sell	Sel	Sell	Sei	Sell	Se Se	×	¥.) ¥	Sell	Sel	Sell Sell	· W	C Sell	C Se	200	D. D.	<u>6</u>	2	6	6 2	6	8 5	ક ક	£ !	6 4	6	Se l	£	8	\$	Se	Sell	Sell	Sell	Sell	Sell	Sell.	¥.	×	18) Se	200	C L	<u>e</u> :	Se	Sell	Sell	Sell	Sell	Sell	YE I	Į.) Yell	, - I
4,9	5,2		15.2		9,442	15,5		2,1	2,100			717	34.		4	22,	6,5	10,	2		يد	1,0			5 250	10	10.1	10.	9	26,	3	23.	12,	14.	6	7,	ر ت	2.	16,	9,	1,	-	ı, çı	4,		4,4	-						٠ -			2	10	200	25			14	23	3		21	23	10	UT.	7		U	22		1,24	74.	17	24	28	27	24	25	27	2.1		U	
35 (940	250 (1,000)	5	25 (2.900	178 (186	142 (1,795	191 (2,96	289 (55	(515)		211	U		100			22,108 (4,211)						(000)			2501 (1,000)			19950				051 (4.399)											741 (1,284)		ľ	4,725		75000	1000	257 (25			7,600		(21	2 641	359 /3 49		767 (4 908		5 250 (1 000)				.449 (276				,250 (1,000)																394 (75)		110
)) Ordinary	T	٦)) Ordinary	il Ordinary	i) Ordinary	 Ordinary 	s) Ordinary	Ť	Г	(1) Ordinary	1) Ordinary	Ť	Ť	Ť	7			Г	Г	Т	Т	T	Т	Т	Т	Т	✝	7	T	T	7	Ť	٦	П	_					ł	1	ı		ı	1	1	T	Τ	Ordinary	Ordinary	炸	1	1	1	Ordinary	(A) Ordinary	Ordinary	1	Ī	Т	Ordinary	T	7	1	-	ؾ	٦	_	П		ı	Ţ	Г	Γ	Τ	Τ	(A) Ordinary	Τ					Γ	Γ	(5) Ordinary	Ţ	1 1 1

21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-MH-11	ZI-Mar-II	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21 Nar 11	21-Mar-11	∠1-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11
1 1	SBS	UBS Securities Australia Ltd	· UBS Securities Australia Ltd	LIBS Securities Australia Hu	UBS Securities Australia Ltd	UBS Securities Australia 11	IIBS Securities Australia I to	UBS Securities Australia Lt	OBS Securities Australia Lti	UBS Securities Australia Lt	UBS Securities Australia I t	LIBS Securities Australia I t	I IBS Securities Australia I t	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia II	I IBC Securities Australia L	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Lt	UBS Securities Australia I t	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia Li	UBS Securities Australia Ltd	- -	1	-	UBS Securities Australia Li	UBS Securities Australia Lt	UBS Securities Australia Ltd	UBS Securities Australia I to	OBS Securities Australia Li	UBS Securities Australia L	UBS Securides Australia L	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L		L	UBS Securities Australia L	UBS Securities Australia Ltd	UBS Securities Australia L d	UBS Securitie	UBS Securitie	UBS Securities	IUBS Securities Australia L																	
Ω		۵	0.30			d	a.	2		3 6					a	a.	O.	۵	۵		3 6	3.16			l a	ā	ď	6	ď.	ď.		1 6	1 2	i a	ā	Id	<u>а</u>	a	G.	o la		i a	ā	ta	đ	<u>ه</u> ا	6.18			ā	.6.	td	E	Tal.	1 6	ដី		ī. īā	ā	ō	td	đ	ď	řđ.	td	td	td	td	đ	4	D	- T	3 6	td.
Se Se	ng	Se	Se	Se	Se	Se	Sell	Se Se	ds.	3	50	Sel	Se	Se	Se	Sell	Se	Sel	Sell	Se	S .	Se Se	Sellac	Se	Se	Se	Se	Se	Se	S	200	Sell	Se	Şel	Se	Sel	Se	Se	Sell	56.34	2	25	SE	35	*	Se 50	Se	200	56	. Se	56	35	56 14	5		26	26	S	Se	8	54	\$	Se	Ş	Se	\$	-Se	Se	35	Sel	56	\$2	×0:	Se
18,47	388	38	285	3,76	18,35	1,551		-		22,750				25,746						23.771								9	394	94	1 37			6,983		17,897								3,974			3 3760									19,535		2,389			19,572								959			10,500		
(3,518)			(1,503)	Ţ	(3				T			(1,133)						(4.090)	(266)	_	(2,00)			(1,000)		8 (3,133)		3 (1.791)	(75)	2 2000	1761)	(1,000)				7 (3,409)		0 (3,779)		(1 220)		1 (2,343)			5 (2,100)			1 (1,122)	1 (265)			Ī	(4.801)					9 (455)											6 (125)			(2,000)		
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Collinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary

Sell 872 Sell 1,062 Sell 5,250 Sell 5,250
Sell
Sell 1,423
Sell 17,163
Sell 23,903
Sell 2,34/
Sell 714
Sell 19,535
Sell 5
Soll Sold Sold Sold Sold Sold Sold Sold
Self 10.670
5ell 7 860
Sell 592
340
7,002
Buy 592
Spill
Sell 3,757
2011
>p
Sell] 9,757
Sell 5240
_
Seil 18,34U
_
Selli 5,240
_
3eii 4,504
1,000
_
4,00
_
Sell 1.318
Sell 26,250
_
Sell 5,250
Sell 8,069

2121	2 2	1 2	:[2	2	21	21	21	21	21	21	21	2	2	21	2 !	21	71	91	21	J N	<u>-1</u> -	: ≥	12	21	21	21	21	얼	21	21	길	21	2	2]	21	<u>ગ</u>	2 ^	<u> </u>	<u> 1</u>	211	2	12	2:	2	2	[2]	121	인	ابح	NI:	717	्रा	না	সূদ	714	<u>, 1</u> √	2 2	N		1	n.	ī	2 N	JIN	۸اد	۸۱۲	1	<u> 1</u>	নাম	ন	আন	না	□	जा	N	- ਹ		72
	Mar-1	1	ı	İ	-Mar-11		H	Mar-11	-	1	ſ	Mar-11	Mar-11	Mar-11	Mar-11	Mar-11	Mar-11	Mar 11	M	-Widi-I	-Mar-11	Mar-11	-Mar-11	·Mar-11		-Mar-11	Mar-11	-Mar-11	1	-Mar-11	-Mar-11	1=	ı	-Mar-11	-Mar-11	-Mar-11	-Mar-11			-Mar-11	-Mar-11	-Mar-11	-Mar-1	-Mar-11	-Mar-11	-Mar-11	Mar 11	I-Mar-II	I-Mar-1	1-Mar-11	1-Mar-11	1-Mar-11	-Mar-11	1-Mar-11	-Mar-II	I-Mar-II	I-Ividi - I I	I-IVIdI-II	-ibivi-	I-IVIGI-II	-IVIdI-II	-IMAI-II	1-Mar-11	1-Mar-11	I-Mar-11	1-Mar-11	1-Mar-11	1-Mar-11	1-Mar-11	1-Mar-11	1-Mar-11							
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I Ird	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	LBS Securities Australia I to	IBS Societies Australia ted	IBS Securities Australia Ind	I IBS Specialities Australia Ltd	LIBS Cocurities Australia (10	ODS Securities Australia Ltd	UBS Securities Australia 1td	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia 1td	Upo securities Australia I td	Upp Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	ODS Securities Australia Ltd	COS Securities Australia LIG	Logo Securicies Adstralia Etg	Obs Securities Australia Ltd	ΉV	UBS Securities Australia Ltd	COO Secondies Australia I to	I BC Securities Australia I td	Securities Austra	UBS Securities Australia I Id	UBS Spourities Australia Ltd	UBS Securities Australia I td	UBS Securities Australia Ltd	IUBS Securities Australia Ltd	UBS Securities Australia Ltd																											
Buy Buy	Sell	Riv	Sell	Buy	Sp.	(4)	195	(a)	Sell Sell	Call	Sell	Sell Sell	Sell	Sell	Sell	Sell	Sell	Sel	Sell	Sell	llas	e/S	190	Se	Sell	Sell	Sell	Sall S	Sell	Sell Sell	Sell	Sell	Sell	Seil	Sell	Sell	Sell	Self	Sell	Sell	Sell	Sell	Sel. 20	Bin	Sell	Se	Sell	Sel	Se	Sell	Ser.	EUV	De la companya de la	Set	Se.	Co.	Bus	<u> </u>	Bira	45	Sel	Sel	Sel															
1,853 1,853	189	189	189	94	35	94	189	188	184	184	184	184	184	л 30	95	100	100	100	100	681	189	189	189	26	189	189	95	9,0	189	7.0	189	189	189	189	55	190	26												4.305		3 320					30,392								1,113	194	205			3,150	502	202					105	310	2,100
353 353 35	(36)	(36)	(36)	(18)	(18)	(18)	(36)	(36)	N.	(35)	25	35)	(34)	(6)	(10)	(36)	(36)	(30)	100	(36)	(36)	(36)	(36)	(18)	(36)	(36)	(18)	(18)	(35)	(18)	(36)	(36)	(36)	(36)	(18)	(30)	(36)	(36)	(36)	(5)	(13)	(18)	(18)	(81)	(18)	(10)	(1.937)	(2.500)	(820)	(100)	92.1	179	(871)	(1)	(634)	(5,789)	(6,146)	(4,270)	(1,876)	(5,789)	(128)	(634)	(5)	712)	(3/,	(45)	48	(20)	(603)	(39)	(30)	300	(48)	70,0	(48)	(20	(59)	(400
Ordinary Ordinary	Ordinary	Ordinary	Ordenay	Ordinary	Ordinary	Colling	Ordinary	Ordinary	Ordinaly	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinaly	Ordinary	П	Т	Ordinary	Т	Т	т	Т	Т	Т	П	г	Г		Г	Г	Г	Т	Т	Т	Ť	Т	Т	Ordinary	T	T	Т	T	1	1	Ordinary		Ť																		

21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	71-Mar-11	21 1/27 11	21-Mar-11	21 1/0/201-17	71 74 77	71.Mar-11	21-Mar-11	21-Mar-11	21-Mar 11	21-Mar-11	21 Mar-11	21-Mar-11	71-100-11	21-Mar-11	21-Wd1-11	21-Mar-11	ZI-Mar-I	21-Md-11	71 V ~ 11	2 - v q-	21 NAT 11	21-Mdr-11	21-Mdr-1	- IDIVI- 1 2	2 - V di -	21 Nd-11	21 - Wall - 1	21-Mar-1	21-Mar-11	ZI-Mar-I	21-Mar-1	21-Mar-11	21-Mar-1	21-Mar-11	21-Mar-11	21-Mar-1	21-Mar-1	21-Mar-11	21-Mar-1	21-Mar-11	21-Mar-1	71-Mar-1	21 Mar 1	21 Mar. 11	21-Mar-1	21 NH-1	21 - IVIGI - I	21-Mar-11	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-11	21-Mar-1	21-Mar-11	21-Mar-1	21-Mar-1	21-Mar-11	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-11	21-Mar-11	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-1	21-Mar-ı	1001	34 88
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	- 1	- 1	- 1		UBS Securities Australia Lio	ı	Upo Securities Australia Ltd	UBS Societies Australia Ltd	П	II BC Securities Australia Ltd	I IBC Contribes Australia 114	I IBS Societies Australia I to	Т		Libe Secretion Australia Lib	I IRS Securities Australia I to		Т			1		ı	CDS Securities Australia Lid	Coo Securities Australia Lio	os securites Australia Ltd	UBS Securities Australia Ltd	Obs securities Australia Ltd	OBS Securities Australia Ltd	Do Securities Australia Lic	OBS Securities Australia Lic	UBS Securities Australia Ltd		l	OBS Securities Australia Ltd			UBS Securities Australia Ltd	 UBS Securities Australia Ltd 	UBS Securities Australia Ltd	 ÚBS Securities Australia Lto 	 UBS Securities Australia Lto 	San		URS	LIBS Securities Australia Ltd	I CO	DR.	UBC Societies Australia Lit	1 libe constant Australia Li		UBS	UBS Securities Australia Lt	1 UBS Securities Australia Ltd	 UBS Securities Australia Lt 	 UBS Securities Australia Lt 	1 UBS Securities Australia Lt	ED	S	E	E G	8	JE J	BU	 UBS Securities Australia Lt 	 UBS Securities Australia Lt 	UB.	BU	UB.	 UBS Securities Australia Lt 	 UBS Securities Australia Lt 	 UBS Securities Australia Lt 	 UBS Securities Australia Lt 	1 UBS Securities Australia Ltd	1 UBS Securities Australia Lt	THE COUNTY AND A COUNTY OF	4 II IDC Com without A metaglig 14
																																								,		3		1		ď					2 6	2. 6							a		<u>a</u>	d	a.	<u>a</u>	d	d	d	д	d	d	d	ď	d	d	d	ď	d	<u>a</u>			7
Sell	£ 1	Selles	(a)	(A)	Sell	<u> </u>	Call Series	S ME	Xell Xell	Sell	Sell	- Seller	Sell	Self	Sell	ye.	S KE	Selle	- X	¥	×	E F	Sell	Sell	: E	×	Sell	виу	Sell	Sell	Виу	Sell	Sell	Sell	Sell	Buy	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sall	Sell Sell	S S	S 18	Sell	Sell	¥	Buy	Sell	Sell	Виу	Sell	Buy	Buy	Sell	Sell	Buy	Buy	Kng	Buy	Виу	Sell	Buy	Виу	Buy	Buy	Вшу	Виу	Buy	Виу	Buy	ywy	0
531	104	1204	40	100	189	242	,004	2,120	321	2,788	912	94	189	1,093	100	751	, 54	566	200	183	84	26/	183	31	215	990	430	540	\$40 1	16	16	540	367	16	555	555	2,179	1,027	42	515	346	1001	90	190	196	189	189	180	000	188	188	198	94	94	94	26	26	184	184	26	26	184	184	26	1,853	15,771	1,853	3,113	14,627	184	7,922	18,097	373	9,382	1,853	1,853	26	26	3,838	200,1	. 077
(101)	(75)	(10)	(10)	(16)	(35)	(04)	(202)	(403)	(6)	(531)	(174)	(18)	(36)	(323)	((61)	(29)	(81)	(108)	(10)	(35)	(16)	(51)	(35)	(6)	(41)	(189)	(82)	103	(103)	(3)	3	(103)	(70)	(3)	(106)	186	(415)	(196)	(8)	(98)	(66)	(19)	(17)	(36)	(36)	(36)	(35)	(35)	(30)	(36)	(36)	(36)	(18)	(18)	(18)) U	(5)	(35)	35	(5)	U1	35	(35)	(5)	353	3,004	353	593	2,786	(35)	1,509	3,447	71	1,787	353	353	5	5	731	פנינ	,
Ordinary	Ordinary	Ordinary	Oldinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	╗	T	Ordinary	٦	П	П	Ordinary	Urginary	1

21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Midi-11	- FIVI-17	2 - Vidi- -	21-WdI-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21 Mar 11	21-Wd-11	7 -IPIVI-I	21-IVId:-11	21-10-11	21 17 11	71-Mar-11	71-Mar-11	2 - Nor 11	2 I-IVIdI- I	21-IVI4-11	21-Midi-11	21-Mar-11	21-Mar-11	21-Mar-1	Z -Mar-11	21-Mar-11	21-Mar-11	- - - -	21-Mar-11	21-Mar-1	23-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	2 - Vidi-	2 -Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21 Mar-11	21-Mar-11	21-10101-11	21 NAT 11	21-Mar-11	21 -IVId: 11	21-Mar-11	21-Mar-11	7 - IBM-17	1 - JPIM-1 7	Z1-IVId1-11	2 I - IVIdI - I 1	71-Mar-17	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	2 I-IVI0I	21_l.dar_11
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Obs Securities Australia Ltd	UBS Securities Australia Ltd	Obs Securities Australia Ltd	Libe Southes Australia do	OBS Securities Australia Ltd	UBS Securities Australia Ltd	Obsided Mestalia Ltd	Obs Securities Australia Liu	I De Constitut Australia I td	Obs Securities Auditable Ltd	LIBS Societies Australia Ita	Cos securities Australia Lto	OBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	I BS Securities Australia Ltd	I IDS Securities Australia I td	OBS Securities Australia Ltd	JUBS Securities Australia Ltd	UBS Securities Australia Ltd		١	- 1	- 1	f	UBS Securities Australia I td	UBS Securities Australia Ltd	Obo Securities Australia Ltd	UBS Securities Australia Ltd	ı		ŀ	ı		ł	ı	П	L	UBS Securities Australia Ltd	L		l	UBS Securities Australia Ltd	l		l	H.	UBS Securities Australia Ltd	<u> </u>	Ę																			
															1		1																											- Wilder															700	70.																				
	Sell					Selli 2,1	Sell 310		Self	Sell 1,9	Sell 9	Sell 3,2	Sell 2,8	Sell	Sell 1,1	Sell 1	Sell			Sell		Sell	Sell	Sell 3,s	Sell 1,2	Sell	Sell	Sell 8,2			Sell			Sell 6.		Sell 6,4	Sell	Self	Sell 6	Sell	See See See See See See See See See See	64	Sell	- K	Sell Sell		Виу	Sell 2,0	Self 2,c	Buy	Sell	₹	Sell	Sell	File	Dell'	Sell	S &	8	Sell 1,	Sell	Sell		Sell 1,					Sell				Sell 1,			_				
,809 (344)		(1,094)						(420)	(2)	,994 (379)	105 (172)	(619)	193 (550)	21 (4)	1,129 (215)		w					358 (68)					558 (106)		53 (10)		589 (112))24 (765)		6,500 (1,238)		6,412 (1,219)		237 (45)	500 (114)	58 (11)	5/	(19)	422 (80)	148 (28)	148 (28)	490 (93)	ş	.082 (395)		290 55	290 (55)	(5)	290 (55)		250	(7)	(801)	833 (158)	32 (6)	323 (251)		306 (58)		375 (261)								621 (118)	082 (206)					184 (35)	(153)	
Ordinary	Ordinary	Ordinary	O CIT GILL	Ordinan	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Т	Г	П	П.	П	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	+	+	+	Ordinally	Ordinary	۲	Ť	T	Ordinary	-	-	Ordinary) Ordinary	Ordinary	Ordinary	_		_				П					Ordinary	Т	1																							

11	ı	ı	1	22-Mar-11	П	ı	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22 Mar 1	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11		22-Mar-11	Т	1	Т	22-Mar-1		ı	22-Mar-11		22-Mar-11	22-Mar-11		22-Mar-11	22-Mar-11	22-Mar-11	22-War-11	22 Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar- 1	22-Mar-11 22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-1	22-Mar-11	22-Mar-11	22-NIGH-11	22-Mar 11	22-Mar-11	27-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	21-Mar-11	121-Mar-11							
UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia Li	UBS Securities Australia Li	UBS Securities Australia Li	UBS Securities Australia Li	UBS Securities Australia Li	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia I	UBS Securities Australia Ltd	UBS Securities Australia C	LIBS Securities Australia	UBS Securities Australia I	UBS Securities Australia	UBS Securities Australia I	URS Securities Australia I	LIBS Securities Australia L	LIBS Contributes Australia I	LIBS Societies Australia	UBS Securities Australia I	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia I	UBS Securities Australia I	UBS Securities Australia I	I IRS Securities Australia I	UBS Securities Australia L	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	ÚBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia	LIBS Securities Australia L	LIBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	LIBS Securities Australia	I IBS Securities Australia Its	UBS AU, London Branch	UBS AG LONDO BOOCH	J DS Securities Australia	Upo Securities Australia Ltd	UBS Securities Australia	IRS Societies Australia	I BC Securities Australia I to	UBS Securities Australia	UBS Securities Australia L	IUBS Securities Australia L	IUBS Securities Australia L
12.12	td	đ	ā	td	id	đ	ī	đ.	5. 8	3,18	1 2	1 2	2 2	3 8	1	118		ila			10	đ	td	td	td	td	E .	a	td	d. la	1 2	1 6	ī	đ	ī	td	td	to l	6. 8	d. a			100	td	.td	td	td	đ	3 8	2 2	ž lie		id	.td	td	đ	a	.td	īđ	đ	ia	td	ic.	a	1	\$ 2	i	Į.	Σįε	1 6	i	1 5	ī	đ. là		td	-i
Buy Sel	Sel	Sel	Sel	Sel	Şel	Sel	Sel	<u> </u>	Sel Sel	Sel Sel	261	co se	Sel	\$ <u>6</u>	2	Sec	56	50	Se	3	Buy	Sel	Şel	Sel	Buy	Se	Se		₽	26	esc acc	52 8	Se	Sel	Sel	Se	Sell	F 19	Se	35	Ye Ye	Se	Se	Bu	Se	Sel	Se	Se	Se	200	90	Se Se	Bu	Sel	Se	Se	Эe	Se	Se	Ş ç	Bu	Sel	4	8	5 2	25	- Superior S	Se Se	50	D.	Slock returned	Section of the section	Stock borrows	Stock horrower	Se	Se Se	Ve.
388 419	388	3,329	10,076	5,320	1,553	9,650	8,586	2,120	2 129	3 937	1,589	1 200	9,420	0,000	335.0	OUN'S SC	174,792	37,347	57,347	3/,34/	14,604	14,604	1,348			37,923					-					972		329	8 900		9,02/				302					1,714					5,290					1,190		20.682		32 747						264,750				NIA	999	6	1.55
73 (79)	(73)	(627)	(1,894)	(1,000)	(292)	(1,814)	(1.614)	(400)	(140)	(7,0)	(261)	(505)	(1,764)	(441)	(1)	(100,000)	(32,/94)	/,00/	(/,00/)	(7,007)	2,740	(2,740)				(7,115)		410				(410)	(1,475)	(105)	(3,799)	(183)	(1,000)	(2,0,0)	(1676)	(52)	(1,700)	(4,963)	(219)	57	(57)	(1,000)	(57)	(2,000)	1,000	(324)	(000,1)	(430)	27	(52)	(1,000,1)	(27)	(800)	(27)	(342)	(225)		(2008 5/		(2027)	(287)	(1)				(50,000)	_		17.650	1 000 000	(190)		(162)
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ciciliary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary

[NIK	ı IN	ı Kı	Ical	NI	Nili	ıl K	al N	JA.	I K I	N H	7.11	N II	N 11	N 1	h il	N I	ь ıI	K iI	N all		. dr	n.	-	d s	d s	.7.	ı.	.14					.1.								1						_	_	_			_	_		_	_	_				_	_					_		_										
2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-1	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	2-Mar-11	22-Mar-11	2-Mar-11	2-Mar-11	22-Mar-11	Mar II	22-Mar-11	72-Mar-11	77 Nor 11	- IDIA1-7	22-Widi-11	22-WIDI-11	22-Midi-	- IDIAL-7	22-Mar-1	72-Mdi-13	- ININ-77	27 Mar 11	22-N/20-11	22 IVIGI - I	22-IVIG-	72-Mar-11	72-Mar-1	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar 1	22-MH-1	22-Midi-1	22-Mar-11	72-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-11	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-11	22-Mar-1	22-Mar-1	22-Mar-1	22-Mar-11	22-Mar-1	22-Mar-1	22-Mar-11	22-Mar-1	22-Mar-1	1-18M-77	1-15 N CC	1-JEW-CC	22-Mar-11
55	- E	E G	H	닏		1			Ш			1	-	S	١	١	딞	-	1	1	<u> </u>	5 5	ç	9			5	<u> </u>	ļ	Ę	2	- 5	2 5		15	S	<u> </u>	Ę.	IU.	JUI.	S	S	<u> </u>		2		2									1	l L	_	_	_ _	_	_		1	1		_	ļ		ļ	1	1	l	l	-	1	1		
S Securi	Secur	UBS Securities Australia Ltd	S Securi	S Securi	JBS Securities	SSECUL	IBS Securities	UBS Securities Australia Ltd	S Securi	S Securi	S Securi	Securi	S Securi	S Securi	S Securi	S Securi	S Securi	S Securi	Securi	S Securi	UBS Securities Australia I to	I Cocci	0 0000	o Secul	UBS Securities Australia Ltd	SOSCU	S Decur	S Secur	S Secur	ווייים כל מ	6	200	o cacal	INDEC CO	S Secur	so secur	S Securities	35 Secur	3S Secur	3S Secur	3S Secur	3S Secur	SSAUL	S Section 1	TO SECU	S S S S S S S S S S S S S S S S S S S	S Secur	S Secur	S Secur	S Secur	3S Secur	35 Secur	3S Secur	BS Secur	BS Secur	B§ Secur	BS Secur	BS Secur	BS Secur	BS Secur	BS Secur	BS Secur	BS Secur	BS Secu	BS Secui	BS Secu	BS Secu	BS Secui	BS Secu	UBS Securities Australia Ltd	BS Secu	BS Secu	BS Secu	BS Secu	BS Secu	BS Secu	I BS SECO	Des Ser	UBS Sacu
ties Australia	ties Aust	ies Aust	ties Aust	ties Ausi	ties Aus	LIES AUS	ties Aus	ties Aus	ties Ausi	ties Aust	ties Aust	ties Aust	ties Aus	ties Aus	ties Aus	ties Aus	ties Aus	SPA Set	ins Aus	ies Aus	ies Aus	LIES AUS	ries Aus	100	(les Aus	ues Aus	ues Aus	ues Aus	ues Aus	ues Aus		10.	CINY CALL	(les Aus	iles Aus	ities Aus	ities Aus	ities Aus	ities Aus	ities Aus	ities Aus	ities Aus	ijes Alis	lies Aus	tion Aus	Securities Austral	ties Aus	LIES AUS	ties Aus	ties Aus	ities Aus	itles Aus	ities Aus	ities Aus	ities Aus	ities Aus	ities Aus	ities Aus	ities Aus	itles Aus	ties Au	ities Aus	ities Aus	ities Aus	ities Aus	ities Au	ities Au	ities Au	ities Au	ities Au	ities Au	ities Au	rities Au	rities Au	rities Au	rities Au	* (5 2-)?	Thios AL	ritios AU
ralia Ltd	raliaLtd	ralia Ltd	ralia Ltd	ralia Ltd	Australia Ltd	ralia Luo	ralia Lto	ralia Ltd	ralia Ltd	ralia Ltd	ralia Ltd	ralia Ltd	ralia Ltd	iralia Ltd	ralia Lto	tralia Lto	ralia Ltc	ralia Ltc	ralia I to	rala i to	rala to	יים שו דור	did LL	משווש בנכ	ualla Lic	ualla Lic	tralla Etc	trana Ltc	rialia Ltc	וו פוופ רנכ	0 0		o Securities Australia Ltd	tidid Lit	ecurities Australia Ltd	ralla Ltc	tralia Ltc	Securities Australia Ltd	tralia Lto	tralia Ltc	tralia Lto	tralia Lto	Securities Australia Ltd	ralia tu	Australia Ltd	uala Li	וושוש רונ	tralia Lto	tralla Lto	Securities Australia Ltd	tralia Lto	tralia Lto	tralia Lto	5 Securities Australia Ltd	stralia Ltı	Securities Australia Ltd	tralia Lt	tralia Lt	stralia Lt	stralia Lt	tralia Lt	S Securities Australia Ltd	stralia Lt	stralia Lt	stralia Lt	stralia Lt	5 Securities Australia Ltd	stralia Lt	stralia Lt	stralia Lt	stralia Lt	stralia Lt	stralia Lt	stralia Lt	stralia Lt	S Securities Australia Ltd	olidia L	ralia F	ctraia
																						-							ľ								ľ	1					1								ľ		ľ	ľ	ď	٩	α,	٩		۵	۵	۵	٦	۵	d	۵	G.	đ	Д	۵	Ď.	ď	٦	ä,	ā	1	i	i	ì
								l																						-																		ĺ	١																													İ	
П					Ì	T						1	1		1	1	Ť	1	1			T	T	ľ	Ť	t	Ì	Ì	İ	İ	T	T	l	T	T	T					1	1	†	†	t	t	t		t	T	T	T		T	Ε,			1	1	1	1	1							F	-						t	†	t	1
																																			ĺ																																												
				Ì																																																																											
Sell Sell		Sell	Æ	ě	Sell la	<u> </u>	Sel	Sell	Вцу	BU _V	Sell	Sell	<u>6</u>	Sell	Sell	<u>≅</u>	Sell	€.	2 2	50	6 4	Ş Ç	Sel	Y.	18	ě	Yell	, FE	ř	Sen	190	1	Jeil.	ı K	ě	νel	₹	Se∥	Sell	≨	₹.	Sel	€ F	S PUY	Sell	<u>%</u>	¥	Sel.	8	SE SE	Sell	₩.	8	Sel	Sell	Sell	Se.	ξ	<u>و</u>	ξ	Buy	₽,	Đuy V	Sel	Buy	Buy	Buy	ě	Buy	Buy	Buy	Sell	ĕ	Sell	Sel	ĭ	2 1	S	େ
																																															ŀ																									2							
95 95	1,860	2,993	1,581	279	127	3 0	ı,	Ş	ų	υī	980	126	559	517	718	132	53	116	3	2 2	3 2	3 2	, ,] 	<u>-</u>	y	192	191	191	191	1	2 4	y o	3 6	ę G	190	95	190	190	ъ	ဖွ	190	3	<u> </u>	A Y	3	3 5	189	<u> </u>	8	121	69	190	190	190	190	190	74,792	<u>-</u> 8	37.347	37.347	37,347	14,604	14,604	14,604	48	133	37,923	37,923	592	37,923	67,000	26,650	53,300	16,015	1,620	1 (7)	A 779	325
																																											-																																				
(167)	(353)	(568)	(300)	(53)	(24)	(1)	Ξ	(1)	_		(186)	24	(106)	(98)	(136)	(25)	(10)	(22)	0	(10)	(30)	36	(0)	(3)	130	(8)	(30)	(36)	(36)	(50)	(00)	(30)	(8)	(81)	(8)	(36)	(18)	(36)	(36)	3	(17)	(36)	300	1361	(10)	(18)	(18)	(36)	(36)	(36)	(23)	(13)	(36)	(36)	(36)	(36)	(36)	32,794	199	(7,007)	7.007	7,007	2,740	(2,740)	2,740	9	25	(7,115)	7,115	111	7,115	(50,000)	(5,000,	(10,000	(3,016	(305)	SUC,	(900)	(73
Ordin	Ordir	Ordinary	Ordinary	Ordin	ordin	Orgi	Ordi	Ordir	Ordin	Ordin	Ordir	Ordin	Ord	Ordi	Ordi	Ordi	Ord	On		0 0	200	Old	Cidi	Cigi	Oral	Cra	Cia	Ord	Ordinary	0.00			C G	200	Cod	Ordi	Ord	Ordinary	Ord	oa	oa	Ordi		Т	Т	Т	Т	Τ	T	Т	Ť	T	og Og	Ord	Ord	Ord	Ord	Ord.	Ť	1	┪	┪	ᅥ	Ord	Ord	Ord	· Ord	Ť	Ord	Ord	т	~	Ĭ	Ē	Ť	Ť	T	Ť	200
ary	ally	ìаry	E L	nary .	al V		vary	ary	nary	Yarv.	VIE		N N	VIE	Ā	VIET	Jar.	y View		Tany	Tary	nany	rary	nary	ary	nary	nary	nary	nary	YNEIT	lal y	diy	nary	VIEW	nary	nary	nary	nary	nary	Vieu		narv		ridly	nary	nary	mary	nary	nary	Ordinary	inary	nary	nary	inary	inary	inary	mary	VIEU	Nem Ven	VIELL	VIEU	VIEU	nary	inary	inary	mary	inary	nary	inary	APU	inary	inary	inary	inary	inary	Ordinary	, lai	Mer	Ordinary

22-Mar-11		ı	l	l	- 1	1	22-Mar-11	ı	l	H	[3	Ш	22-Mar-11	77 May 11	22-Mar-11	77-Mar-11	22-Ividi - 11	22-Mar-11	22-Mar 11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Mar-11	22-Ividi-11	22-Mar-11	22-Mar 11	22-Mar-11																									
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Itd	OBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIRS Securities Australia Ltd	LIDS Secondes Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia ttd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	Obs Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Upo Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	Upo Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Sequities Australia Ind	UBS Securities Australia Ltd	UB5 Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Societies Australia Ltd	Upo Securates Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	
															1																											+														-															
Stock returned	Stock borrowed	Buy	Buy	Buy	Buy	Buy	Buy.	<u>%</u>	Sell	Sell	Sell	Sell	Selle	S 8	Sell	Sell	Sell	Sell	Sell	<u>%</u>	£ 5	2	¥E	Sell	Sell	Sell	Sell	<u>₹</u>	Buy	Sell	(less	Riv	S 25	BUY	BUY	Buy	Sell	Sell	Sell	Sell	56	Sall	Buy	Sell	Sell	Sell	Buy	Buy	Sall	Sell	Sell	Sell	Sell	Sel	Sell.	Sell	<u>₹</u>	Sell	Se	Sall	SET SET	Sell	Sell	Sell							
Z Z	N/A	1,223	5	1,757	58	500	725	2,185	26	693	3,854	185	7 265	2,190	2,915	1,014	269	169	370	290	106	777	3,131	449	1,104	2,566	58	1.758	222	222	222	777	777	69	69	375	69	375	69	375	1 077	964 6 07E	142	290	142	1,534	1,534	137	790	200	137	137	1.033	370	290	964	7,515	2,071	69	16	16	2.635	295	2.635	385	1 202	3,829	64 64	352	789	
(65,000)	500,000	232	-	334	11 2	g,	15.0	(413)	(5)	(131)	(730)	(35)	(1 376)	(414)	(551)	(192)	(51)	(32)	(70)	(55)	(121)	(42)	(593)	(85)	(209)	(486)	(11)	(333)	42	(42)	127)	(12)	(51)	13	13	71	(13)	(71)	(13)	(71)	(1,521)	183	27	(55)	(27)	(291)	291	76) C	(183)	(26)	(26)	(1961)	(167)	(55)	(183)	(1,426)	(393)	(13)	(3)	(i)	(500)	(95)	(500)	(F/)	(24)	(728)	(15)	(67)	(150)	
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cicling	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	-															

23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-1	23-Mar-11	23-Mar-11	73-Mar 11	22-Mar-11	23-Mar-11	22 May 11	23-Mar-1	23-Mar-11	23-Mar-11	23-Mar-11	ZS-Mar- I	23-Mar-11	23-Mar. 11	23-Mar-11	23-Mar 11	22 - Mar 11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	20 14101-11	Da.Mar. 11	23-Mar-11	11-JE/VT-2-C																																																			
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	URS Sequition Ametralia I to	LIBS Securities Australia Ltd	LIBS SOCIETION AUSTRALIA LIG	Opp Secondes Australia LIO	LIBS SOCIETION AUSTRALIA LEG	UBS Securities Australia Ltd	Los Securites Australia Lio	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia IId	UBS Securities Australia Ltd	LIBS Societies Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	Lips contributes Acceptable Ltd	LIBS Securities Australia Ltd	Obsider Australia Ltd	Coo Seculties Australia Ltd	UBS Securities Australia Ltd	JUBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Con Secondary Australia Ltd	LIBS Securities Australia Ltd	I IBS Soo without Australia I to	II IDC Committee Ametralia I to																			
Bu	Ang	Bu	Bu	Bu	Bu	Bu	Bu	Sell	Bu	BU.	Bu	Ви	Bu	Se	Bu	\$	Bu	Bil	Bi		80	000	Bu	BU		Bu	Buy	Bu	Ви	Bu	Buy	Bu	Bu	Buy.	p. 92	0.00	50	Bu	Buy	BL	BL	Buy	. Bu	80		P 5	B	P. D.	B. B.	5	Виу	200	5	В		Buy		81	Buy	Bu	Şel	Buy	Bu	Bu	Bı	Bu	Bı		Buy	B	D.								
y 3,425 y 491		3		y 16	_	147		1	W 84	47	1.306	:1	y 147	. 1,:		11 84	1,881		y 647		1		110		500				7		94		424	77.04	1,0		Jy 152		y 162		158	31	176	79	Jy 835		Jy 489			Jy 768	JV 84	4		ار د	2673 767	2,		189	13,587	105	3/,5	2	ell 95						ell 290				uy 559			21	24	21	202
(648) 93	648	648	1	w	24	28	16	(28)	16	اه	248	248	28	(248)	56	(16)	357	16	123	104	227	2	21	150	100	7,0	2	1	4	_	18	22	81	32	197		29	70	31	1	U.E.	9	7,4	2,1	159	25	£6	1	104	146	16	914	91	736	1.46	438	170	36	2,593	20	7,064	560	(18))								106					40	
Ordinary	Ordinary	Ordinary		П		П	П	Т	Т	Ţ	Т	T	╗	П			Ordinary	Ordinally	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	П	П		┪	Ordinary	Ordinary	Ordinary	Ť	t	Ordinary	T	┢	Ordinary	Ordinary	Ordinary	Ordinary	†	†	T	Ordinary	T	T	1	t	T	t	t	Г						П		7	П		Ordinary	П	Ī	t	t	Ť									

ונקומו	NI N	जार	il.	ı I N	Z	2	NJ.	N	ा	NI	N)	N)	INI	INI	Iki	INI	II.	In	ž.	IN	IST	r _{s i}	N 1	NII	I	N d	h .I		s all	. Tr		. 1	h .I	s J	1		1	ır		ı,																																			• •				
3-Mar-11 3-Mar-11	3-Mar-1	3-Mar-11	Mar-I	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	:3-Mar-11	3-Mar-11	23-Mar-11	23-Mar-11	3-Mar-11	3-Mar-11	3-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	11-ipral-02	25-Mar-1	23-Mar-11	25-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-IVIAT-11	23-Mar-11	23-Mar-11	23 MAY 11	23-Mar-11	23-Mar-11	23-Mar. 11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11																														
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UB\$ Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	URS Securities Australia Ltd	LIBC Securities Australia Ltd	I BS Securities Australia Ltd	Obo Securities Australia I to	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIDS Securities Australia Ltd	LIBS Societies Australia Ltd	I DC Cocuring Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	De Contition Australia 14d	I BC Securities Australia Ltd	I IRS Securities Australia I td	LIBS Sprintips Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd		UBS Securities Australia Etd
Buy Buy	Виу	Buy	Buy	Buy	Buy	Buy	Sell	Ang	Sell	Ang	θυν	Sell	Buy	Buy	Buy	Виу	Виу	Виу	Buy	Buy	Виу	Buy	Buy	Buy	Buy	Bluv	BUV	BLW	Buy	Buo	Buv	Buy	Buv	Buy	Buy	Buy	BUV	But	eu)		50	lug u	Buy	Buy	Buy	Buy	Sell	Buy	Se	Buy	New Year	500		Buy	Bu	Se	Bu	Bu	Bu	Se	Bu	Bu	Bu	Bu	Bu	Buy	Bu	Bu	Se Se	50	Bu	000	B	Bull	Bio	Bi	Bu	Виу	Bu
367 534	246	346	189	457	876	91	1,621	36	16			876			37		2	47			241					507			418			277	0	11	21	2.		70	ب						1,288		222			1 394						766				1,612		-				127				y 634								y 540	
70 102	9	66	36	87	167	w	(309)	4	(3)	309	60E	(167)	167	1	7	2	53	9	52	5	46	17	51	9	27	815	Q2	99	8 2	77	57	23	1.199	2	4	46	_ ;	17	3 8	(5)	,5	100	42	13	244	10	(42)	184	(264)	264	18(1)	100	3 2	205	550	(145)	308	310	145	(305)	145	305	4	24	24	24	6	120	020	07.1	56	200	200	350	727	707	5.1	102	60
Ordinary Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Т	Т	T	Т				ı		- 1	- 11			Ordinary			1	1-	Т	т	Т	г	Т	г		г	_		Н	-		г	Т	_	_	_	Τ	Т	T	T	7	Ordinary	П																

24-Mar-11	1	24-Mar-11	ı	1	24-Mar-11	l	l		24-Mar-11			24-Mar-11	_	24-Mar-11	- 1	24-Mar-11	=	3	24-Mar-11	24-Mar- 11	24-Mar-11	24-Wd-11	74-MH-11	1 - IPIA-4-7	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar 11	23-Mar 11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar-11	23-War-11	23-Mar-11	23-Mar-11	23-Mar-11	23-Mar 11	23-Mar-11	23-Mar-11	23.Mar.11	23 Mar 11	23-Mar-11	73 May 11	23-Mar-11	23-Mar-11	73-MBI-11	23 Mar	23-Mar-11	23-IVIAT-11	23-Mar-11	23-Wdr-11	1 - IPIN-67	23-Mar 11	70 Mar 11	22-Mar-11										
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ind	LIBS Securities Australia Itd	LIRS Securities Australia Ltd	I IBS Securities Australia I to	I DS Securities Australia Ltd	Coo Securities Australia Lia	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	Coo Securities Australia Cio	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LBS Securities Australia Hd	Ups Securities Australia Ltd	IDBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	LIBS Securities Australia I to	UBS Securities Australia Ltd	LIRS Securities Australia I td	I IRK Securities Australia I to	UBS Securities Australia 14d	LIBS Securities Australia Hd	On Decolines Australia Ltd	URS Securities Australia IId	USS Securities Australia Lig	UBS Securities Australia Lto	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	OBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Lid	Cos securities Australia Lio	OBS Securities Australia Ltd	I IBS Securities Australia I to	I DC Committies Australia [16]
						Vera										100															i																															-																
Sell	Buy	Sell.	Sell	Sell	Sell	Sell	Sell	Sel	Sell	Sell	Sell	Sell	Sell	Buv	Sell	ક	8	Sell	Sell	ouy ouy	Buy	Виу	Виу	Buy	Sell	Buy	Sell	Buy	Sell	Sell	Stock returned	Stock returned	Stock borrowed	Buy	Buy	D. D.	2 16	Sell	Вчу	Buy	Вцу	Виу	S I	SPI SPI	Buy	Buy	Виу	Buy	Buy	Buy	Buy	Buy	Buy.	виу	Виу	Buy	Sel.	Bird	Ruy	Birt	Sally	buy	P. W.	700	Buy	BUY	Seil	Buy	Buy	Sell	Вшу	Sell	Buy	Виу	BUY	yell	eally	pl
6,490	521,000	3,319	3,934	83	10	750	146	141	4,376	229	761	229	662	521,000	14 400	COO SE	8,060	20,004	7 17,17	74,412	74,412	2,/56	377		· ·	5	8,422	29,911	15,815	1,461,124	N/A	N/A	WA	137	1	, Year	369	767	132	105	767	631	2 4 70	122	854	132	767	2,356	2,470	2,470	73	4,857	593	1,029	74	793	7,504	7507	73	1 100	110	110	25/	/54	430	121	430	1,923	430	734	/3	/3	299	121	/34	121	2/2	מפר
(1,248)	100,000	(637)	(755)	(16)	(2)	(144)	(28)	(27)	(840)	(44)	(146)	(44)	(127)	00000	(2/2/2)	7 7 7 7	1 517	(0,2,0)	(14,070)	(4,393	14,393	533	73	<u> </u>	(1)	1	(1,617)	5,779	(3,039)	(280,000)	(100,000)	(500,000)	200.000	36	2 4	, è	(70)	(146)	25	20	146	120	(107.17)	07/	162	25	146	448	470	470	14	927	113	196	14	151	/1/4	777	203	(22)	72	2 2	42	140	8	23	(82)	367	82	(140)	14	(14)	5/	23	140	(23)	CC (CC)	-
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cromary	Ciuliary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	> 1:

24-Mar-11 24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24 Mar 1	24-h/ar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-I	24-Mdr-11	24-War-11	Z4-[Vid]-	24 Mar 11	24-Mar. 11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24_Mar_11	24-Mar-11	24_Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	74-Mar-11	74.Mar. 11	74-Mar-11	24-Mar 11	24-Mar-11	24-Mar-	1 - IDIA - 1-7	24-MIN-1-1	11 - IDIAL 447	24. Mar. 11	24-Mar-11	24-Mar-11	24 Mar 11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	11-INIV-17	24 W.	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11																							
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Coo occurres Australia Liu	I BC Consisting Australia I to	I BS Sen intres Australia I to	UBS Securities Australia Itd	UBS Securities Australia Ito	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Obo Securities Australia cto	TIDE CONTINUES AUSTRALIA LIO	Life Secretary August 150	1 IRS Securities Australia Irs	IRS Securities Australia Ital	IRS Sporting Australia Itd	LIBS Securities Australia Ltd	II BS Securities Arretralia I td	I BS Securities Australia I td	LIBS Securities Australia I to	LIBS Securities Australia Ltd	I IBS Securities Australia I to	IBS Secritives Allegand CIO	IBS Securities Australia III	I De con rete d'instrain 114	I IBC Cocurities Australia Ltd	IRS Securities Australia Ind	UBS Securities Australia I td	UBS Securities Australia I to	UBS Securities Australia I to	IRS Securities Australia 11d	I IBS Securities Australia 114	I IBC Contribes Australia Ltd	Obs Securities Australia Lid	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	Cop Sectiones Adstralia Fig.	ODS SECURIES AUSTRAIN LTG	COD DECOMES AUSTRALIA ETC	I III Constitute Australia I to	IBS Securities Australia Itd	I IRS Securities Australia Itd	UBS Securities Australia I Id	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Cap Securities Australia Ltd	lo	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd
																										110000									70.10																																												70000					
Sell Sell	Sell	Sell	Sell	Sell.	Jell.	Sell	Call	Sell I	8	Self	Sel	Sell	Sell	Sell	Sell	Sell	Sell	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	Dell'	000	O E	CON CONTRACTOR	coll Jell	Coll	Sell	Sall	Jell Jell	Dell Dell	Sell Sell	Self	Cell	UE:	Call	Sell Sell	Sell	S I	SAII	Sel Sel	<u> </u>	Sell	Sell	Sell	Sell	Sell	Self	Sell	Sell	Sell	Sell	Sell	<u>%</u>	- Se	Se	Sel. 1	E S	Sell	2	555	200	6	ક	Sell	Sell	Sell	Sell	<u>sel</u>) ¥	Sel.	Sall	Sell	Sell	Sell												
724 (140) 2,585 (500)	620 (120							Ī	558 (100	120 053					2,859 (552)			5,6/2 [1,288	3 8	144			4,545			T	20,3/3 (3,933		2,0/2 (400)				ļ	1000	Ī	Ţ	1300	888	595 (114)		7,015 (1,349		(19)	94 (18)		ĺ			10,910 (2,098)	7		177	796	7	88 (17)		3,042 (585)			9,402 (1.80				8,388 (1,613)		4,/11 (90		18,398 (3,538)		101	I				5,418 (1,042)						1 144 (220)	83		(1
$\neg \neg$	Ordinary	╗	П	Г	Т	T	۲	T	Ť	Ť	T	7	_	_	_	_	ř	Ť	Ť	Ť	1	Ţ	1	Ť	T	Ť	1	T	Ordinary	T	Т	Т	T	Т	1	T	1	Ť	Ť		Ĩ)		٦	Ī	7	_	7	7	1	1	3) Ordinary	Ordinary	7	Ť	T	5) Ordinary	T	Т	T	٦	7	٦	-	Ť	(6) Ordinary	1	Ť	T	Old lidy	Т	Т	T	T	2) Ordinary	٦	П		Ī	T	T	6) Ordinary	1	

24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24 Mar 11	24-Mar-11	24-Mdi-	24-IVIdI-1	24 Nat 11	24-Mdr-1	24-Mar- 1	24-Mar-13	24-Mar-11 24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	74-Midi-1	24-Mar-11	24-IVIdI- I I	24-Mar-11	24-Mar-	24-Mar-11	24-Md-11	24-Mar-11	7.4.Mar.11																		
UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIRS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia (1d	UBS Sequences Australia I to	LIBS Securities Australia Ltd	URS Securities Australia I to	LIBS Secondes Australia Lto	Cooperatives Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LBS Securities Austrana Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Sequities Australia Ind	LIBS Sequities Australia Ind	I BS Securities Australia Ltd	I IBS Securities Australia I to	Obo Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Obsideration Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia III	LIBS Societies Australia Ltd	I BS Securities Australia I to	LIBS Sociation Autoria 140	Upo Secondes Australia Ltu	I BS Sociation Australia Ltd	TIBE COLUMN AUSUALIA LLO	UBS Securities Australia Ltd	Obo Securaties Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltc	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBS Securities A state and Lic	I DC Coording Australia He	II DC Committee Australia I to
													-	-																																																																	
Se	Sell	26	Se Se	6	*	dS.	92	8	\$6	ac ac	Sec.	Se	Se	Se Se	33.	Se	Se	9	Se	6	6	Sell	36		Z. Z.	ye.	26	Se	56	Se	Se	Sel	S .	8	Riv	oc a	, see	Se	Se	Se	Se	Se.	Sel	56	56	2	×1	2		Se Se	יות ע	Se Se	Sel.		5		Se	Se Se	· · ·	Se	S	\$	S	Š	Ş	Se	Ş	2	Se) Y		c.							
15,783	3,207									341		284	429		6,7						15,510				15.510				250		200		408		269		393		403		35.5		5,138				372		159	27/	37/	N .	6.199	796	626	707	36,		3,945	,		3	/8/			83		-	JI 79i	88	12		4	253	11 2,	212			
(3,041)	0 8	(550,51)		(1)	(1.964)	04349			(10)	(55)	(76)	(55)	(83)	(61)	(1,307)	(147)	(1,772)	(2,000)	(52)	(16)	(3,0				(3,000)		(1 A)	(77)	(5/3)		(0/)	(/6)				(53)	(76)	(72)	(78)	(74)	(46)	14,090/	(00/)	(14,393)		(17)	(72)	(120)	(30)	(53)	(72)	(50)	(1 100)	(154)	(171)	(/	(70)	(1)	(/65)			(194)				(16)			()			3 (193)		(49)		2		101	14.51
Ordinary	Ordinary	Ordinary	Ordinary	Ordinaly	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cidinal	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Urdinary)						

25-Mar-11 25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	S-Mar-11	25-Mar-11	25-Mar-1	25-Mar-11	25.Mar. 1	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25 Mar 11	Z5-Mar-11	25-Mar-11	75 MAY 11	25-Mar-1]	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	25-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-1	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11	24-Mar-11																													
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	I IBC Securities Australia to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia I Id	UBS Securities Australia Ltd	UBS Securities Australia Ltd	U85 Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG. London Branch	ILIBS Securities Australia Ltd	11 BC Securities Australia I td	OBS Securities Australia 14d	UBS Securities Australia Ltd	CBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	LIBS COST FEEDS AUSTRALIA LIG	LIBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd
Sell	Sell	Sell	Sell	Sel	Sel Sel	Sel	Sell	Sel	Sel Sel	Sel	Sell	Sel	Buy	Sell	Sel	Se	Sel	Sel	95	26	URI	Sell	Sel	Sel	Sel	Se	Se	de Se	28	Sell	Sel	Se	Se	Se	Sel	26	26	. *	Se	Se	Se	Se	3	Sea	8 X		Se	Se	Se	Buy	Bu	Buy	95	35	B	6	Ç.	56			Sell	Se	36		Se	56	Se	Se	2 %	5	\$	- Se		56
1,582	291	18,203	3,798	4,978	7,967			2,70	5,0		1,391			1,132				-	3320	1084					598				2/5		29	48,086			1 402							2,116				300				15,354								3,753			142,641			22,436			ell 15,783						all 10,966	
(10) Ordinary (299) Ordinary (52) Ordinary	(55) Ordina	(3,441) Ordina	<u></u>	Т	(1,506) Ordin	П	(91) Ordin	(275) (275)	T	Ť	(263) Ordin	Ť.	П	۲	(52) Ordi	7	(7) Ordin	(309) (309)	(561) Ordinary	T	T	(874) Ordi	Ē	(1,360) Ordi	(113) Ordi	=	(107) Ordi	(17) Ord	(52) Ordi	(2) Ord	(5,503) Ordinary	(9,090); Ord		(265) Ordi	7 0	(265) Ord	Ť	Ť	(18) Ord	Г		(400) Ord	Т	(294) Ordinary	T	(18) Ord	T	Ť	Ē	2,908 Ord	ı	5,702 Ord	Т	Ī	Т	Т	Т	(1,849) Ordinary	Τ	Ť	Г	Τ	(5,618) Orc	T	ľ	Г	Г	-	Ť	(CI)	Т		(2,113) On	П

29-Mar-11	29-Mar-11	29-Mar-11	29-WdF-11	70.Mar.11	29-Mar-11	29-Mar 11	29-Mar-1	29-Mar-11	29 Mar 11	20 Mar 11	20 May 11	29-Mar-11	29-Mar-11	11-8M-67	29-Mar-11	29-Mar-11	29-Mar-11	29-Mar 11	29-Mar-11	29-Mar-11	29-Mar-11	29-Mar-11	28-Mar-11	28-Mar-11	28-Mar-11	28-Mar-11	28-Mar-11	28-Mar-11	28-Mar-11	28-Mar-11	28-Mar-11	28 Mar 11	28-Mar-11	78-Mar-11	28 Mar 11	28-Mar-11	28-Mar	28-Mar 11	20-IVIdI - I I	2)- Vidi-	27-Mar-11	27 Mar 11	27-Mdr-11	27-Mar-11	25-Mar-11	25 Mar 11	25 Mar. 11	25-Mar-11	75-Mar-11																														
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	I BS Sequettion Alestralia I to	LIBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ind	LIBS Securities Australia Hd	TIBS Societies Australia (10	Operation Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS securities Australia Ltd	U85 Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	UBS Securities Australia I to	I IRS Securities Australia Ltd	UBS AG, LONGON BRANCH	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia (1d	UBS Securities Australia I td	IRS Securities Australia I td	I BS Securities Australia I to	UBS Securities Australia Ltd	JBS Securities Australia Itd	IIBS AG London Branch	I BS AC LORGO BETTOO	OBS Securities Australia Ltd	IBS Securities Australia 142	URS Societies Australia Hd	I IBS Societies Australia I to	IRS Securities Australia Ital	LISC ACT LOCACO BETTER	Obs Securities Australia Ltd	IBS Sen rities Australia I to	UBS Securities Australia I tri	II JRS Socurities Australia Ltd
																																	,																																		,		7.000				1						-
Sell	<u>श्</u>	Sell	Sell	Sell	Sell	Sell	Selli	Sell	Self	Self	Sell	Sell	Sell		Sell 1			1	Sell		Sell 1	Sell	Sell	Sell	Sell	Sell	Se	<u>6 }</u>	Sell	Sell		Ser		Sell	Sell	Sell	Sell	Sell	Self 2	Sell	Sell.	Selli	8	Sell	Sell	BUY)ell			Buy			1		Sell	Sell	Sell	Sell	Sell	Selles	\$A	So.	Sel	Sell	Soli	C6 II	Sall Sall	Buy				Sell					CALL	1 Let 1	Calif
95 (18)		(16)	95 (18)	446 (85)		42 (8)	310 (59)	95 (18)	45.2 (86)	487 (93)		(2)			.601 (305)				74 (14)	131 (25)	,853 (353)	373 (71)	483 (92)	5 (1)	84 (16)	79 (15)	105	A7 (0)	221 (42)	1,738 (331)	2,515 (4/9)	(317)	100 (19)			3,506 (669)	84 (16)	205 (39)	4,053 (772)	304 (58)	5 (1)	73 (14)	645 (123)	(100)		134	26 (5)	1,5	19,786 (3,774)	6,438 1,227		0,664 (93,458)		1,281 (245)			760 (145)				1887 (801)		(91)	100	(2)	37 (7)	110,4	2,717	07,972 (58,649)	947 (179)	942 (1/8)		15,325 (2,908)		(2,728)		(01)		(10)
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary			Ordinary	Ordinary	T	T	1	T	T	1	Cidilaly	Ordinary	Ordinary	Clustery	Ordinary	V III I	Ordinary	Ordinary	Ordinary	Ť	T	Ordinary	П	ı	Ordinary	Т	Ordinaly	O-dinan,

	30-Mar-11	-[- 1	- 1		30-Mar-11	30-Mar-11	П	30-Mar-11				l	ı	L	50-Mar-1	ı	DO-Mai-	ſ		1	1	SO Mar 11	1-	1	1-	30-Mar-II	-	ı	SO-Mar-11	30-Mar-11	SO-Mar-		30-Mar-11	30-Mar-11		Ш	30-Mar-11	- 1	30-Mar-11	ŧ		30-Mar-11	OC MAC	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	29 Mar-11	29-Mar-11	29-Mar-11	29-Mar-11	29 Mar-11	20-Mar-11	11-18M-67	20 Var 1	29-Mar-11	29-Mar-11	29-Mar-11	29-Mar-11	29-Mar-11	11-IDIAI-67	29-Mar 11	1 - IPIAI - 67	20 Vol. 1	20 Ver 11	20 7 1	29-Mar-11	79-Mar-11	29-Mar-11	29 Mar-11	29-Mar-11	29-Mar-11	29-Mar-11	29-Mar-11	29-Mar-11
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Hd	LIBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JUBS Securities Australia Ltd	OBS Securities Australia Ltd	ODO DECUTIVES AUSTRANA LICI	OD SECURIES AUSTRALIA LIG	DBC Securities Australia Ltd	LIBC Cod within Australia Ltd	OBS Securities Australia Ltd	USS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ctd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	U85 Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIBS Securities Australia Ltd	LIBS Securities Australia Ltd	LIBS Securities Australia 14d	Obs Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG, London Branch	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	LIBS AG London Branch	UBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia I td	Lift's Securities Australia Ltd	Upo Securities Australia Lto	UBS Securities Australia Ltd	CBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	CDS Securities Australia Ltd	Coo Securities Australia Ltd	OBS Securities Australia Lto	OBS Securiles Australia Ltd	CDS Security Australia Etg	OGS SECURIES AUSURIA CIG	UBS Securities Australia Ltd	IIRS Securities Abstralia Ltd	UBS Securities Australia Itcl	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd
Buy	Buy	Britis	Din.	Bin	Biv	BUV	Buy	Buy	Buy	Buy	Виу	Ang.	Sell	Buy	Виу	Buy	Buy	Buy	Buy.	Buy	RIJ.	виу		Bus	Buy	Sel	Buy	Buj	Ви	Buy	Sel	Sel	Sel	Sell	Se	Se	Se	Se	Se	Se	\$	Sec.	ec.	See See See See See See See See See See	Se Se	*	Sel	Se	Bu	Se	Bu.	8	Bull	39	Sala	95			Se	Se)e	Sell	Se	Se	ž.	Se)e	×	Se Se	Ye Ye	See See See See See See See See See See	36	50	50	Rity		6 9	aş	Sel	Se	8.
174				100	7	1 205				1.131						37		205			121				210				53		189	ii 3,222			5,290	138	5,322	2,619	7.998	148	10,000	148	5,105	1/,923	243	804	667	804		4.447	13.658	975 107	13 151					115		95	5	9,319		21	445	3						10	Tropin		7 143		5	,	2,143		,
33	404	19	100		222	229	42	279	234	215	175	59	(39)	w	95	7	34	39	33	109	23	138	10	124	40	(124)	40	30	10	40	(36)	(609)	(28)	(1,483)	(1,000)	(26)	(300,1)	(495)	(1.512)	(86)	(457'1)	(28)	(265)	(3,388)	(46)	(152)	(126)	(152)	ر ا	(845)	7 505	2,500	3 500	(8/1)	(194)	(104)	(23)	(22)	(1)	(18)	(1)	(1,775)	(231)	(4)	(85)	(737)	(236)		ĺ		(294)		(408)	409	1343	ייין ייין	102	(103	(409)	(18	18
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Colling	Ordinary	Ordinary	Ordinany	Ordinary	Ordinary	Ordina	Ordnan	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	╗	╗	7	┰	т	Т	Т	Т	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	┰	┰	т	Т	7	┰	Ordinary	т	Т	_	F	Ordinary	Ordinary	Ordinary				П	П	ř	Ordinary	Ordinary	Ordinary	+	÷	Ŧ	~	1	Ordinary	T	

31-Mar-11	Mar. 1	31 Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	31-Mar-11	30-Mar-11	30-Mar 1	SO-Mar-1	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-1	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30-Mar-11	30 No. 11	30-Mar-11	DIAL-OC																															
UBS Securities Australia Ltd	UBS Securities Australia Hd	I BS Securities Australia Ltd	LIBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Segrifies Australia I td	LIBS Securities Australia Ltd	COURSE VISITED IN	LIBS Sources Australia Ltd	UBS Securities Australia Ltd	JUBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Etd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Its	UBS Securities Australia Ltd	I IBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I to	UBS Securities Australia Ltd	J IRS Securities Australia Ita	LIBS Securities Australia Lto	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	CES SECONINES MUSUALI CITY
																		-			1													-								1																									3											127770	-
Buy	Buy	Buy	Diag.	Blov	Buy	Buy	Buy	Виу	Buy	Birv	Bin	Buy	Bin	Buy	Biw	Buy	Ring	Dini	Call	Suy	YUB	Buy	Buy	Buy	Sell	Buy	Buy	Buy	Buy	Buy	Sell	Buy	Buy	Biv	Sall 9	Biry	Buy	Buy	Buy	Sell	Виу	Buy	Sell	Buy	Sell	Buy	Design	Soll	Виу	Buy	Buy	Buy	Buy	Buy	guy	Sell	Виу	Buy	Buy	Buy	Buy	Buv	Виу	Birv	Biv	Buy	Buy	Buy	Buy	Виу	Sell	Buy	Rha						
6,127	α./	2,/32	1,100	292 2	265	DPC 6	2.276	3.528	4 483	1 098	2020	1 200	2,060	T 781	308	13,494	122	34134	0,040	116	504	814	7.4	53	53	53	100	142	725	137	137	137	331	787	782	92	2/9	752	331	226	236	226	236	236	236	226	1/0,1	1,071	1,0/1	2,058	2,053	173	992	778	1,020	221	74	221	221	53	1.502	530	5	52	53 5	53	53	579	37	772	772	463	826	2,677	95	289	2,372	1,841	24
1,156	154	515	101	AGA	50	1 752	479	5915	270	207	202	262	1000	1050	£,5,70	2.546	0,430	1,07,061	(00,757)	22	95	155	14	10	(10)	10	19	27	138	26	(36)	26	63	1/19	(1.40)	100	ū	48	83	(43)	45	43	(45)	45	(45)	4	(407)	204	204	392	391	33	189	148	194	(42)	14	42	42	10	286	101	_ [10.	10	10	10	110	7	147	(147)	88	157	509	18	55	451	350	ĵς
Ordinary	Ordinary	Ordinary	Clullery	000	Ordinary	Ordina	Ordinary	Ordinary	Ordinary	Ordinary	Cidinaly	Ordinary	O di lai	Colonia	Ordinary	Ordinary	Ordinary	O O I I I	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary

ाठाठा	510		ol C	जट	16	वट	ıc	10	10	Io	10	ol c	जट	al c	i.c	17	নাব	n F	ਨਾ	<u>~</u> 1.	<u>حاد</u>	<u> </u>	512	गर	STA	SI &	য়ত	al c	T-	J.	SI o	-TF	- T	-T-	-T-	जार	NI -	, I z-			- -	a tra			1.	-		-=-		_,	_,																														
7-Apr-11	7-Apr-11	b-Apr-11	6-Apr-11	6-Apr-11	6-Apr-11	6-Apr-11	6-Apr-11	5-Apr-11	5-Apr-11	ı	5-Apr-11	Ĺ	i ADI-II	ı		ſ	N A 25 11	5-Apr-11	1	5-Apr-11	ı)5. Apr-11	7 NOT 11	S April	S ADV 11	S-ADI-I	ı		4-Apr-11	ı		A 200	Т		M-A27-11	A April 1	A-A-D-11	04-Apr-11	J4-Apr-1	J4-ADI-11	J4-Apr-11	J4-Apr-11	04-ADT-11	04-Apr-11	04-Apr-11	04-Apr-11	0 4- Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-Apr-11	04-401-11	04-Apr-11	04-10-11	01-Anr-11	01-Apr-11	31-Mar-11											
UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IUBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Obo Decurities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	Line Securities Australia Ltd	I IBS Securities Australia I td	LIBS Securities Australia I to	LIBS Securities Australia Ltd	I IRK Securities Australia I to	IIBS Securities Australia I to	I IBS Socialities Australia IIG	Obsider Mostralia Ltd	Obo pecurities Australia cio	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	Cos securities Australia Lio	Opp Securities Australia Ltd	LIBS Securities Australia Ltd	LIDS Securities Australia Ltd	Obsider Australia Ita	Coo securities Australia Ltd	Copy Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia and	UBS Securities Australia Ltd	LIRS Securities Australia I td	URS Securities Australia Ind	ILIBS Securities Australia Ltd	LIBS Securities Australia Ltd	I IBS Securities Australia I to	OBS Securities Australia Ltd	URS Securities Australia Ltd	URS Securities Australia I to	UBS Securities Australia I td	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securales Australia Ltd	OBS Securities Australia Ltd	Obs Securities Australia Ltd	TIDS Sportistics Australia 14d	UBS Securities Australia Ltd	LUBS Securities Australia Ltd	UBS Securities Australia Ltd
Buy Sell		BI		2	5	Ę.	Bi	Ş	Ş	5	Ş	2	2	2	5				0.5	200								5	S	5			1			-	2		5			5	2			2																																			
лу 21,979 лу 36,161 ell 62,944				ell 21	de la constante de la constant																	20/ 20/		ell 15,797			Buy 31,935		Self 10,042													298						Sell 1,809	sell 389		3.98.4 3.98.4	076		96	Sell 3.96			Sell (977							Sall 42					Sell 148	Sell 5,54		Sell . 18,794		Sell 1,364	Self Control of the C		16,100		Sell. 3,70	EVI IVUB
6,800 (11,852)		79	(4)	(4)	(14,574)			(1,062)							(34)	(32)	20,2			(52)	l			(2,45%)	(4,/61)				(1,877)				(CZC)			(540)				~1		(56)			=			(339)	(73)	(359)	746	(183)	(240	(18		7 (37)		7 (194)				/10/1	Ī			367			(83	9 (28	9 {1,041		(3,5		(256				3.424	77 (700	or or
Ordinary Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary						Ordinary	$^{+}$	+	\top	т	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary		Т	۲	Ī	T	+	+	T	1	+	т	т	1	П	т	т	т	✝	1	_	Ordinary	T	Ordinary	T	T	Т	T	T	Ť	T	T	Ť	1	1	7	3	Ordinary	Ť	_		-	-	Т	Т	Т	Т	Т														

08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-401-11	08-Apr-11	08-407-11	00 00	08 Apr-11	U8-Apr-11	08-Apr-11	O8-Apr-11	08-Apr-11	US-ADI-1	DO ADI-I	OG ADI-II	OB-ADI-1	00 10-	00 1	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08.40	OR-Apr-11	08-Apr-1 1	O8 Apr-11	OS ADT	08.00	000	08-00-11	08 40-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-01-11	08-Apr-11	08-Apr-11	08-Apr-11	08 40-	OR Apr	08-Apr-11	08-401-11	08-Apr-11	08-Apr-11	08 40	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	00 00	08-407-11	00 Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08 Apr 11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08 Apr 11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	07 Apr 11	07-Apr-11	
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBS Securities Australia III	LIRS Sequities Australia Ltd	Obs securities Australia Ltd	Coo Securities Australia Lid	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	ODS SECURIES AUSTRALIA LEG	DDC Securities Australia Ltd	IRS Securities Australia Ltd	IRA Securities Australia Ltd	LIBS Contribut Australia Ltd	LIBS Securities Australia Its	IBS Securities Australia Ltd	I III Coordinates Australia Ltd	IRS Securities Australia 11d	I IBC Contribut Australia III	I BC Constitutes Australia Ltd	LIBS Securities Australia Liu	I los seculties Australia Liu	I IBS Securities Australia Lid	I DC Contribut Australia Ltd	LIBS Securities Australia Ltd	UBS Sequifies Australia I to	UBS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	LIBS Securities Australia Ltd	LBS Securities Australia Ltd	LBS Securities Australia I to	UBS Securities Australia I fd	IBS Securities Australia Ltd	IBS Societies Australia Ed	JBS Securities Alietralia Itd	BS Securities Australia Ltd	I BS Securities Australia Ltd	I IBS Securities Australia 14d	UBS Securities Australia I to	LIBS Securities Australia Ind	UBS Securities Australia IId	UBS Securities Australia Ltd	LIBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia I td	UBS Securities Australia Hd	I IBS Securities Australia III	LIBS Contracts Alexandra Lin	IRS Securities Australia Hd	I BS Securities A retrain 14d	IRS Securities Australia Hd	LIBS Sequition Alletralia I th	UBS Securities Australia I td	UBS Securities Australia Ind	JRS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	!
																																													7.1								840	-	78.000	77.000					THE PARTY OF THE P																			;	O Table		
Sell 6,162		Bin A 201	4	Sell 56		Sell 474		Buy 4,283	Sell 21		Sell 5,708				Sell: 883		Sell 5,054			5,631			6,12			2,113		Sell 1,/24		Sell 394		0,131			Blav 6.10				Brivi A A E 1	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Buy 5,049				163 1,890	Sell Sell		2007		6,151		6060	Bray 6.1	5611 0,102				Sell 6 163		D				(A)		Call 504	Soli O,		Buv 6.139						5,				Buy 15,148		Sell 1	Sell	
2 (1,154)		(12/)						3 805	9 (41)		1,073)		8 (41)	1)2 (38)		1,096		960'1	(1,11/)	(3/)	1,154	1,154	(/56)	(398)		(324)	1,154	34 (74)					1 184		51 (1154)			/	1,13/	1,124	1	17 000	30 (355	112		(200	17		(1,305)		121	02 (1,154)		62 1 154		62 (1,19)				(1,118)							39			56 (1,082	97 (37		756 (1,082	92 (36	192 (36	5)	148 3,026		76 (33	21 (
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Г	Τ	T	T	1	T	1	†	t	Ť	Ordinary	t	Ť	Ordinary	Ordinary	Ť	Ť	╬	1	T	Ordinary	T	y Ordinary	T	T	Ordinary	7	1	Ť	(1) Ordinary	Γ	П	Ordinary	Τ	Т	Т	Ŧ	Τ	Т	Т	Т	Ť	7) Ordinary	٦	٦	5) Ordinary	5) Ordinary	┪	6 Ordinary		1	4) Ordinary	

11-Apr-11	11-Apr-11	11-Apr-11	11-Apr-11	11-Apr-11	11-Apr-11	11-Apr-11	11-Apr-11	11-Apr-11	08-Apr-11	08-Apr-11	US-Apr-11	US-ADI-II	00 401-1	08 A	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	US-Apr-11	U8-ADT-11	00-Api-	00 Apr 11	00-Apr-11	00.00	08-22-11	OG ADI-	OS-ADI-II	00 Apr 11	OG-ADI-11	OG-ADI-II	00 ADI-11	00 10-	OR ADT	08-Apr-11	08.00	OR-Apr-11	00 Apr 11	OG ACT 11	00 Apr 11	00-20-11	OG ADI-II	00-401-11	OG ADI-II	OB-ADI-II	08-Apr-11	US-Apr-11	08-Apr-11	U8-Apr-11	08-Apr-11	U8-Apr-11	08-Apr-11	08-Apr-11	08-Apr-11	US-Apr-11	08-Apr-11	U8-Apr-11	U8-Apr-II	U8-ADI-11	US-Apr-II	US-Apr-	OG-ADI-II	00-Apr-11	00 ADI-11	00 Apr 11	00 00	OR ANY 11	08-Apr-11	08-Apr-11														
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIDS SOCIETIES AUSUALIA LIO	I DC Constitute Allertain I to	BS Securities Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securios Australia Ltd	Obo Securities Australia Ltd	LIDS COCITION AUSTRALIA LIO	Opp Securities Adstralia Ltd	I IDC Contribut Australia I to	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	U65 Securities Australia Ltd	OBS Securities Australia Ltd	OBS Securities Australia Ltd	CDD DECUTIONS ADSIGNED LIQ	Tipo constition Australia 144	I IBS Securities Australia Ital	LIBS Sociation Australia Ltd	I IBC Securities Australia Ltd	I De son which Australia I tel	Con secondes Australia Ltd	CBC Securities Adstraila Ftd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	OBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UB5 Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	OBS Securities Australia Ltd	UBS Securities Australia Etg	UBS Securities Australia Ltd	Inc Contribute Australia Ltd	Coo Secol files Adopting Life	IBC Contraine Australia Hd	UBS Securities Australia Ltd	t IRS Securities Australia I td	LIBS Securities Australia Ltd
Se	50	5	Se.		Se	Bu	56	ия	Br	BL	Bu	BL			009	P	25	8		54	S	S	×	F	1 4		- B	. v.	Bi		9	S		·	SU							Bu			8		8	S	B	S	S	\$	5		8	S	В	В		S	3	S	В	3				3	2	В	8													
2,273			0.037							11,326			7,391		40,450		2005	38C	ell 2.71:	ell 3,450	elf 1,185	ell 1,180	lle 89:	Jy 6,16	ell: 369	ell 530	6,1/	ell 6,174	<u>1791.</u>	uy 3,25i	uy 3,25	6,162	ell 76		b,151		54	ell 62	5,52	Sell 304						8uy 4,16		ell 5,77	1,970	ell 36	6,139			Sell 292		ell 33	uy 3,25	Luy 2,88			Sell 128				Sell 2,43		Sell 32	3,022		Buy 97	tuy 5,185		Buy 5,18			Sell 246						Sell 6 162		
			T	T	1			2.177	i	2.113					203/		1		(805)					1,154			1,154	(1		609	2) 609		(144)				2 (54)	(118)	(1,036)		(61)	9 1,154	_	(1,154)	_) 	8 (1,086)			9 (1,154	7 (118		Ţ			611	9 543		0 (564	8 (24		_		0 (456)	'		2 (567								6 (46)						71 154		41 674
Ordinary	Т	Т	Τ	Τ	1	Т	Ordinary	Ordinary	1	7	``†		Г	t	T	Ť	Ť	7	Ť	Ť	Ordinary	~	_	Ordinary	Ť	Ī	T	Ť	П	Ordinary	Т		Г		Г	۲	Ť	Ordinary	۲	-	Ť	H	Г	Ť	П	Η.	Н		T	~	7	Ordinary	Ordinary	7	Ť	Ÿ	7	Н	Ť	-	Ordinary	Ĭ	╗	┪	7		Ť	Ĭ	Ť		Ordinary			П	Г	Г		Ordinary	Г		Т	Ordinary	Т	Total nance

l
Stock returned
Stock returned
Sell
Sell
Sell
Sell
Sell
Sell
Sell
Sell
Seil
Sell
Sell
Sell
Self
Sell
Se
Sell
Sell
Sell
Sell
Виу
Sell
Yell

PRIME BROKERAGE AGREEMENT

Details

Interpretation – definitions are at the end of the General terms

Parties	UBS and Customer		
UBS	Name	UBS AG, Australia Branch	
	ABN	47 088 129 613	
	AFSL	231 087	
	Address	Level 16 Chifley Tower, 2 Chifley Square, Sydney, NSW, 2000	
	Telephone	+61 2 9324 2000	
	Fax	+61 2 9324 2558	
	Attention	General Counsel	
Customer	Name		
	ABN		
•	AFSL		
	Address		
	Telephone		
	Fax		
	Attention -		
	Scheme		
	ARSN		
Recitals	A The Customer wishes to appoint UBS to provide a prime brokerage service to the Customer in its capacity as trustee and manager of the Scheme.		
	B UBS wishes to accept that appointment.		
Governing law	New South Wales		
Date of agreement	See Signing page		

10.2 Custodial Assets

UBS may request the Custodian at any time to pay or deliver to UBS any of the Custodial Assets, provided that UBS may only request the Custodian to deliver to UBS Custodial Assets which are ASX listed securities if such delivery would not cause UBS to have a 'relevant interest' for the purposes of the Corporations Act 2001 (C'th) of greater than 18% in any ASX listed entity. Subject to clauses 10.3, 10.4, 10.5 and 10.6 the Customer agrees that any Custodial Assets which UBS requests the Custodian to pay or deliver to UBS. or any Collateral, may be used by UBS for UBS's own account (including to borrow, lend, charge, re-hypothecate, dispose of or otherwise use for its own purposes) and in respect of UBS's obligations (or those of other customers of UBS) and, as a consequence, those Assets are not held by UBS for the Customer or the Custodian. UBS will have a contractual obligation to return equivalent Custodial Assets to the Custodian in accordance with clause 10.6. The Customer and the Custodian will in relation to the obligation to return equivalent Custodial Assets rank as one of UBS's general creditors in the event of UBS becoming Insolvent. Subject to clause 4.2 of the ASLA, UBS may retain all fees, profits and other benefits received in connection with such activities.

10.3 Express Authorisation for Collateral

· ;. .

Without limiting UBS's right to request transfer of any of the Custodial Assets under clause 10.2, the Customer expressly authorises UBS in its discretion to:

- (a) identify any Collateral as being held as margin or security against a particular obligation of the Customer under this agreement or against an UBS Transaction;
- (b) subject to clauses 10.2 and 10.5, transfer any Custodial Assets to UBS expressly as Collateral for any obligations of the Customer under this agreement or an UBS Transaction; and
- (c) transfer the proceeds of a cash advance made to the Customer to any Related Entity to satisfy any margin or security requirement of a Related Entity in relation to a Transaction (provided that the Customer and UBS have previously agreed in writing that the Transaction is a Transaction to which this clause 10.3 applies).

10.4 No Derogation from Liability to provide Collateral

The authorisation of UBS set out in clause 10.3 does not derogate from the Customer's obligation to meet a demand for Outstanding Margin Requirement under clause 5.3 or any margin or security requirement owed to a Related Entity. Unless UBS agrees otherwise in a particular case, UBS is only deemed to have agreed to transfer Custodial Assets to meet an Outstanding Margin Requirement or transferring the proceeds of a cash advance to the Customer to meet a margin or security requirement of a Related Entity upon it actually transferring those Custodial Assets or proceeds (as applicable) and is not liable for failure to do so.

10.5 Provision of Collateral

Subject to UBS's rights under clause 10.2, Collateral which is required by UBS pursuant to clause 5, if provided to the Custodian, will be held by the Custodian as bare trustee for the Customer subject to the Security. Any other Collateral provided to UBS in accordance with clause 5 will be provided to UBS in accordance with the terms and conditions of the ASLA and with the Rules. Securities delivered by the Custodian to UBS under clause 10.2 will be deemed to be provided by the Customer to UBS in accordance with the terms and conditions of the ASLA and with the Rules. UBS will become the legal and beneficial owner of those securities upon taking delivery of them from the Custodian.

10.6 Custodial Assets to be borrowed by UBS

Any Custodial Assets which UBS has the Custodian pay or deliver to it, will be borrowed by UBS from the Customer in accordance with the terms and conditions of the ASLA and with the Rules. If any of the terms of the ASLA are inconsistent with any of the terms of this agreement, this agreement prevails to the extent of the inconsistency.

10.7 Withholding Taxes on Income

If a law requires UBS to deduct an amount in respect of Taxes in relation to any income or other payments to the Customer under this agreement, the Customer authorises UBS to make such deductions without any further express instructions. UBS will pay to the Customer the amount of income or other payments net of Taxes.

11 Representations, Warranties and Acknowledgment

11.1 Customer's representations and warranties

The Customer represents and warrants to UBS that:

- (a) it has the power to enter into and perform its obligations under this agreement, and has duly executed this agreement so as to constitute valid and binding obligations of the Customer;
- it has duly executed this agreement in its capacity as trustee and manager of the Scheme and for the benefit of the beneficiaries of the Scheme;
- (c) it holds such licences and authorities as are necessary to lawfully perform its obligations under this agreement;
- (d) in giving any instructions under this agreement, the Customer will act as principal;
- (e) in giving any instructions under this agreement, the Customer will act in accordance with the provisions of its constitution, the constitution of the Scheme or other constituent documents, any applicable laws and regulations and comply with any investment restrictions in any prospectus, information memorandum, investment management

INTERNATIONAL SECURITIES LENDERS ASSOCIATION ISLA GLOBAL MASTER SECURITIES LENDING AGREEMENT

CLIFFORD CHANCE

another as provided for in this Agreement, the Party obtaining such title being obliged to redeliver Equivalent Securities or Equivalent Collateral as the case may be.

2.4 Currency conversions

For the purposes of determining any prices, sums or values (including Market Value, Required Collateral Value, Relevant Value, Bid Value and Offer Value for the purposes of paragraphs 5 and 10 of this Agreement) prices, sums or values stated in currencies other than the Base Currency shall be converted into the Base Currency at the latest available spot rate of exchange quoted by a bank selected by Lender (or if an Event of Default has occurred in relation to Lender, by Borrower) in the London interbank market for the purchase of the Base Currency with the currency concerned on the day on which the calculation is to be made or, if that day is not a Business Day the spot rate of exchange quoted at Close of Business on the immediately preceding Business Day.

2.5 The parties confirm that introduction of and/or substitution (in place of an existing currency) of a new currency as the lawful currency of a country shall not have the effect of altering, or discharging, or excusing performance under, any term of the Agreement or any Loan thereunder, nor give a party the right unilaterally to alter or terminate the Agreement or any Loan thereunder. Securities will for the purposes of this Agreement be regarded as equivalent to other securities notwithstanding that as a result of such introduction and/or substitution those securities have been redenominated into the new currency or the nominal value of the securities has changed in connection with such redenomination.

2.6 Modifications etc to legislation

Any reference in this Agreement to an act, regulation or other legislation shall include a reference to any statutory modification or re-enactment thereof for the time being in force.

3. LOANS OF SECURITIES

Lender will lend Securities to Borrower, and Borrower will borrow Securities from Lender in accordance with the terms and conditions of this Agreement. The terms of each Loan shall be agreed prior to the commencement of the relevant Loan either orally or in writing (including any agreed form of electronic communication) and confirmed in such form and on such basis as shall be agreed between the Parties. Any confirmation produced by a Party shall not supersede or prevail over the prior oral, written or electronic communication (as the case may be).

4. **DELIVERY**

4.1 Delivery of Securities on commencement of Loan

Lender shall procure the delivery of Securities to Borrower or deliver such Securities in accordance with this Agreement and the terms of the relevant Loan. Such Securities shall be deemed to have been delivered by Lender to Borrower on delivery to Borrower or as it shall direct of the relevant instruments of transfer, or in the case of Securities held by an agent or within a clearing or settlement system on the effective instructions to such

agent or the operator of such system which result in such Securities being held by the operator of the clearing system for the account of the Borrower or as it shall direct, or by such other means as may be agreed.

4.2 Requirements to effect delivery

The Parties shall execute and deliver all necessary documents and give all necessary instructions to procure that all right, title and interest in:

- (a) any Securities borrowed pursuant to paragraph 3;
- (b) any Equivalent Securities redelivered pursuant to paragraph 8;
- (c) any Collateral delivered pursuant to paragraph 5;
- (d) any Equivalent Collateral redelivered pursuant to paragraphs 5 or 8;

shall pass from one Party to the other subject to the terms and conditions set out in this Agreement, on delivery or redelivery of the same in accordance with this Agreement with full title guarantee, free from all liens, charges and encumbrances. In the case of Securities, Collateral, Equivalent Securities or Equivalent Collateral title to which is registered in a computer based system which provides for the recording and transfer of title to the same by way of book entries, delivery and transfer of title shall take place in accordance with the rules and procedures of such system as in force from time to time. The Party acquiring such right, title and interest shall have no obligation to return or redeliver any of the assets so acquired but, in so far as any Securities are borrowed or any Collateral is delivered to such Party, such Party shall be obliged, subject to the terms of this Agreement, to redeliver Equivalent Securities or Equivalent Collateral as appropriate.

4.3 Deliveries to be simultaneous unless otherwise agreed

Where under the terms of this Agreement a Party is not obliged to make a delivery unless simultaneously a delivery is made to it, subject to and without prejudice to its rights under paragraph 8.6 such Party may from time to time in accordance with market practice and in recognition of the practical difficulties in arranging simultaneous delivery of Securities, Collateral and cash transfers waive its right under this Agreement in respect of simultaneous delivery and/or payment provided that no such waiver (whether by course of conduct or otherwise) in respect of one transaction shall bind it in respect of any other transaction.

4.4 Deliveries of Income

In respect of Income being paid in relation to any Loaned Securities or Collateral, Borrower in the case of Income being paid in respect of Loaned Securities and Lender in the case of Income being paid in respect of Collateral shall provide to the other Party, as the case may be, any endorsements or assignments as shall be customary and appropriate to effect the delivery of money or property equivalent to the type and amount of such Income to Lender, irrespective of whether Borrower received the same in respect of any

Loaned Securities or to Borrower, irrespective of whether Lender received the same in respect of any Collateral.

5. COLLATERAL

5.1 Delivery of Collateral on commencement of Loan

Subject to the other provisions of this paragraph 5, Borrower undertakes to deliver to or deposit with Lender (or in accordance with Lender's instructions) Collateral simultaneously with delivery of the Securities to which the Loan relates and in any event no later than Close of Business on the Settlement Date. In respect of Collateral comprising securities, such Collateral shall be deemed to have been delivered by Borrower to Lender on delivery to Lender or as it shall direct of the relevant instruments of transfer, or in the case of such securities being held by an agent or within a clearing or settlement system, on the effective instructions to such agent or the operator of such system, which result in such securities being held by the operator of the clearing system for the account of the Lender or as it shall direct, or by such other means as may be agreed.

5.2 Deliveries through payment systems generating automatic payments

Unless otherwise agreed between the Parties, where any Securities, Equivalent Securities, Collateral or Equivalent Collateral (in the form of securities) are transferred through a book entry transfer or settlement system which automatically generates a payment or delivery, or obligation to pay or deliver, against the transfer of such securities, then:-

- (i) such automatically generated payment, delivery or obligation shall be treated as a payment or delivery by the transferee to the transferor, and except to the extent that it is applied to discharge an obligation of the transferee to effect payment or delivery, such payment or delivery, or obligation to pay or deliver, shall be deemed to be a transfer of Collateral or redelivery of Equivalent Collateral, as the case may be, made by the transferee until such time as the Collateral or Equivalent Collateral is substituted with other Collateral or Equivalent Collateral if an obligation to deliver other Collateral or redeliver Equivalent Collateral existed immediately prior to the transfer of Securities, Equivalent Securities, Collateral or Equivalent Collateral; and
- (ii) the party receiving such substituted Collateral or Equivalent Collateral, or if no obligation to deliver other Collateral or redeliver Equivalent Collateral existed immediately prior to the transfer of Securities, Equivalent Securities, Collateral or Equivalent Collateral, the party receiving the deemed transfer of Collateral or redelivery of Equivalent Collateral, as the case may be, shall cause to be made to the other party for value the same day either, where such transfer is a payment, an irrevocable payment in the amount of such transfer or, where such transfer is a delivery, an irrevocable delivery of securities (or other property, as the case may be) equivalent to such property.

5.3 Substitutions of Collateral

Borrower may from time to time call for the repayment of Cash Collateral or the redelivery of Collateral equivalent to any Collateral delivered to Lender prior to the date on which the same would otherwise have been repayable or redeliverable provided that at the time of such repayment or redelivery Borrower shall have delivered or delivers Alternative Collateral acceptable to Lender and Borrower is in compliance with paragraph 5.4 or paragraph 5.5, as applicable.

5.4 Marking to Market of Collateral during the currency of a Loan on aggregated basis

Unless paragraph 1.3 of the Schedule indicates that paragraph 5.5 shall apply in lieu of this paragraph 5.4, or unless otherwise agreed between the Parties:-

- (i) the aggregate Market Value of the Collateral delivered to or deposited with Lender (excluding any Equivalent Collateral repaid or redelivered under Paragraphs 5.4(ii) or 5.5(ii) (as the case may be) ("Posted Collateral") in respect of all Loans outstanding under this Agreement shall equal the aggregate of the Market Value of the Loaned Securities and the applicable Margin (the "Required Collateral Value") in respect of such Loans;
- (ii) if at any time on any Business Day the aggregate Market Value of the Posted Collateral in respect of all Loans outstanding under this Agreement exceeds the aggregate of the Required Collateral Values in respect of such Loans, Lender shall (on demand) repay and/or redeliver, as the case may be, to Borrower such Equivalent Collateral as will eliminate the excess;
- (iii) if at any time on any Business Day the aggregate Market Value of the Posted Collateral in respect of all Loans outstanding under this Agreement falls below the aggregate of Required Collateral Values in respect of all such Loans, Borrower shall (on demand) provide such further Collateral to Lender as will eliminate the deficiency.

5.5 Marking to Market of Collateral during the currency of a Loan on a Loan by Loan basis

If paragraph 1.3 of the Schedule indicates this paragraph 5.5 shall apply in lieu of paragraph 5.4, the Posted Collateral in respect of any Loan shall bear from day to day and at any time the same proportion to the Market Value of the Loaned Securities as the Posted Collateral bore at the commencement of such Loan. Accordingly:

- (i) the Market Value of the Posted Collateral to be delivered or deposited while the Loan continues shall be equal to the Required Collateral Value;
- (ii) if at any time on any Business Day the Market Value of the Posted Collateral in respect of any Loan exceeds the Required Collateral Value in respect of such Loan, Lender shall (on demand) repay and/or redeliver, as the case may be, to Borrower such Equivalent Collateral as will eliminate the excess; and
- (iii) if at any time on any Business Day the Market Value of the Posted Collateral falls below the Required Collateral Value, Borrower shall (on demand) provide such further Collateral to Lender as will eliminate the deficiency.

5.6 Requirements to redeliver excess Collateral

Where paragraph 5.4 applies, unless paragraph 1.4 of the Schedule indicates that this paragraph 5.6 does not apply, if a Party (the "first Party") would, but for this paragraph 5.6, be required under paragraph 5.4 to provide further Collateral or redeliver Equivalent Collateral in circumstances where the other Party (the "second Party") would, but for this paragraph 5.6, also be required to or provide Collateral or redeliver Equivalent Collateral under paragraph 5.4, then the Market Value of the Collateral or Equivalent Collateral deliverable by the first Party ("X") shall be set-off against the Market Value of the Collateral or Equivalent Collateral deliverable by the second Party ("Y") and the only obligation of the Parties under paragraph 5.4 shall be, where X exceeds Y, an obligation of the first Party, or where Y exceeds X, an obligation of the second Party to repay and/or (as the case may be) redeliver Equivalent Collateral or to deliver further Collateral having a Market Value equal to the difference between X and Y.

5.7 Where Equivalent Collateral is repaid or redelivered (as the case may be) or further Collateral is provided by a Party under paragraph 5.6, the Parties shall agree to which Loan or Loans such repayment, redelivery or further provision is to be attributed and failing agreement it shall be attributed, as determined by the Party making such repayment, redelivery or further provision to the earliest outstanding Loan and, in the case of a repayment or redelivery up to the point at which the Market Value of Collateral in respect of such Loan equals the Required Collateral Value in respect of such Loan, and then to the next earliest outstanding Loan up to the similar point and so on.

5.8 Timing of repayments of excess Collateral or deliveries of further Collateral

Where any Equivalent Collateral falls to be repaid or redelivered (as the case may be) or further Collateral is to be provided under this paragraph 5, unless otherwise agreed between the Parties, it shall be delivered on the same Business Day as the relevant demand. Equivalent Collateral comprising securities shall be deemed to have been delivered by Lender to Borrower on delivery to Borrower or as it shall direct of the relevant instruments of transfer, or in the case of such securities being held by an agent or within a clearing or settlement system on the effective instructions to such agent or the operator of such system which result in such securities being held by the operator of the clearing system for the account of the Borrower or as it shall direct or by such other means as may be agreed.

5.9 Substitutions and extensions of Letters of Credit

Where Collateral is a Letter of Credit, Lender may by notice to Borrower require that Borrower, on the Business Day following the date of delivery of such notice, substitute Collateral consisting of cash or other Collateral acceptable to Lender for the Letter of Credit. Prior to the expiration of any Letter of Credit supporting Borrower's obligations hereunder, Borrower shall, no later than 10.30a.m. UK time on the second Business Day prior to the date such Letter of Credit expires, obtain an extension of the expiration of such Letter of Credit or replace such Letter of Credit by providing Lender with a substitute Letter of Credit in an amount at least equal to the amount of the Letter of Credit for which it is substituted.

7. RATES APPLICABLE TO LOANED SECURITIES AND CASH COLLATERAL

7.1 Rates in respect of Loaned Securities

In respect of each Loan, Borrower shall pay to Lender, in the manner prescribed in subparagraph 7.3, sums calculated by applying such rate as shall be agreed between the Parties from time to time to the daily Market Value of the Loaned Securities.

7.2 Rates in respect of Cash Collateral

Where Cash Collateral is deposited with Lender in respect of any Loan, Lender shall pay to Borrower, in the manner prescribed in paragraph 7.3, sums calculated by applying such rates as shall be agreed between the Parties from time to time to the amount of such Cash Collateral. Any such payment due to Borrower may be set-off against any payment due to Lender pursuant to paragraph 7.1.

7.3 Payment of rates

In respect of each Loan, the payments referred to in paragraph 7.1 and 7.2 shall accrue daily in respect of the period commencing on and inclusive of the Settlement Date and terminating on and exclusive of the Business Day upon which Equivalent Securities are redelivered or Cash Collateral is repaid. Unless otherwise agreed, the sums so accruing in respect of each calendar month shall be paid in arrear by the relevant Party not later than the Business Day which is one week after the last Business Day of the calendar month to which such payments relate or such other date as the Parties shall from time to time agree.

8. REDELIVERY OF EQUIVALENT SECURITIES

8.1 Delivery of Equivalent Securities on termination of a Loan

Borrower shall procure the redelivery of Equivalent Securities to Lender or redeliver Equivalent Securities in accordance with this Agreement and the terms of the relevant Loan on termination of the Loan. Such Equivalent Securities shall be deemed to have been delivered by Borrower to Lender on delivery to Lender or as it shall direct of the relevant instruments of transfer, or in the case of Equivalent Securities held by an agent or within a clearing or settlement system on the effective instructions to such agent or the operator of such system which result in such Equivalent Securities being held by the operator of the clearing system for the account of the Lender or as it shall direct, or by such other means as may be agreed. For the avoidance of doubt any reference in this Agreement or in any other agreement or communication between the Parties (howsoever expressed) to an obligation to redeliver or account for or act in relation to Loaned Securities shall accordingly be construed as a reference to an obligation to redeliver or account for or act in relation to Equivalent Securities.

8.2 Lender's right to terminate a Loan

Subject to paragraph 10 and the terms of the relevant Loan, Lender shall be entitled to terminate a Loan and to call for the redelivery of all or any Equivalent Securities at any time by giving notice on any Business Day of not less than the standard settlement time for such Equivalent Securities on the exchange or in the clearing organisation through

which the Loaned Securities were originally delivered. Borrower shall redeliver such Equivalent Securities not later than the expiry of such notice in accordance with Lender's instructions.

8.3 Borrower's right to terminate a Loan

Subject to the terms of the relevant Loan, Borrower shall be entitled at any time to terminate a Loan and to redeliver all and any Equivalent Securities due and outstanding to Lender in accordance with Lender's instructions and Lender shall accept such redelivery.

8.4 Redelivery of Equivalent Collateral on termination of a Loan

On the date and time that Equivalent Securities are required to be redelivered by Borrower on the termination of a Loan, Lender shall simultaneously (subject to paragraph 5.4 if applicable) repay to Borrower any Cash Collateral or, as the case may be, redeliver Collateral equivalent to the Collateral provided by Borrower pursuant to paragraph 5 in respect of such Loan. For the avoidance of doubt any reference in this Agreement or in any other agreement or communication between the Parties (however expressed) to an obligation to redeliver or account for or act in relation to Collateral shall accordingly be construed as a reference to an obligation to redeliver or account for or act in relation to Equivalent Collateral.

8.5 Redelivery of Letters of Credit

Where a Letter of Credit is provided by way of Collateral, the obligation to redeliver Equivalent Collateral is satisfied by Lender redelivering for cancellation the Letter of Credit so provided, or where the Letter of Credit is provided in respect of more than one Loan, by Lender consenting to a reduction in the value of the Letter of Credit.

8.6 Redelivery obligations to be reciprocal

Neither Party shall be obliged to make delivery (or make a payment as the case may be) to the other unless it is satisfied that the other Party will make such delivery (or make an appropriate payment as the case may be) to it. If it is not so satisfied (whether because an Event of Default has occurred in respect of the other Party or otherwise) it shall notify the other party and unless that other Party has made arrangements which are sufficient to assure full delivery (or the appropriate payment as the case may be) to the notifying Party, the notifying Party shall (provided it is itself in a position, and willing, to perform its own obligations) be entitled to withhold delivery (or payment, as the case may be) to the other Party.

9. FAILURE TO REDELIVER

9.1 Borrower's failure to redeliver Equivalent Securities

(i) If Borrower does not redeliver Equivalent Securities in accordance with paragraph 8.1 or 8.2, Lender may elect to continue the Loan (which Loan, for the avoidance of doubt, shall continue to be taken into account for the purposes of paragraph 5.4 or 5.5 as applicable) provided that if Lender does not elect to continue the Loan, Lender may either by written notice to Borrower terminate

"Offer Price" in relation to Equivalent Securities or Equivalent Collateral means the best available offer price on the most appropriate market in a standard size;

"Offer Value" subject to paragraph 10.5 means:-

- (a) in relation to Collateral equivalent to Collateral in the form of a Letter of Credit zero and in relation to Cash Collateral the amount of the currency concerned; and
- (b) in relation to Equivalent Securities or Collateral equivalent to all other types of Collateral the amount it would cost to buy such Equivalent Securities or Equivalent Collateral at the Offer Price at Close of Business on the relevant Business Day together with all costs, fees and expenses that would be incurred in connection therewith, calculated on the assumption that the aggregate thereof is the least that could reasonably be expected to be paid in order to carry out the transaction and adding thereto the amount of any interest, dividends, distributions or other amounts, in the case of Equivalent Securities, paid to Borrower and in respect of which equivalent amounts have not been paid to Lender and in respect of which equivalent Collateral, paid to Lender and in respect of which equivalent amounts have not been paid to Borrower, in accordance with paragraph 6.1 prior to such time in respect of such Equivalent Securities, Equivalent Collateral or the original Securities or Collateral held, gross of all and any tax deducted or paid in respect thereof;

10.2 Termination of delivery obligations upon Event of Default

Subject to paragraph 9, if an Event of Default occurs in relation to either Party, the Parties' delivery and payment obligations (and any other obligations they have under this Agreement) shall be accelerated so as to require performance thereof at the time such Event of Default occurs (the date of which shall be the "Termination Date" for the purposes of this clause) so that performance of such delivery and payment obligations shall be effected only in accordance with the following provisions:

- (i) the Relevant Value of the securities which would have been required to be delivered but for such termination (or payment to be made, as the case may be) by each Party shall be established in accordance with paragraph 10.3; and
- (ii) on the basis of the Relevant Values so established, an account shall be taken (as at the Termination Date) of what is due from each Party to the other and (on the basis that each Party's claim against the other in respect of delivery of Equivalent Securities or Equivalent Collateral or any cash payment equals the Relevant Value thereof) the sums due from one Party shall be set-off against the sums due from the other and only the balance of the account shall be payable (by the Party having the claim valued at the lower amount pursuant to the foregoing) and such balance shall be payable on the Termination Date.

If the Bid Value is greater than the Offer Value, and the Non-Defaulting Party had delivered to the Defaulting Party a Letter of Credit, the Defaulting Party shall draw on

the Letter of Credit to the extent of the balance due and shall subsequently redeliver for cancellation the Letter of Credit so provided.

If the Offer Value is greater than the Bid Value, and the Defaulting Party had delivered to the Non-Defaulting Party a Letter of Credit, the Non-Defaulting Party shall draw on the Letter of Credit to the extent of the balance due and shall subsequently redeliver for cancellation the Letter of Credit so provided.

In all other circumstances, where a Letter of Credit has been provided to a Party, such Party shall redeliver for cancellation the Letter of Credit so provided.

10.3 Determination of delivery values upon Event of Default

For the purposes of paragraph 10.2 the "Relevant Value":-

- (i) of any securities to be delivered by the Defaulting Party shall, subject to paragraph 10.5 below, equal the Offer Value of such securities; and
- (ii) of any securities to be delivered to the Defaulting Party shall, subject to paragraph 10.5 below, equal the Bid Value of such securities.
- 10.4 For the purposes of paragraph 10.3, but subject to paragraph 10.5, the Bid Value and Offer Value of any securities shall be calculated for securities of the relevant description (as determined by the Non-Defaulting Party) as of the first Business Day following the Termination Date, or if the relevant Event of Default occurs outside the normal business hours of such market, on the second Business Day following the Termination Date (the "Default Valuation Time");
- 10.5 Where the Non-Defaulting Party has following the occurrence of an Event of Default but prior to the close of business on the fifth Business Day following the Termination Date purchased securities forming part of the same issue and being of an identical type and description to those to be delivered by the Defaulting Party or sold securities forming part of the same issue and being of an identical type and description to those to be delivered by him to the Defaulting Party, the cost of such purchase or the proceeds of such sale, as the case may be, (taking into account all reasonable costs, fees and expenses that would be incurred in connection therewith) shall (together with any amounts owing pursuant to paragraph 6.1) be treated as the Offer Value or Bid Value, as the case may be, of the amount of securities to be delivered which is equivalent to the amount of the securities so bought or sold, as the case may be, for the purposes of this paragraph 10, so that where the amount of securities to be delivered is more than the amount so bought or sold as the case may be, the Offer Value or Bid Value as the case may be, of the balance shall be valued in accordance with paragraph 10.4.
- 10.6 Any reference in this paragraph 10 to securities shall include any asset other than cash provided by way of Collateral.

10.7 Other costs, expenses and interest payable in consequence of an Event of Default

The Defaulting Party shall be liable to the Non-Defaulting Party for the amount of all reasonable legal and other professional expenses incurred by the Non-Defaulting Party in

connection with or as a consequence of an Event of Default, together with interest thereon at the one-month London Inter Bank Offered Rate as quoted on a reputable financial information service ("LIBOR") as of 11.00 am, London Time, on the date on which it is to be determined or, in the case of an expense attributable to a particular transaction and where the parties have previously agreed a rate of interest for the transaction, that rate of interest if it is greater than LIBOR. The rate of LIBOR applicable to each month or part thereof that any sum payable pursuant to this paragraph 10.7 remains outstanding is the rate of LIBOR determined on the first Business Day of any such period of one month or any part thereof. Interest will accrue daily on a compound basis and will be calculated according to the actual number of days elapsed.

11. TRANSFER TAXES

Borrower hereby undertakes promptly to pay and account for any transfer or similar duties or taxes chargeable in connection with any transaction effected pursuant to or contemplated by this Agreement, and shall indemnify and keep indemnified Lender against any liability arising as a result of Borrower's failure to do so.

12. LENDER'S WARRANTIES

Each Party hereby warrants and undertakes to the other on a continuing basis to the intent that such warranties shall survive the completion of any transaction contemplated herein that, where acting as a Lender:

- (a) it is duly authorised and empowered to perform its duties and obligations under this Agreement;
- (b) it is not restricted under the terms of its constitution or in any other manner from lending Securities in accordance with this Agreement or from otherwise performing its obligations hereunder;
- (c) it is absolutely entitled to pass full legal and beneficial ownership of all Securities provided by it hereunder to Borrower free from all liens, charges and encumbrances; and
- (d) it is acting as principal in respect of this Agreement or, subject to paragraph 16, as agent and the conditions referred to in paragraph 16.2 will be fulfilled in respect of any Loan which it makes as agent.

13. BORROWER'S WARRANTIES

Each Party hereby warrants and undertakes to the other on a continuing basis to the intent that such warranties shall survive the completion of any transaction contemplated herein that, where acting as a Borrower:

(a) it has all necessary licenses and approvals, and is duly authorised and empowered, to perform its duties and obligations under this Agreement and will do nothing prejudicial to the continuation of such authorisation, licences or approvals;

- (b) it is not restricted under the terms of its constitution or in any other manner from borrowing Securities in accordance with this Agreement or from otherwise performing its obligations hereunder;
- (c) it is absolutely entitled to pass full legal and beneficial ownership of all Collateral provided by it hereunder to Lender free from all liens, charges and encumbrances; and
- (d) it is acting as principal in respect of this Agreement.

14. EVENTS OF DEFAULT

- 14.1 Each of the following events occurring in relation to either Party (the "**Defaulting Party**", the other Party being the "**Non-Defaulting Party**") shall be an Event of Default for the purpose of paragraph 10 but only (subject to sub-paragraph (v) below) where the Non-Defaulting Party serves written notice on the Defaulting Party:-
 - Borrower or Lender failing to pay or repay Cash Collateral or deliver Collateral or redeliver Equivalent Collateral or Lender failing to deliver Securities upon the due date;
 - (ii) Lender or Borrower failing to comply with its obligations under paragraph 5;
 - (iii) Lender or Borrower failing to comply with its obligations under paragraph 6.1;
 - (iv) Borrower failing to comply with its obligations to deliver Equivalent Securities in accordance with paragraph 8;
 - (v) an Act of Insolvency occurring with respect to Lender or Borrower, an Act of Insolvency which is the presentation of a petition for winding up or any analogous proceeding or the appointment of a liquidator or analogous officer of the Defaulting Party not requiring the Non-Defaulting Party to serve written notice on the Defaulting Party;
 - (vi) any representation or warranty made by Lender or Borrower being incorrect or untrue in any material respect when made or repeated or deemed to have been made or repeated;
 - (vii) Lender or Borrower admitting to the other that it is unable to, or it intends not to, perform any of its obligations under this Agreement and/or in respect of any Loan;
 - (viii) Lender (if applicable) or Borrower being declared in default or being suspended or expelled from membership of or participation in, any securities exchange or association or suspended or prohibited from dealing in securities by any regulatory authority;
 - (ix) any of the assets of Lender or Borrower or the assets of investors held by or to the order of Lender or Borrower being transferred or ordered to be transferred to a trustee (or a person exercising similar functions) by a regulatory authority pursuant to any securities regulating legislation, or

- (x) Lender or Borrower failing to perform any other of its obligations under this Agreement and not remedying such failure within 30 days after the Non-Defaulting Party serves written notice requiring it to remedy such failure.
- 14.2 Each Party shall notify the other (in writing) if an Event of Default or an event which, with the passage of time and/or upon the serving of a written notice as referred to above, would be an Event of Default, occurs in relation to it.
- 14.3 The provisions of this Agreement constitute a complete statement of the remedies available to each Party in respect of any Event of Default.
- 14.4 Subject to paragraph 9.3 and 10.7, neither Party may claim any sum by way of consequential loss or damage in the event of failure by the other party to perform any of its obligations under this Agreement.

15. INTEREST ON OUTSTANDING PAYMENTS

In the event of either Party failing to remit sums in accordance with this Agreement such Party hereby undertakes to pay to the other Party upon demand interest (before as well as after judgment) on the net balance due and outstanding, for the period commencing on and inclusive of the original due date for payment to (but excluding) the date of actual payment, in the same currency as the principal sum and at the rate referred to in paragraph 10.7. Interest will accrue daily on a compound basis and will be calculated according to the actual number of days elapsed.

16. TRANSACTIONS ENTERED INTO AS AGENT

16.1 Power for Lender to enter into Loans as agent

Subject to the following provisions of this paragraph, Lender may (if so indicated in paragraph 6 of the Schedule) enter into Loans as agent (in such capacity, the "Agent") for a third person (a "Principal"), whether as custodian or investment manager or otherwise (a Loan so entered into being referred to in this paragraph as an "Agency Transaction").

16.2 Conditions for agency loan

A Lender may enter into an Agency Transaction if, but only if:-

- (i) it specifies that Loan as an Agency Transaction at the time when it enters into it;
- (ii) it enters into that Loan on behalf of a single Principal whose identity is disclosed to Borrower (whether by name or by reference to a code or identifier which the Parties have agreed will be used to refer to a specified Principal) at the time when it enters into the Loan or as otherwise agreed between the Parties; and
- (iii) it has at the time when the Loan is entered into actual authority to enter into the Loan and to perform on behalf of that Principal all of that Principal's obligations under the agreement referred to in paragraph 16.4(ii).



Australian Securities Lending Association Limited

(ACN 054 944 482)

www.asla.com.au

Registered Office
Level 50, MLC Centre
19-29 Martin Place
Sydney NSW 2000

(Version: December 2002)

AUSTRALIAN MASTER SECURITIES LENDING AGREEMENT *

Dated as of:			
Between:	(1)	(Name of Company) UBS Securities Australia Ltd	
		(ACN or ARBN (as applicable)) 62 008 586 481	
		a company incorporated under the laws of the ACT, Australia	
		of (Business address) Level 16 Chifley Tower, 2 Chifley Square, Sydney, NSW, 2000	
And:	(2)	(Name of Company)	
		(ACN or ARBN (as applicable))	
		a company incorporated under the laws of	
		of (Business address)	

- * The original (Version: 4 April 1997) version of this agreement was adapted from the ISLA Overseas Securities Lender's Agreement (Version: December 1995, as amended by 1996 UK Tax Addendum), prepared by Clifford Chance, London, England for use by parties required to meet UK Inland Revenue tax requirements. The 4 April 1997 version has been updated in December 2002 to take account of, among other things, intervening Australian tax, stamp duty and regulatory changes, and also to better reflect Australian market practice.
- * The original and updated versions of this agreement are both also subject to the "Warning and Disclaimer" on the coversheet to the original (Version: 4 April 1997) and updated (Version: December 2002) "User's Guide" relating to this agreement.

©

m

Level 16 Chifley Tower, 2 Chifley Square, Sydney, NSW 2000 Telephone (02) 9296 2000 Fax (02) 9296 3999 DX 113 Sydney Ref: JCK

9-1-2

("title") shall pass from one Party to the other free and clear of any liens, claims, charges or encumbrances or any other interest of the Transferring Party or of any third party (other than a lien routinely imposed on all securities in a relevant clearance system), the Party obtaining such title being obliged to redeliver Equivalent Securities or Equivalent Collateral, as the case may be. Each Transfer under this Agreement will be made so as to constitute or result in a valid and legally effective transfer of the Transferring Party's legal and beneficial title to the recipient.

- (c) Where, in respect of any transaction, any distribution is made, or Income or fee is paid, other than in cash, the provisions of this agreement (other than clause 4.2(b)) shall apply, with necessary modifications, to the same extent as if the distribution, Income or fee had been made or paid in cash, and terms such as "pay" and "amount" shall be construed accordingly.
- 1.5 [Headings] All headings appear for convenience only and shall not affect the interpretation of this Agreement.
- 1.6 [Currency conversion] For the purposes of clauses 6, 8.3 and 8.4, when a conversion into the Base Currency is required, all prices, sums or values (including any Value, Offer Value and Bid Value) of Securities, Equivalent Securities, Collateral or Equivalent Collateral (including Cash Collateral) stated in currencies other than the Base Currency shall be converted into the Base Currency at the rate quoted by an Australian bank selected by the Lender (or, if an Event of Default has occurred in relation to the Lender, by the Borrower) at or about 11.00am (Sydney time) on the day of conversion as its spot rate for the sale by the bank of the Base Currency in exchange for the relevant other currency.
- 1.7 [Other agreements] Where at any time there is in existence any other agreement between the Parties the terms of which make provision for the lending of Securities (as defined in this Agreement) as well as other securities, the terms of this Agreement shall apply to the lending of such Securities to the exclusion of any other such agreement.
- 1.8 [Nominees] If payment is made or Securities, Equivalent Securities, Collateral or Equivalent Collateral is Transferred to a Party's nominee or otherwise in accordance with the directions of a Party (whether by the other Party or by a third party), it shall be deemed, for the purposes of this agreement, to have been paid or made or Transferred to the first mentioned Party.

2 Loans of Securities

- 2.1 [Borrowing Request and acceptance thereof] The Lender will lend Securities to the Borrower, and the Borrower will borrow Securities from the Lender, in accordance with the terms and conditions of this Agreement and with the Rules. The terms of each Loan should be agreed prior to the commencement of the relevant Loan, either orally or in writing (including any agreed form of electronic communication) and confirmed in such form and on such basis as is agreed between the Parties. Any confirmation produced by a Party shall not supersede or prevail over the prior oral, written or electronic communication (as the case may be).
- 2.2 [Changes to a Borrowing Request] The Borrower has the right to reduce the amount of Securities referred to in, or otherwise vary, a Borrowing Request provided that:
 - (a) the Borrower has notified the Lender of such reduction or variation no later than midday Australian Eastern standard or summer (as appropriate) time on

- the day which is two Business Days prior to the Settlement Date, unless otherwise agreed between the Parties, and
- (b) the Lender shall have accepted such reduction or variation (by whatever means).

3 Delivery of Securities

[Delivery of Securities] The Lender shall procure the delivery of Securities to the Borrower or deliver such Securities in accordance with the relevant agreement together with appropriate instruments of transfer (where necessary) duly stamped (where necessary) and such other instruments (if any) as may be requisite to vest title thereto in the Borrower. Such Securities shall be deemed to have been delivered by the Lender to the Borrower on delivery to the Borrower or as it shall direct of the relevant instruments of transfer and certificates or other documents of title (if any), or in the case of Securities title to which is registered in a computer based system which provides for the recording and transfer of title to the same by way of electronic entries (such as CHESS), on the transfer of title in accordance with the rules and procedures of such system as in force from time to time, or by such other means as may be agreed.

4 Title, Distributions and Voting

- 4.1 [Passing of title] The Parties shall execute and deliver all necessary documents and give all necessary instructions to procure that all right, title and interest in:
 - (a) any Securities borrowed pursuant to clause 2;
 - (b) any Equivalent Securities redelivered pursuant to clause 7;
 - (c) any Collateral delivered pursuant to clause 6;
 - (d) any Equivalent Collateral redelivered pursuant to clauses 6 or 7,

shall pass from one Party to the other, free from all liens, charges, equities and encumbrances, on delivery or redelivery of the same in accordance with this Agreement. In the case of Securities, Collateral, Equivalent Securities or Equivalent Collateral title to which is registered in a computer based system which provides for the recording and transfer of title to the same by way of electronic entries, delivery and transfer of title shall take place in accordance with the rules and procedures of such system as in force from time to time.

4.2 [Distributions]

(a) [Cash distributions] Unless otherwise agreed, where Income is paid by the issuer in relation to any Securities on or by reference to an Income Payment Date on which such Securities are the subject of a loan under this Agreement, the Borrower shall, on the date of the payment of such Income, or on such other date as the Parties may from time to time agree, (the "Relevant Payment Date") pay to the Lender a sum of money (a "Substitute payment") equivalent to the amount that the Lender would have been entitled to receive (after any deduction, withholding or payment for or on account of any tax made by the relevant issuer (or on its behalf) in respect of such Income) had such Securities not been loaned to the Borrower and been held by the Lender on the Income Payment Date, irrespective of whether the Borrower received the same.

- (b) [Corporate actions] Subject to paragraph (c) (unless otherwise agreed), where, in respect of any borrowed Securities or any Collateral, any rights relating to conversion, sub-division, consolidation, pre-emption, rights arising under a takeover offer or other rights, including those requiring election by the holder for the time being of such Securities or Collateral, become exercisable prior to the redelivery of Equivalent Securities or Equivalent Collateral, then the Lender or Borrower, as the case may be, may, within a reasonable time before the latest time for the exercise of the right or option, give written notice to the other Party that, on redelivery of Equivalent Securities or Equivalent Collateral, as the case may be, it wishes to receive Equivalent Securities or Equivalent Collateral in such form as will arise if the right is exercised or, in the case of a right which may be exercised in more than one manner, is exercised as is specified in such written notice.
- (c) [1936 Tax Act ss 26BC(3)(c)(ii) and (v) requirements] Notwithstanding paragraph (b), where, in respect of any borrowed Securities or any Collateral, the relevant issuer company, trustee, government or government authority issues any right or option in respect of the borrowed Securities or Collateral, as the case may be, the Borrower or the Lender, respectively, must deliver or make available, as the case may be, to the other Party on the date of such issue or on such other date as the Parties may from time to time agree:
 - (i) the right, or option; or
 - (ii) an identical right or option; or
 - (iii) a payment equal to the value to the Lender or the Borrower, respectively, of the right or option;

together with any such endorsements or assignments as shall be customary and appropriate.

- (d) [Manner of payment] Any payment to be made by the Borrower under this clause shall be made in a manner to be agreed between the Parties.
- 4.3 [Voting] Unless paragraph 4 in Schedule 1 specifies that this clause 4.3 does not apply, each Party undertakes that, where it holds Securities of the same description as any Securities borrowed by it or transferred to it by way of Collateral at a time when a right to vote arises in respect of such Securities, it will use its best endeavours to arrange for the voting rights attached to such Securities to be exercised in accordance with the instructions of the other Party provided always that each Party shall use its best endeavours to notify the other of its instructions in writing no later than seven Business Days prior to the date upon which such votes are exercisable, or as otherwise agreed between the Parties, and that the Party concerned shall not be obliged so to exercise the votes in respect of the number of Securities greater than the number so lent or transferred to it. For the avoidance of doubt, the Parties agree that, subject as hereinbefore provided, any voting rights attaching to the relevant Securities, Equivalent Securities, Collateral and/or Equivalent Collateral shall be exercisable by the persons in whose name they are registered, or in the case of Securities, Equivalent Securities, Collateral and/or Equivalent Collateral in bearer form by the persons by or on behalf of whom they are held, and not necessarily by the Borrower or the Lender (as the case may be).

5 Fees

- (a) for which the Collateral is cash:
 - (i) the Lender must pay a fee to the Borrower in respect of the amount of that Collateral, calculated at the rate agreed between them; and
 - (ii) unless the Parties otherwise agree, the Borrower is not obliged to pay a fee to the Lender;
- (b) for which there is no Cash Collateral, the Borrower must pay a fee to the Lender, calculated at the rate agreed between them.
- 5.2 [Where there are different types of Collateral] Where the Collateral comprises only partly cash, clause 5.1 is to be construed as if there were separate loans of Securities, one secured solely by Cash Collateral and the other secured solely by non-cash Collateral.
- [Calculation of fees] In respect of each loan of Securities, the payments referred to in clause 5.1 of this clause shall accrue daily in respect of the period commencing on and inclusive of the Settlement Day and terminating on and exclusive of the Business Day upon which Equivalent Securities are redelivered or Cash Collateral is repaid. Unless otherwise agreed, the sums so accruing in respect of each calendar month shall be paid in arrears by the Borrower to the Lender or to the Borrower by the Lender (as the case may be) not later than the Business Day which is one week after the last Business Day of the calendar month to which such payment relates or such other date as the Parties from time to time agree. Any payment made pursuant to clause 5.1 shall be in Australian currency, unless otherwise agreed, and shall be paid in such manner and at such place as shall be agreed between the Parties.

6 Collateral

[Borrower's obligation to provide Collateral] Unless otherwise agreed, subject to the other provisions of this clause 6, the Borrower undertakes to deliver to or deposit with the Lender (or in accordance with the Lender's instructions) Collateral of the kind specified in the relevant Borrowing Request or as otherwise agreed between the Parties (together with appropriate instruments of transfer duly stamped (where necessary) and such other instruments as may be requisite to vest title thereto in the Lender) simultaneously with delivery of the Borrowed Securities by the Lender.

6.2 [Global margining]

- (a) [Adjustments to Collateral] Unless otherwise agreed between the Parties, subject to paragraph (b), clause 6.4 and paragraph 1.5 in Schedule 1:
 - (i) The aggregate Value of the Collateral delivered to or deposited with the Lender or its nominated bank or depositary (excluding any Collateral repaid or redelivered under paragraph (ii) below (as the case may be)) in respect of all loans of Securities outstanding under this Agreement ("Posted Collateral") shall from day to day and at any time be at least the aggregate of the Required Collateral Values in respect of such loans.
 - (ii) If at any time the aggregate Value of the Posted Collateral in respect of all loans of Securities outstanding under this Agreement exceeds the aggregate of the Required Collateral Values in respect of such loans, the Lender shall (on demand) repay such Cash Collateral and/or redeliver to the Borrower such Equivalent Collateral as will eliminate the excess.

- (iii) If at any time the aggregate Value of the Posted Collateral in respect of all loans of Securities outstanding under this Agreement falls below the aggregate of Required Collateral Values in respect of all such loans, the Borrower shall (on demand) provide such further Collateral to the Lender as will eliminate the deficiency.
- [Netting of Collateral obligations where a Party is both Lender and (b) Borrowerl Unless otherwise agreed between the Parties, subject to clause 6.4 and paragraph 1.5 in Schedule 1, where paragraph (a) applies, if a Party (the "first Party") would, but for this paragraph, be required under paragraph (a) to repay Cash Collateral, redeliver Equivalent Collateral or provide further Collateral in circumstances where the other Party (the "second Party") would, but for this paragraph, also be required to repay Cash Collateral, redeliver Equivalent Collateral or provide further Collateral under paragraph (a), then the Value of the Cash Collateral, Equivalent Collateral or further Collateral deliverable by the first Party ("X") shall be set-off against the Value of the Cash Collateral, Equivalent Collateral or further Collateral deliverable by the second Party ("Y") and the only obligation of the Parties under paragraph (a) shall be, where X exceeds Y, an obligation of the first Party, or where Y exceed X, an obligation of the second Party, (on demand) to repay Cash Collateral, redeliver Equivalent Collateral or deliver further Collateral having a Value equal to the difference between X and Y.
- [Required Collateral Value] For the purposes of clause 6.2(a), the Value of the Posted Collateral to be delivered or deposited in respect of any loan of Securities, while the loan of Securities continues, shall be equal to the aggregate of the Value of the borrowed Securities and the Margin applicable thereto (the "Required Collateral Value").
- 6.4 [Time for payment/repayment of Collateral] Except as provided in clause 6.1 or clause 6.6 or as otherwise agreed, where any Cash Collateral is to be repaid, Equivalent Collateral is to be redelivered or further Collateral is to be provided under this clause 6, it shall be paid or delivered as stated in paragraph 1.4 in Schedule 1.
- 6.5 [Substitution of Alternative Collateral] The Borrower may from time to time call for the repayment of Cash Collateral or the redelivery of Equivalent Collateral prior to the date on which the same would otherwise have been repayable or redeliverable, provided that, at the time of such repayment or redelivery, the Borrower shall have delivered or delivers Alternative Collateral acceptable to the Lender.
- 6.6 [Return of Collateral/Equivalent Collateral on redelivery of Equivalent Securities]
 - (a) Cash Collateral shall be repaid and Equivalent Collateral shall be redelivered at the same time as Equivalent Securities in respect of the Securities borrowed are redelivered.
 - (b) Where Collateral is provided through a book entry transfer system (such as Austraclear or RITS), the obligation of the Lender shall be to redeliver Equivalent Collateral through such book entry transfer system in accordance with this Agreement. If the loan of Securities in respect of which Collateral was provided has not been discharged when the Equivalent Collateral is redelivered, any payment obligation generated within the book entry transfer system on such redelivery shall, until the loan of Securities is discharged or further Collateral is provided, be deemed to constitute an obligation to pay Cash Collateral.

- 6.7 [Receipt by Lender of Income on Collateral] Where Collateral (other than Cash Collateral) is delivered in respect of which any Income may become payable and an Income Payment Date in respect of that Collateral occurs prior to the redelivery of Equivalent Collateral, then, unless such Income is paid directly to the Borrower, the Lender shall, on the date on which such Income is paid or on such other date as the Parties may from time to time agree, pay to the Borrower a sum of money (a "Substitute payment") equivalent to the amount of such Income that (after any deduction, withholding or payment for or on account of any tax made by the relevant issuer (or on its behalf) in respect of such Income) the Lender either actually received, or would have been entitled to receive had such Collateral been held by the Lender on the Income Payment Date, irrespective of whether the Lender received the same. If the Lender is required by law, as modified by the practice of any relevant taxing authority, to make any deduction or withholding from any Substitute payment to be made under the preceding sentence, then the Lender must:
 - (a) promptly pay to the relevant taxing authority the full amount of the deduction or withholding; and
 - (b) forward to the Borrower on request a copy of any official receipt or other evidence showing that the full amount of any such deduction or withholding has been paid over to the relevant taxing authority.
- 6.8 [Borrower's rights re Collateral are not assignable] The Borrower may not assign, transfer or otherwise dispose of, or mortgage, charge or otherwise encumber, or otherwise deal with its rights in respect of any Collateral without the prior written consent of the Lender.
- 6.9 [Lender may set off obligation to repay or return Equivalent Collateral] If the Borrower fails to comply with its obligation to redeliver Equivalent Securities, the obligation of the Lender in respect of any Collateral may be the subject of a set-off in accordance with clause 8.
- 6.10 [Collateral provided to Lender's Nominee] Without limiting clause 1.8, where Collateral is provided to the Lender's nominee, any obligation under this Agreement to repay or redeliver or otherwise account for Equivalent Collateral shall be an obligation of the Lender, notwithstanding that any such repayment or redelivery may be effected in any particular case by the nominee.
- 6.11 [Letters of Credit] If the Collateral in respect of one or more loans of Securities is or includes a letter of credit, the Lender may only draw down under that letter of credit when an Event of Default occurs in relation to the Borrower and, upon the Lender drawing down, whether or not permitted under this clause 6.11, the Collateral (or that part of it represented by the letter of credit) becomes cash Collateral.
- 6.12 [Non-Cash Collateral] If the Collateral in respect of one or more loans of Securities is or includes other Securities and either the Borrower is a taxpayer to whom the Tax Act applies in respect of the disposal of those other Securities or in any other case the Parties so agree:
 - (a) The Parties acknowledge that the provision of those other Securities is by way of a loan of Securities under this Agreement, to which section 26BC(3)(a) of the 1936 Tax Act may apply (subject to the re-acquisition time being less than 12 months after the original disposal time).
 - (b) For the purposes of section 26BC(3)(d) of the 1936 Tax Act, the notifiable consideration in respect of the provision of those Securities by way of loan is specified as follows:

- (i) There is no fee.
- (ii) There is no adjustment for variations in the market value of the Collateral or Equivalent Collateral.
- (iii) There is other consideration: see the obligations of the recipient of the Collateral under clauses 4.2(b), 4.2(c), 4.3 and 6.7.
- (c) For the avoidance of doubt, this clause 6.12 is directed solely at clarifying either or both of the following issues: that the provision of the other Securities as Collateral is eligible for the application of first section 26BC and secondly, where applicable, the successor to sections 160AQUA and 160AQUD of the 1936 Tax Act. Accordingly, clauses 2, 4.2(a), 5, 6.1 to 6.11, 7, 8, 9.1, 9.2 (unless otherwise agreed), 9.4 and 12 do not apply to any loan of Securities under paragraph (a). Instead, those Securities are simply to be regarded as Collateral for the purposes of those clauses.

7 Redelivery of Equivalent Securities

- 7.1 [Borrower's obligation to redeliver Equivalent Securities] The Borrower undertakes to redeliver Equivalent Securities in accordance with this Agreement and the terms of the relevant Borrowing Request.
- 7.2 [Lender may call for redelivery of Equivalent Securities] Subject to clause 8 and the terms of the relevant Borrowing Request, the Lender may call for the redelivery of all or any Equivalent Securities at any time by giving notice on any Business Day of not less than the Standard Settlement Time for such Equivalent Securities or the equivalent time on the exchange or in the clearing organisation through which the relevant borrowed Securities were originally delivered. The Borrower shall as hereinafter provided redeliver such Equivalent Securities not later than the expiry of such notice in accordance with the Lender's instructions.
- 7.3 [Lender may terminate loan if Borrower defaults] If the Borrower does not redeliver Equivalent Securities in accordance with such call, the Lender may elect to continue the loan of Securities; provided that, if the Lender does not elect to continue the loan, the Lender may by written notice to the Borrower elect to terminate the relevant loan. Upon the expiry of such notice the provisions of clauses 8.2 to 8.5 shall apply as if upon the expiry of such notice an Event of Default had occurred in relation to the Borrower (who shall thus be the Defaulting Party for the purposes of this Agreement) and as if the relevant loan were the only loan outstanding.
- 7.4 [Consequence of exercise of "buy-in" against Lender, as a result of Borrower default] In the event that, as a result of the failure of the Borrower to redeliver Equivalent Securities to the Lender in accordance with this Agreement, a "buy-in" is exercised against the Lender, then, provided that reasonable notice has been given to the Borrower of the likelihood of such a "buy-in", the Borrower shall account to the Lender for the total costs and expenses reasonably incurred by the Lender as a result of such "buy-in".
- 7.5 [Right of Borrower to terminate loan early] Subject to the terms of the relevant Borrowing Request, the Borrower shall be entitled at any time to terminate a particular loan of Securities and to redeliver all and any Equivalent Securities due and outstanding to the Lender in accordance with the Lender's instructions.





Master Securities Loan Agreement

2000 Version		
Dated as of:		
Between:		
and		

1. Applicability.

From time to time the parties hereto may enter into transactions in which one party ("Lender") will lend to the other party ("Borrower") certain Securities (as defined herein) against a transfer of Collateral (as defined herein). Each such transaction shall be referred to herein as a "Loan" and, unless otherwise agreed in writing, shall be governed by this Agreement, including any supplemental terms or conditions contained in an Annex or Schedule hereto and in any other annexes identified herein or therein as applicable hereunder. Capitalized terms not otherwise defined herein shall have the meanings provided in Section 25.

2. Loans of Securities.

- 2.1 Subject to the terms and conditions of this Agreement, Borrower or Lender may, from time to time, seek to initiate a transaction in which Lender will lend Securities to Borrower. Borrower and Lender shall agree on the terms of each Loan (which terms may be amended during the Loan), including the issuer of the Securities, the amount of Securities to be lent, the basis of compensation, the amount of Collateral to be transferred by Borrower, and any additional terms. Such agreement shall be confirmed (a) by a schedule and receipt listing the Loaned Securities provided by Borrower to Lender in accordance with Section 3.2, (b) through any system that compares Loans and in which Borrower and Lender are participants, or (c) in such other manner as may be agreed by Borrower and Lender in writing. Such confirmation (the "Confirmation"), together with the Agreement, shall constitute conclusive evidence of the terms agreed between Borrower and Lender with respect to the Loan to which the Confirmation relates, unless with respect to the Confirmation specific objection is made promptly after receipt thereof. In the event of any inconsistency between the terms of such Confirmation and this Agreement, this Agreement shall prevail unless each party has executed such Confirmation.
- 2.2 Notwithstanding any other provision in this Agreement regarding when a Loan commences, unless otherwise agreed, a Loan hereunder shall not occur until the Loaned Securities and the Collateral therefor have been transferred in accordance with Section 15.

3. Transfer of Loaned Securities.

- 3.1 Unless otherwise agreed, Lender shall transfer Loaned Securities to Borrower hereunder on or before the Cutoff Time on the date agreed to by Borrower and Lender for the commencement of the Loan.
- 3.2 Unless otherwise agreed, Borrower shall provide Lender, for each Loan in which Lender is a Customer, with a schedule and receipt listing the Loaned Securities. Such schedule and receipt may consist of (a) a schedule provided to Borrower by Lender and executed and returned by Borrower when the Loaned Securities are received, (b) in the case of Securities transferred through a Clearing Organization which provides transferors with a notice evidencing such transfer, such notice, or (c) a confirmation or other document provided to Lender by Borrower.
- 3.3 Notwithstanding any other provision in this Agreement, the parties hereto agree that they intend the Loans hereunder to be loans of Securities. If, however, any Loan is deemed to be a loan of money by Borrower to Lender, then Borrower shall have, and Lender shall be deemed to have granted, a security interest in the Loaned Securities and the proceeds thereof.

4. Collateral.

- 4.1 Unless otherwise agreed, Borrower shall, prior to or concurrently with the transfer of the Loaned Securities to Borrower, but in no case later than the Close of Business on the day of such transfer, transfer to Lender Collateral with a Market Value at least equal to the Margin Percentage of the Market Value of the Loaned Securities.
- 4.2 The Collateral transferred by Borrower to Lender, as adjusted pursuant to Section 9, shall be security for Borrower's obligations in respect of such Loan and for any other obligations of Borrower to Lender hereunder. Borrower hereby pledges with, assigns to, and grants Lender a continuing first priority security interest in, and a lien upon, the Collateral, which shall attach upon the transfer of the Loaned Securities by Lender to Borrower and which shall cease upon the transfer of the Loaned Securities by Borrower to Lender. In addition to the rights and remedies given to Lender hereunder, Lender shall have all the rights and remedies of a secured party under the UCC. It is understood that Lender may use or invest the Collateral, if such consists of cash, at its own risk, but that (unless Lender is a Broker-Dealer) Lender shall, during the term of any Loan hereunder, segregate Collateral from all securities or other assets in its possession. Lender may Retransfer Collateral only (a) if Lender is a Broker-Dealer or (b) in the event of a Default by Borrower. Segregation of Collateral may be accomplished by appropriate identification on the books and records of Lender if it is a "securities intermediary" within the meaning of the UCC.
- 4.3 Except as otherwise provided herein, upon transfer to Lender of the Loaned Securities on the day a Loan is terminated pursuant to Section 6, Lender shall be obligated to transfer the Collateral (as adjusted pursuant to Section 9) to Borrower no later than the Cutoff Time on such day or, if such day is not a day on which a transfer of such Collateral may be effected under Section 15, the next day on which such a transfer may be effected.
- 4.4 If Borrower transfers Collateral to Lender, as provided in Section 4.1, and Lender does not transfer the Loaned Securities to Borrower, Borrower shall have the absolute right to the return of the Collateral; and if Lender transfers Loaned Securities to Borrower and

- Borrower does not transfer Collateral to Lender as provided in Section 4.1, Lender shall have the absolute right to the return of the Loaned Securities.
- 4.5 Borrower may, upon reasonable notice to Lender (taking into account all relevant factors, including industry practice, the type of Collateral to be substituted, and the applicable method of transfer), substitute Collateral for Collateral securing any Loan or Loans; provided, however, that such substituted Collateral shall (a) consist only of cash, securities or other property that Borrower and Lender agreed would be acceptable Collateral prior to the Loan or Loans and (b) have a Market Value such that the aggregate Market Value of such substituted Collateral, together with all other Collateral for Loans in which the party substituting such Collateral is acting as Borrower, shall equal or exceed the agreed upon Margin Percentage of the Market Value of the Loaned Securities.
- 4.6 Prior to the expiration of any letter of credit supporting Borrower's obligations hereunder, Borrower shall, no later than the Extension Deadline, (a) obtain an extension of the expiration of such letter of credit, (b) replace such letter of credit by providing Lender with a substitute letter of credit in an amount at least equal to the amount of the letter of credit for which it is substituted, or (c) transfer such other Collateral to Lender as may be acceptable to Lender.

5. Fees for Loan.

- 5.1 Unless otherwise agreed, (a) Borrower agrees to pay Lender a loan fee (a "Loan Fee"), computed daily on each Loan to the extent such Loan is secured by Collateral other than cash, based on the aggregate Market Value of the Loaned Securities on the day for which such Loan Fee is being computed, and (b) Lender agrees to pay Borrower a fee or rebate (a "Cash Collateral Fee") on Collateral consisting of cash, computed daily based on the amount of cash held by Lender as Collateral, in the case of each of the Loan Fee and the Cash Collateral Fee at such rates as Borrower and Lender may agree. Except as Borrower and Lender may otherwise agree (in the event that cash Collateral is transferred by clearing house funds or otherwise), Loan Fees shall accrue from and including the date on which the Loaned Securities are transferred to Borrower to, but excluding, the date on which such Loaned Securities are returned to Lender, and Cash Collateral Fees shall accrue from and including the date on which the cash Collateral is transferred to Lender to, but excluding, the date on which such Collateral is returned to Borrower.
- 5.2 Unless otherwise agreed, any Loan Fee or Cash Collateral Fee payable hereunder shall be payable:
 - (a) in the case of any Loan of Securities other than Government Securities, upon the earlier of (i) the fifteenth day of the month following the calendar month in which such fee was incurred and (ii) the termination of all Loans hereunder (or, if a transfer of cash in accordance with Section 15 may not be effected on such fifteenth day or the day of such termination, as the case may be, the next day on which such a transfer may be effected); and
 - (b) in the case of any Loan of Government Securities, upon the termination of such Loan and at such other times, if any, as may be customary in accordance with market practice.

Notwithstanding the foregoing, all Loan Fees shall be payable by Borrower immediately in the event of a Default hereunder by Borrower and all Cash Collateral Fees shall be payable immediately by Lender in the event of a Default by Lender.

6. Termination of the Loan.

- 6.1 (a) Unless otherwise agreed, either party may terminate a Loan on a termination date established by notice given to the other party prior to the Close of Business on a Business Day. The termination date established by a termination notice shall be a date no earlier than the standard settlement date that would apply to a purchase or sale of the Loaned Securities (in the case of a notice given by Lender) or the non-cash Collateral securing the Loan (in the case of a notice given by Borrower) entered into at the time of such notice, which date shall, unless Borrower and Lender agree to the contrary, be (i) in the case of Government Securities, the next Business Day following such notice and (ii) in the case of all other Securities, the third Business Day following such notice.
 - (b) Notwithstanding paragraph (a) and unless otherwise agreed, Borrower may terminate a Loan on any Business Day by giving notice to Lender and transferring the Loaned Securities to Lender before the Cutoff Time on such Business Day if (i) the Collateral for such Loan consists of cash or Government Securities or (ii) Lender is not permitted, pursuant to Section 4.2, to Retransfer Collateral.
- Unless otherwise agreed, Borrower shall, on or before the Cutoff Time on the termination date of a Loan, transfer the Loaned Securities to Lender; provided, however, that upon such transfer by Borrower, Lender shall transfer the Collateral (as adjusted pursuant to Section 9) to Borrower in accordance with Section 4.3.

7. Rights in Respect of Loaned Securities and Collateral.

- 7.1 Except as set forth in Sections 8.1 and 8.2 and as otherwise agreed by Borrower and Lender, until Loaned Securities are required to be redelivered to Lender upon termination of a Loan hereunder, Borrower shall have all of the incidents of ownership of the Loaned Securities, including the right to transfer the Loaned Securities to others. Lender hereby waives the right to vote, or to provide any consent or to take any similar action with respect to, the Loaned Securities in the event that the record date or deadline for such vote, consent or other action falls during the term of the Loan.
- 7.2 Except as set forth in Sections 8.3 and 8.4 and as otherwise agreed by Borrower and Lender, if Lender may, pursuant to Section 4.2, Retransfer Collateral, Borrower hereby waives the right to vote, or to provide any consent or take any similar action with respect to, any such Collateral in the event that the record date or deadline for such vote, consent or other action falls during the term of a Loan and such Collateral is not required to be returned to Borrower pursuant to Section 4.5 or Section 9.

8. Distributions.

8.1 Lender shall be entitled to receive all Distributions made on or in respect of the Loaned Securities which are not otherwise received by Lender, to the full extent it would be so entitled if the Loaned Securities had not been lent to Borrower.